#### UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

# **FORM 10-Q**

(Mark One)	
[X]	<b>QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE</b>
	SECURITIES EXCHANGE ACT OF 1934
	For the quarterly period ended June 30, 2008
	or
[]	TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE
	SECURITIES EXCHANGE ACT OF 1934
	For the transition period from to
	Commission File Number: 0-19989



# **Stratus Properties Inc.**

(Exact name of registrant as specified in its charter)

**Delaware** (State or other jurisdiction of incorporation or organization) 72-1211572 (I.R.S. Employer Identification No.)

98 San Jacinto Blvd., Suite 220 Austin, Texas (Address of principal executive offices)

78701 (Zip Code)

(512) 478-5788

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. R Yes  $\Box$  No

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. Large accelerated filer  $\square$  Accelerated filer R Non-accelerated filer  $\square$  Smaller reporting company  $\square$ 

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). 🗆 Yes R No

On July 31, 2008, there were issued and outstanding 7,635,316 shares of the registrant's Common Stock, par value \$0.01 per share.

#### STRATUS PROPERTIES INC. TABLE OF CONTENTS

Part I. Financial Information	Page 2
Item 1. Financial Statements:	
Consolidated Balance Sheets (Unaudited)	2
Consolidated Statements of Operations (Unaudited)	3
Consolidated Statements of Cash Flows (Unaudited)	4
Notes to Consolidated Financial Statements (Unaudited)	5
Report of Independent Registered Public Accounting Firm	11
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	12
Item 3. Quantitative and Qualitative Disclosures about Market Risk	21
Item 4. Controls and Procedures	21
Part II. Other Information	21
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	21
Item 6. Exhibits	22
Signature	22
Exhibit Index	E-1

## STRATUS PROPERTIES INC. Part I. FINANCIAL INFORMATION

# Item 1. Financial Statements.

#### STRATUS PROPERTIES INC. CONSOLIDATED BALANCE SHEETS (Unaudited) (In Thousands)

		June 30, 2008		ember 31, 2007
ASSETS				
Current assets:				
Cash and cash equivalents	\$	34,693	\$	40,873
Restricted cash		6		112
Accounts receivable		1,306		2,315
Notes receivable from property sales		186		311
Deposits, prepaid expenses and other		6,393		79
Deferred tax asset		557		1,401
Total current assets		43,141		45,091
Real estate, commercial leasing assets and facilities, net:				
Property held for sale – developed or under development		139,238		129,759
Property held for sale – undeveloped		16,878		16,523
Property held for use, net		24,931		24,421
Investment in unconsolidated affiliate		2,004		4,226
Deferred tax asset		7,054		5,534
Other assets		5,736		2,803
Total assets	\$	238,982	\$	228,357
LIABILITIES AND STOCKHOLDERS' EQUITY Current liabilities: Accounts payable and accrued liabilities	\$	5,858	\$	6,324
Accrued interest, property taxes and other	Ŷ	5,709	Ψ	5,623
Current portion of long-term debt		271		242
Total current liabilities		11,838		12,189
Long-term debt		63,142		61,258
Other liabilities		1,859		2,510
Total liabilities		76,839		75,957
Minority interest in consolidated subsidiary		10,614		-
Stockholders' equity: Preferred stock				
Common stock		82		81
Capital in excess of par value of common stock		197,234		195,898
Accumulated deficit		(30,484)		(29,300)
Common stock held in treasury		(15,303)		(14,279)
Total stockholders' equity		151,529		152,400
	\$		\$	
Total liabilities and stockholders' equity	\$	238,982	\$	228,357

The accompanying notes are an integral part of these consolidated financial statements.

# STRATUS PROPERTIES INC. CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited) (In Thousands, Except Per Share Amounts)

	Three Months Ended June 30,			Six Months Ended June 30,			
		2008		2007	2008		2007
Revenues:							
Real estate	\$	2,399	\$	5,317	\$ 6,303	\$	9,743
Rental income		1,169		711	2,120		1,381
Commissions, management fees and other		655		760	 867		981
Total revenues		4,223		6,788	9,290		12,105
Cost of sales:							
Real estate, net		2,652		3,406	5,870		4,989
Rental		923		763	1,739		1,531
Depreciation		394		206	 777		483
Total cost of sales		3,969		4,375	8,386		7,003
General and administrative expenses		1,897		1,828	 3,554		3,814
Total costs and expenses		5,866		6,203	11,940		10,817
Operating (loss) income		(1,643)		585	(2,650)		1,288
Interest expense, net		(329)		(10)	(659)		(13)
Interest income		154		31	1,103		535
Equity in unconsolidated affiliate's income		222		-	778		-
(Loss) income from continuing operations							
before income taxes and minority interest		(1,596)		606	(1, 428)		1,810
Benefit from (provision for) income taxes		364		(189)	285		(631)
Minority interest in net loss of consolidated subsidiary		64		-	 64		-
(Loss) income from continuing operations		(1,168)		417	(1,079)		1,179
Loss from discontinued operations		(105)		(176)	(105)		(200)
Net (loss) income	\$	(1,273)	\$	241	\$ (1,184)	\$	979
Basic net (loss) income per share of common stock:							
Continuing operations	\$	(0.16)	\$	0.05	\$ (0.15)	\$	0.16
Discontinued operations		(0.01)		(0.02)	 (0.01)		(0.03)
Basic net (loss) income per share of common stock	\$	(0.17)	\$	0.03	\$ (0.16)	\$	0.13
Diluted net (loss) income per share of common stock:							
Continuing operations	\$	(0.16)	\$	0.05	\$ (0.15)	\$	0.15
Discontinued operations		(0.01)		(0.02)	 (0.01)		(0.02)
Diluted net (loss) income per share of common stock	\$	(0.17)	\$	0.03	\$ (0.16)	\$	0.13
Weighted average shares of common stock outstanding:							
Basic		7,631		7,568	 7,599		7,559
Diluted		7,631		7,690	 7,599		7,680

The accompanying notes are an integral part of these consolidated financial statements.

## STRATUS PROPERTIES INC. CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited) (In Thousands)

	Six Months Ended June 30,		
		2008	2007
Cash flow from operating activities:			
Net (loss) income	\$	(1,184) \$	979
Adjustments to reconcile net (loss) income to net cash (used in)			
provided by operating activities:			• • •
Loss from discontinued operations		105	200
Depreciation		777	483
Minority interest in net loss of consolidated subsidiary		(64)	-
Cost of real estate sold Deferred income taxes		4,634	5,358
		(676) 483	(157) 759
Stock-based compensation Equity in unconsolidated affiliate's income		(778)	/39
Distribution of unconsolidated affiliate's income		1,266	-
Deposits		(1,148)	(2,922)
Other		(361)	(2,922)
(Increase) decrease in working capital:		(301)	(10)
Accounts receivable, prepaid expenses and other		(5,764)	(276)
Accounts payable, accrued liabilities and other		335	1,314
Net cash (used in) provided by continuing operations		(2,375)	5,728
Net cash used in discontinued operations		(2,575)	(93)
		(2,375)	5,635
Net cash (used in) provided by operating activities		(2,373)	5,055
Cash flow from investing activities:			
Purchases and development of real estate properties		(19,065)	(17, 143)
Development of commercial leasing properties and other expenditures		(352)	(216)
Municipal utility district reimbursements		3,753	2,557
Return of investment in unconsolidated affiliate		2,374	-
Net cash used in continuing operations		(13,290)	(14,802)
Net cash used in discontinued operations		-	(118)
Net cash used in investing activities		(13,290)	(14,920)
Cash flow from financing activities:			
Borrowings from revolving credit facility		-	15,450
Payments on revolving credit facility		-	(18,450)
Borrowings from construction loan		2,022	-
Repayments on Lantana promissory note		(109)	-
Borrowings from unsecured term loans		-	15,000
Minority interest contributions		10,678	-
Net payments for exercised stock options		(114)	(35)
Excess tax benefit from exercised stock options		281	655
Purchases of Stratus common shares		(428)	(153)
Financing costs		(2,845)	(284)
Net cash provided by continuing operations		9,485	12,183
Net cash used in discontinued operations			(154)
Net cash provided by financing activities		9,485	12,029
Net (decrease) increase in cash and cash equivalents		(6,180)	2,744
Cash and cash equivalents at beginning of year		40,873	1,839
Cash and cash equivalents at end of period		34,693	4,583
Less cash at discontinued operations		-	(496)
Cash and cash equivalents at end of period	\$	34,693 \$	4,087
cash and cash squitaients at one of period			

The accompanying notes are an integral part of these consolidated financial statements.

#### STRATUS PROPERTIES INC. NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### 1. GENERAL

The accompanying unaudited consolidated financial statements should be read in conjunction with the consolidated financial statements and notes thereto for the year ended December 31, 2007, included in Stratus Properties Inc.'s (Stratus) Annual Report on Form 10-K (Stratus 2007 Form 10-K) filed with the Securities and Exchange Commission (SEC). In the opinion of management, the accompanying consolidated financial statements reflect all adjustments (consisting only of normal recurring items) considered necessary for a fair statement of the financial position of Stratus at June 30, 2008, and the results of operations for the three-month and six-month periods ended June 30, 2008 and 2007, and cash flows for the six-month periods ended June 30, 2008 and 2007. Operating results for the three-month and six-month periods ended June 30, 2008 are not necessarily indicative of the results that may be expected for the year ending December 31, 2008. Certain amounts from prior periods' financial statements have been reclassified to conform to the current year presentation. These reclassifications had no effect on previously reported net income or stockholders' equity.

#### 2. EARNINGS PER SHARE

Stratus' basic net (loss) income per share of common stock was calculated by dividing the (loss) income from continuing operations, loss from discontinued operations and net (loss) income by the weighted average number of common shares outstanding during the period. The following is a reconciliation of net (loss) income and weighted average common shares outstanding for purposes of calculating diluted net (loss) income per share (in thousands, except per share amounts):

	Three Months June 30		Six Months Ended June 30,			
	2008	2007	2008	2007		
(Loss) income from continuing operations	\$ (1,168) \$	417	\$ (1,07	9) \$ 1,179		
Loss from discontinued operations	 (105) <sup>a</sup>	(176)	(10:	$(200)^{a}$		
Net (loss) income	\$ (1,273) \$	241	\$ (1,184	4) <u>\$ 979</u>		
Weighted average common shares outstanding	7,631	7,568	7,59	9 7,559		
Add: Dilutive stock options	-	97		- 100		
Restricted stock	-	25		- 21		
Weighted average common shares outstanding for purposes of calculating diluted net (loss) income per share	 7,631	7,690	7,59	9 7,680		
Diluted net (loss) income per share of common stock:						
Continuing operations	\$ (0.16) \$	0.05		5) \$ 0.15		
Discontinued operations	 <u>(0.01)</u> <sup>a</sup>	(0.02)	(0.0	_//		
Diluted net (loss) income per share of common stock	\$ (0.17) \$	0.03	\$ (0.1	<u>6)</u> <u>\$ 0.13</u>		

a. Relates to the revised amount of Texas Margin Tax accrued on Escarpment Village income earned during 2007 (see Note 7).

Stock options and restricted stock units representing approximately 45,000 shares for the second quarter of 2008 and approximately 65,000 shares for the first six months of 2008 that otherwise would have been included in the earnings per share calculations were excluded because of the net loss reported for the 2008 periods.

## 3. JOINT VENTURE WITH CANYON-JOHNSON URBAN FUND II, L.P.

Effective May 1, 2008, Stratus entered into a joint venture with Canyon-Johnson Urban Fund II, L.P. ("Canyon-Johnson") for the development of a 36-story mixed-use development in downtown Austin, Texas, anchored by a W Hotel & Residences (the "W Austin Hotel & Residences project"). Stratus' initial capital contribution to the joint venture consisted of a 1.76 acre tract of land located across the street from Austin City Hall and the related property and development agreements.

In connection with the formation of the joint venture, Stratus received credit for its \$31.8 million of prior capital contributions. Stratus is the manager of the joint venture and has an approximate 40 percent interest in the joint venture. Canyon-Johnson has an approximate 60 percent interest in the joint venture. Canyon-Johnson contributed initial capital of \$10.7 million and will contribute additional capital until certain capital contribution requirements are met. In the aggregate, Canyon-Johnson will contribute approximately 60 percent of the joint venture's required capital and Stratus will contribute approximately 40 percent. The maximum capital contributions shall not exceed \$52.6 million for Stratus and \$74.6 million for Canyon-Johnson.

A Stratus subsidiary has been designated as the developer of the W Austin Hotel & Residences project and will be paid a \$6.0 million developer's fee over the term of construction, including \$0.1 million for the three-month and six-month periods ended June 30, 2008.

On May 2, 2008, the joint venture entered into a construction loan agreement with Corus Bank, N.A., (the "Loan Agreement") to finance the construction of the W Austin Hotel & Residences project. Pursuant to the Loan Agreement, the joint venture may borrow up to an aggregate of \$165.0 million to fund the construction, development and marketing costs of the W Austin Hotel & Residences project. Upon execution of the Loan Agreement, approximately \$2.0 million was advanced to the joint venture. In connection with the Loan Agreement, the joint venture paid \$2.8 million of financing costs. Pursuant to the terms of the Loan Agreement, additional borrowings are not permitted until Stratus and Canyon-Johnson have contributed their required capital, which is anticipated to occur in the fourth quarter of 2009.

The Loan Agreement contains customary financial covenants and other restrictions. Amounts borrowed under the Loan Agreement bear interest at an annual rate equal to the greater of (1) the sum of 3.5 percent per year plus the three-month London Interbank Offered Rate (LIBOR) quoted in the Money Rates section of The Wall Street Journal or (2) 6.5 percent. The rate was 6.5 percent at June 30, 2008. On August 1, 2008, the joint venture paid \$0.7 million to enter into an agreement to cap the floating LIBOR rate on the loan at 4.5 percent. The LIBOR rate cap notional amount varies based on projected loan balances throughout the term of the loan and was \$2.0 million on August 1, 2008. The agreement terminates on July 1, 2011.

Optional prepayments during the twelve months immediately following the execution of the Loan Agreement are not permitted. From May 2, 2009 through November 2, 2010, optional prepayments of the loan are permitted, subject to a prepayment premium. Optional prepayments made after November 2, 2010 are not subject to prepayment premiums. Repayments made from proceeds of the sale of residential condominiums or other components of the W Austin Hotel & Residences project are permitted, beginning after the first year of the loan, without prepayment penalty. Repayments under the Loan Agreement may be accelerated by the lenders upon the occurrence of customary events of default. The Loan Agreement matures on September 2, 2011. Certain obligations of the joint venture under the Loan Agreement are guaranteed by Stratus, including construction and completion of the project, environmental indemnification and joint and several liability for the payment of \$20.0 million of the principal of the loan.

Under the guidance of FASB Interpretation No. (FIN) 46R, "Consolidation of Variable Interest Entities (revised December 2003) - an Interpretation of ARB No. 51," Stratus evaluated whether the W Austin Hotel & Residences project is a variable interest entity (VIE). Stratus concluded that the project is a VIE and, although not majority owned, Stratus is currently the primary beneficiary.

Accordingly, the W Austin Hotel & Residences project has been consolidated in Stratus' financial statements. At June 30, 2008, Stratus' consolidated balance sheet includes \$49.1 million in total assets and \$6.6 million in total liabilities associated with the W Austin Hotel & Residences project. In accordance with FIN 46R, certain triggering events, including when the VIE has additional equity investment at risk, require a company to reconsider whether or not an entity is still a VIE and also requires consideration of the primary beneficiary. Therefore, as future capital contributions are made by Canyon-Johnson and Stratus, Stratus will update its evaluation of whether the project is a VIE and whether Stratus is the primary beneficiary. If it is determined that the W Austin Hotel & Residences project is no longer a VIE under the guidance of FIN 46R or that Stratus is no longer the primary beneficiary of the entity, the project will be deconsolidated from Stratus' financial statements and would be accounted for under the equity method of accounting.

#### 4. INVESTMENT IN UNCONSOLIDATED AFFILIATE

In 2005, Stratus formed a joint venture with Trammell Crow Central Texas Development, Inc. (Trammell Crow) to acquire an approximate 74-acre tract at the intersection of Airport Boulevard and Lamar Boulevard in Austin, Texas, for \$7.7 million. The property, known as Crestview Station, is a single-family, multi-family, retail and office development, which is located on the commuter rail line approved by City of Austin voters. With Trammell Crow, Stratus has completed environmental remediation and permitting of the property and is now proceeding with infrastructure development. In September 2007, the State of Texas certified that the remediation was complete.

At June 30, 2008, Stratus' investment in the Crestview Station project totaled \$2.0 million and the joint venture partnership had \$5.3 million of outstanding debt, of which each joint venture partner guarantees \$2.7 million.

Stratus has a 50 percent interest in the Crestview Station project, which it accounts for under the equity method in accordance with the provisions of the American Institute of Certified Accountants Statement of Position 78-9, "Accounting for Investments in Real Estate Ventures." Stratus has determined that consolidation of the Crestview Station project is not required under the provisions of Financial Accounting Standards Board Interpretation No. 46R, "Consolidation of Variable Interest Entities."

Summary unaudited information for Crestview Station for the 2008 periods follows (in thousands):

		Three Months Ended June 30, 2008		Six Months Ended June 30, 2008
	<u> </u>		<i>.</i>	2.020
Total revenues	\$	1,701	\$	3,258
Net income	\$	443	\$	1,555

During the six months ended June 30, 2007, the Crestview Station project recorded no revenues or expenses as the project was under development and all expenditures were capitalized as project costs.

## 5. DEBT OUTSTANDING

At June 30, 2008, Stratus had total debt of \$63.4 million, including \$0.3 million of current debt, and total debt of \$61.5 million at December 31, 2007. Stratus' debt outstanding at June 30, 2008 consisted of the following:

- \$40.0 million of borrowings outstanding under seven unsecured term loans, including two \$5.0 million loans, two \$8.0 million loans, a \$7.0 million loan and two \$3.5 million loans, all of which will mature in December 2011.
- \$21.4 million of borrowings outstanding under the Lantana Promissory Note, which matures in January 2018.
- \$2.0 million of borrowings outstanding under the W Austin Hotel & Residences project construction loan, which matures in September 2011.

Effective May 30, 2008, Stratus entered into a third modification and extension agreement to extend the maturity and modify the interest rate on its \$45.0 million revolving credit facility. The maturity was extended from May 30, 2009, to May 30, 2010. In addition, the interest rate applicable to amounts borrowed under the facility was modified to an annual rate of either the base rate minus 0.45 percent with a minimum interest rate of 5 percent or the London Interbank Offered Rate plus 2 percent with a minimum interest rate of 5 percent. No amounts were outstanding under this facility at June 30, 2008. For a further discussion of Stratus' debt see Note 4 of the Stratus 2007 Form 10-K.

## 6. INTEREST COST AND STOCK-BASED COMPENSATION

Interest Cost. Capitalized interest totaled \$0.8 million in the second quarter of 2008, \$0.7 million in the second quarter of 2007, \$1.5 million in the first six months of 2008 and \$1.3 million in the first six months of 2007.

Stock-Based Compensation. Stock-based compensation costs are capitalized as appropriate. Compensation cost charged against earnings for stock-based awards is shown below (in thousands).

	Three Months Ended June 30,			Six Months Ended June 30,				
		2008		2007		2008		2007
Stock options awarded to employees (including directors)	\$	94	\$	118	\$	189	\$	235
Restricted stock units		193		157		386		665
Less capitalized amounts		(46)	_	(43)		(92)		(141)
Impact on (loss) income from continuing operations								
before income taxes	\$	241	\$	232	\$	483	\$	759

For more information regarding Stratus' stock-based awards see Notes 1 and 6 of the Stratus 2007 Form 10-K.

#### 7. DISCONTINUED OPERATIONS

On October 12, 2007, Stratus sold the Escarpment Village shopping center, located in Austin, Texas, to Lake Villa, L.L.C. (the Purchaser) for \$46.5 million, before closing costs and other adjustments. The Purchaser paid approximately \$23.0 million in cash at closing and assumed the \$22.4 million principal balance remaining under Stratus' loan from Teachers Insurance and Annuity Association of America (TIAA). Upon completion of the sale of Escarpment Village, Stratus ceased all involvement with the Escarpment Village shopping center. The results of operations of Escarpment Village, which have been classified as discontinued operations in the accompanying consolidated statements of income, previously represented a component of Stratus' commercial leasing segment.

In June 2008, Stratus revised the amount of Texas Margin Tax accrued on Escarpment Village income earned during 2007. The revised accrual resulted in \$0.1 million additional tax expense related to 2007, which has been recognized in June 2008. As the results of operations of Escarpment Village have been appropriately classified as discontinued operations, the additional Texas Margin Tax has also been classified as discontinued operations in the accompanying consolidated statements of operations.

Summary Escarpment Village results for the 2007 periods follows (in thousands):

	Months Ended e 30, 2007	Six Months Ended June 30, 2007
Rental income	\$ 867 \$	1,756
Rental property costs	(552)	(896)
Depreciation	(264)	(526)
General and administrative expenses	(18)	(33)
Interest expense <sup>a</sup>	(329)	(659)
Interest income	25	50
Loss before income taxes	(271)	(308)
Benefit from income taxes	95	108
Loss from discontinued operations	\$ (176) \$	(200)

a. Relates to interest expense from the Escarpment Village loan from TIAA and does not include any additional allocations of interest.

For a further discussion of Stratus' discontinued operations see Note 7 of the Stratus 2007 Form 10-K.

#### 8. BUSINESS SEGMENTS

Stratus has two operating segments, "Real Estate Operations" and "Commercial Leasing." The Real Estate Operations segment is comprised of all Stratus' developed properties, properties under development and undeveloped properties in Austin, Texas, which consist of its properties in the Barton Creek community, the Circle C community and Lantana. In January 2008, Stratus sold the final lots of the Deerfield property in Plano, Texas, which is also included in the Real Estate Operations segment.



The Commercial Leasing segment primarily includes the two 75,000 square-foot office buildings at 7500 Rialto Boulevard of which one is 97 percent leased and the other is 94 percent leased as of June 30, 2008. In addition, the commercial leasing segment for second-quarter 2008 includes rental income from Barton Creek Village, which includes a retail building completed in second-quarter 2007 and a bank building completed in early 2008.

The Escarpment Village operating results are reported as discontinued operations.

The segment data presented below were prepared on the same basis as Stratus' consolidated financial statements.

		Real Estate Operations <sup>a</sup>		Commercial Leasing		Other		Total
				(In Tho	usan	nds)		
Three Months Ended June 30, 2008 Revenues	\$	3.054 \$	¢	1.169	\$		\$	4.223
Cost of sales, excluding depreciation	φ	(2,652)	φ	(923)	Ф	-	Ф	(3,575)
Depreciation		(48)		(346)		-		(394)
General and administrative expenses		(1,631)		(266)		-		(1,897)
Operating loss	\$	(1,277) \$	\$	(366)	\$	-	\$	(1,643)
Loss from discontinued operations	\$	- \$	\$	(105) <sup>b</sup>	\$	-	\$	(105) <sup>b</sup>
Benefit from income taxes	\$	364 \$	\$	-	\$	-	\$	364
Capital expenditures	\$	10,765 \$	\$	79	\$	_	\$	10,844
	\$	197,199 \$	\$	33,740	\$	8,043 <sub>c</sub>	\$	238,982
Total assets	ψ	197,199 <del>u</del>	Ψ	55,740	Ψ	<u>0,045</u> c	Ψ	250,902
Three Months Ended June 30, 2007								
Revenues	\$	6,077 \$	\$	711	\$	-	\$	6,788
Cost of sales, excluding depreciation		(3,406)		(763)		-		(4,169)
Depreciation		(38)		(168)		-		(206)
General and administrative expenses	-	(1,587)		(241)	_	-		(1,828)
Operating income (loss)	\$	1,046 \$	\$	(461)	\$	-	\$	585
Loss from discontinued operations	\$	- \$		(176)	\$		\$	(176)
Provision for income taxes	\$	(189) \$	\$		\$	_	\$	(189)
Capital expenditures	\$	7,967 \$	\$	212	\$	-	\$	8,179
Total assets	\$	145,473 \$	\$	61,768 <sub>d</sub>	\$	8,795 <sub>c</sub>	\$	216,036
Six Months Ended June 30, 2008								
Revenues	\$	7,170 \$	\$	2,120	\$	-	\$	9,290
Cost of sales, excluding depreciation		(5,870)		(1,739)		-		(7,609)
Depreciation		(94)		(683)		-		(777)
General and administrative expenses	<u>_</u>	(3,056)	¢.	(498)	<b></b>	-	<u>_</u>	(3,554)
Operating loss	\$	(1,850) \$		(800)	\$	=	\$	(2,650)
Loss from discontinued operations	\$	- \$		<u>(105</u> ) <sup>b</sup>	\$	-	\$	<u>(105)</u> <sup>b</sup>
Benefit from income taxes	\$	285 \$	\$	-	\$	-	\$	285
Capital expenditures	\$	19,065 \$	\$	352	\$		\$	19,417
Six Months Ended June 30, 2007								
Revenues	\$	10,724 \$	\$	1,381	\$	-	\$	12,105
Cost of sales, excluding depreciation		(4,989)		(1,531)		-		(6,520)
Depreciation		(70)		(413)		-		(483)
General and administrative expenses	¢	(3,308)	ሰ	(506)	¢		¢	(3,814)
Operating income (loss)	\$	2,357 \$		(1,069)	\$	-	\$	1,288
Loss from discontinued operations	\$	- \$		(200)	\$	-	\$	(200)
Provision for income taxes	\$	(631) \$			\$	-	\$	(631)
Capital expenditures	\$	17,143 \$	\$	334	\$		\$	17,477

- a. Includes sales commissions, management fees and other revenues together with related expenses.
- b. Relates to the revised amount of Texas Margin Tax accrued on Escarpment Village income earned during 2007 (see note 7).
- c. Primarily includes deferred tax assets.
- d. Includes assets from the discontinued operations of Escarpment Village, which Stratus sold on October 12, 2007, totaling \$34.9 million, net of accumulated depreciation of \$1.3 million, at June 30, 2007.

#### 9. NEW ACCOUNTING STANDARDS

*Fair Value Measurements.* In September 2006, the Financial Accounting Standards Board (FASB) issued Statement of Financial Accounting Standards (SFAS) No. 157, "Fair Value Measurements," which provides enhanced guidance for using fair value to measure assets and liabilities. SFAS No. 157 does not require any new fair value measurements under U.S. GAAP but rather establishes a common definition of fair value, provides a framework for measuring fair value under U.S. GAAP and expands disclosure requirements about fair value measurements. In February 2008, the FASB issued FSP FAS 157-2, which delays the effective date of SFAS No. 157 for nonfinancial assets or liabilities that are not required or permitted to be measured at fair value on a recurring basis to fiscal years beginning after November 15, 2008, and interim periods within those years. Effective January 1, 2008, Stratus adopted SFAS No. 157 for financial assets and liabilities subject to fair value measurement on a recurring basis. Stratus did not have financial assets and liabilities subject to fair value measurement on a recurring basis. Stratus is currently evaluating the impact that the adoption of SFAS No. 157 for nonfinancial assets or liabilities that are not required or permitted to be measured at fair value on a recurring basis will have on its financial reporting and disclosures.

<u>The Fair Value Option for Financial Assets and Liabilities</u>. In February 2007, the FASB issued SFAS No. 159, "The Fair Value Option for Financial Assets and Liabilities – Including an amendment of FASB Statement No. 115," which permits entities to choose to measure many financial instruments and certain other items at fair value that are not currently required to be measured at fair value. Effective January 1, 2008, Stratus adopted SFAS No. 159, which did not impact Stratus' financial reporting and disclosures.

<u>Noncontrolling Interests in Consolidated Financial Statements</u>. In December 2007, the FASB issued SFAS No. 160, "Noncontrolling Interests in Consolidated Financial Statements – an amendment of ARB No. 51," which clarifies that noncontrolling interests (minority interests) are to be treated as a separate component of equity and any changes in the ownership interest (in which control is retained) are to be accounted for as capital transactions. However, a change in ownership of a consolidated subsidiary that results in a loss of control is considered a significant event that triggers gain or loss recognition, with the establishment of a new fair value basis in any remaining ownership interests. SFAS No. 160 also provides additional disclosure requirements for each reporting period. SFAS No. 160 applies to fiscal years beginning on or after December 15, 2008, with early adoption prohibited. This statement is required to be adopted prospectively, except for the following provisions, which are expected to be applied retrospectively: (i) the reclassification of noncontrolling interests to equity in the consolidated balance sheets and (ii) the adjustment to consolidated net income to include net income attributable to both the controlling and noncontrolling interests. Stratus is currently evaluating the impact that the adoption of SFAS No. 160 will have on its financial reporting and disclosures.

<u>The Hierarchy of Generally Accepted Accounting Principles.</u> In May 2008, FASB issued SFAS No. 162, "The Hierarchy of Generally Accepted Accounting Principles," which identifies the sources of accounting and the framework for selecting the principles to be used in the preparation of financial statements of nongovernmental entities that are presented in conformity with U.S. GAAP. SFAS No. 162 is effective 60 days following the SEC's approval of the Public Company Accounting Oversight Board amendments to AU Section 411, "The Meaning of Presenting Fairly in Conformity with Generally Accepted Accounting Principles." The adoption of SFAS No.162 is not expected to result in a change in Stratus' current accounting practices.

#### REVIEW BY INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

The consolidated financial statements as of June 30, 2008 and for the three-month and six-month periods ended June 30, 2008 and 2007 have been reviewed by PricewaterhouseCoopers LLP, an independent registered public accounting firm. Their report (dated August 11, 2008) is included below. The report of PricewaterhouseCoopers LLP states that they did not audit and they do not express an opinion on that unaudited financial information. Accordingly, the degree of reliance on their report on such financial information should be restricted in light of the limited nature of the review procedures applied. PricewaterhouseCoopers LLP is not subject to the liability provisions of Section 11 of the Securities Act of 1933 for their report on the unaudited financial information because that report is not a "report" or a "part" of a registration statement prepared or certified by PricewaterhouseCoopers LLP within the meaning of Sections 7 and 11 of the Act.

## REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors and Stockholders of Stratus Properties Inc.:

We have reviewed the accompanying consolidated balance sheet of Stratus Properties Inc. and its subsidiaries as of June 30, 2008 and the related consolidated statements of operations for each of the three-month and six-month periods ended June 30, 2008 and 2007 and the consolidated statements of cash flows for the six-month periods ended June 30, 2008 and 2007. These interim financial statements are the responsibility of the Company's management.

We conducted our review in accordance with the standards of the Public Company Accounting Oversight Board (United States). A review of interim financial information consists principally of applying analytical procedures and making inquiries of persons responsible for financial and accounting matters. It is substantially less in scope than an audit conducted in accordance with the standards of the Public Company Accounting Oversight Board (United States), the objective of which is the expression of an opinion regarding the financial statements taken as a whole. Accordingly, we do not express such an opinion.

Based on our review, we are not aware of any material modifications that should be made to the accompanying consolidated interim financial statements for them to be in conformity with accounting principles generally accepted in the United States of America.

We previously audited in accordance with the standards of the Public Company Accounting Oversight Board (United States), the consolidated balance sheet as of December 31, 2007, and the related consolidated statements of income, of changes in stockholders' equity and of cash flows for the year then ended (not presented herein), and in our report dated March 14, 2008, we expressed an unqualified opinion on those consolidated financial statements with an explanatory paragraph for the Company's change in accounting for stock-based compensation. In our opinion, the information set forth in the accompanying consolidated balance sheet information as of December 31, 2007, is fairly stated in all material respects in relation to the consolidated balance sheet from which it has been derived.

#### /s/ PricewaterhouseCoopers LLP

Dallas, Texas August 11, 2008

## Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

#### **OVERVIEW**

Management's discussion and analysis presented below should be read in conjunction with our discussion and analysis of financial results contained in our 2007 Annual Report on Form 10-K (2007 Form 10-K) filed with the Securities and Exchange Commission (SEC). The operating results summarized in this report are not necessarily indicative of our future operating results. All subsequent references to Notes refer to Notes to Consolidated Financial Statements, unless otherwise stated.

We are engaged in the acquisition, development, management and sale of commercial, multi-family and residential real estate properties located primarily in the Austin, Texas area. We conduct real estate operations on properties we own.

Our principal real estate holdings are currently in southwest Austin, Texas. We also own undeveloped commercial property in San Antonio, Texas. Our developed lots, developed or under development acreage and undeveloped acreage as of June 30, 2008, are provided in the following table.

		Acreage								
		Γ	Developed of	r Under Developmer	nt		Undeveloped			
	Developed Lots	Single Family	Multi- family	Commercial	Total	Single Family	Commercial	Total	Total Acreage	
<u>Austin</u>										
Barton Creek	23	642	249	376	1,267	391	20	411	1,678	
Lantana	-	-	-	223	223	-	-	-	223	
Circle C	163	a 148a	-	12	160	-	375	375	535	
W Austin Hotel										
& Residences	-	-	-	2	ь 2	-	-	-	2	
San Antonio										
Camino Real			-				2	2	2	
Total	186	790	249	613	1,652	391	397	788	2,440	

a. Relates to Meridian, an 800-lot residential development.

b. Represents a city block in downtown Austin planned for a mixture of hotel, residential, retail, office and entertainment uses.

Our remaining Austin holdings at June 30, 2008, consisted of two 75,000-square-foot office buildings at 7500 Rialto Boulevard (7500 Rialto) located in Lantana.

In November 2005, we formed a joint venture partnership with Trammell Crow Central Texas Development, Inc. (Trammell Crow) to acquire an approximate 74-acre tract at the intersection of Airport Boulevard and Lamar Boulevard in Austin, Texas for \$7.7 million. The property, known as Crestview Station, is a single-family, multi-family, retail and office development.

On October 12, 2007, we sold Escarpment Village, which is a 168,000-square-foot retail center anchored by a grocery store in the Circle C Ranch (Circle C) community, for \$46.5 million, before closing costs and other adjustments. Accordingly, we have reported Escarpment Village's results of operations for the three-month and six-month periods ended June 30, 2007, as discontinued operations.

## BUSINESS STRATEGY

Our financial condition and results of operations are highly dependent upon market conditions in Austin, Texas. Our future operating cash flows and, ultimately, our ability to develop our properties and expand our business will be largely dependent on the level of our real estate sales. In turn, these sales will be significantly affected by future real estate market conditions in Austin, Texas, development costs, interest rate levels and regulatory issues including our land use and development entitlements. In the fourth quarter of 2007, the real estate market in the United States began to show signs of weakness, as credit markets became unpredictable and mixed views developed regarding the economy.

Our future performance may in part be dependent upon the credit markets settling and the underlying strength of the U.S. economy.

Over the past several years, we have successfully worked cooperatively with the City of Austin (the City) to obtain approvals that allow the development of our properties to proceed in a timely manner while protecting the environment. We believe the desirable location and overall quality of our properties, in combination with the land use and development entitlements we have obtained, will command a premium over the value of other Austin-area properties.

Our long-term success will depend on our ability to maximize the value of our real estate through obtaining required approvals that permit us to develop and sell our properties in a timely manner at a reasonable cost. We must incur significant development expenditures and secure additional permits prior to the development and sale of certain properties. In addition, we continue to pursue additional development opportunities, and believe we can obtain bank financing for developing our properties at a reasonable cost. See "Risk Factors" located in Item 1A. of our 2007 Form 10-K.

#### DEVELOPMENT AND OTHER ACTIVITIES

The number of our residential developed lots, lots under development and development potential by area are shown below (excluding lots and units associated with our Canyon-Johnson and Crestview joint ventures):

	As of June 30, 2008							
	Developed	Under Development	Potential Development <sup>a</sup>	Total				
Barton Creek:								
Verano Drive	-	71	-	71				
Calera Drive	8	-	-	8				
Calera Court Courtyard Homes	5	-	-	5				
Mirador Estate	2	-	-	2				
Wimberly Lane Phase II	1	-	-	1				
Amarra Drive	7	35	89	131				
Amarra Townhomes	-	-	97	97				
Section N Multi-family	-	-	1,860	1,860				
Other Barton Creek Sections	-	-	154	154				
Circle C:								
Meridian	163	57		220				
Total Residential Lots	186	163	2,200	2,549				

a. Our development of the properties identified under the heading "Potential Development" is dependent upon the approval of our development plans and permits by governmental agencies, including the City of Austin. Those governmental agencies may either not approve one or more development plans and permit applications related to such properties or require us to modify our development plans. Accordingly, our development strategy with respect to those properties may change in the future.

The number of square feet of our commercial property developed, under development and our remaining entitlements are shown below (excluding property associated with our Canyon-Johnson and Crestview joint ventures):

	As of June 30, 2008								
	Developed	Under Development	Potential Development <sup>a</sup>	Total					
Barton Creek:									
Barton Creek Village Phase I	22,000	-	-	22,000					
Barton Creek Village Phase II		-	18,000	18,000					
Entry Corner	-	-	5,000	5,000					
Amarra Retail/Office	-	-	90,000	90,000					
Section N	-	-	1,500,000	1,500,000					
Circle C:									
Chase Ground Lease	4,000			4,000					
Tract 106	4,000	22,000	-	22,000					
Tract 110	_		760,000	760,000					
Tract 107	_	-	110,000	110,000					
Tract 101	_	-	90,000	90,000					
Tract 102	_	-	25,000	25,000					
Tract 114	-	-	5,000	5,000					
			2,000	-,					
Lantana:									
7500 Rialto	150,000	-	-	150,000					
Advanced Micro Devices	· · · · · · · · · · · · · · · · · · ·			, i i i i i i i i i i i i i i i i i i i					
Option Tracts	-	-	760,000	760,000					
Tract GR1	-	-	325,000	325,000					
Tract G07	-	-	210,000	210,000					
Tract CS5	-	-	175,000	175,000					
Tract CS1-CS3	-	-	150,000	150,000					
Tract LR1	-	-	75,000	75,000					
Tract L04	-	-	70,000	70,000					
Austin 290 Tract			20,000	20,000					
Austin 290 Tract	-	-							
Total Square Feet	176,000	22,000	4,388,000	4,586,000					

a. Our development of the properties identified under the heading "Potential Development" is dependent upon the approval of our development plans and permits by governmental agencies, including the City of Austin. Those governmental agencies may either not approve one or more development plans and permit applications related to such properties or require us to modify our development plans. Accordingly, our development strategy with respect to those properties may change in the future.

*W Austin Hotel & Residences.* In 2005, the City selected our proposal to develop a mixed-use project in downtown Austin immediately north of the new City Hall complex. The W Austin Hotel & Residences project includes an entire city block and is planned for a mixture of hotel, residential, retail, office and entertainment uses. In December 2006, we acquired the property for \$15.1 million. We have executed agreements with Starwood Hotels & Resorts Worldwide, Inc. for the development of a W Hotel & Residences on the site. In May 2007, we announced our proposed partnership with Canyon-Johnson Urban Fund II, L.P. (Canyon-Johnson) for the development of the W Austin Hotel & Residences project. The grand opening for the onsite sales center was held in conjunction with the groundbreaking ceremony in October 2007. Effective May 1, 2008, we entered into a joint venture with Canyon-Johnson for the development of the project (see Note 3). Construction of the project commenced in the second quarter of 2008.

*Crestview Station.* In 2005, we formed a joint venture with Trammell Crow to acquire an approximate 74-acre tract at the intersection of Airport Boulevard and Lamar Boulevard in Austin, Texas, for \$7.7 million. The property, known as Crestview Station, is a single-family, multi-family, retail and office development, which is located on the commuter rail line approved by City of Austin voters. With Trammell Crow, we have completed environmental remediation and permitting of the property and are now proceeding with infrastructure development. In September 2007, the State of Texas certified that the remediation was complete. At June 30, 2008, our investment in the Crestview Station project totaled \$2.0 million and the

joint venture partnership had \$5.3 million of outstanding debt, of which each joint venture partner guarantees \$2.7 million.

Barton Creek Community. In 2002, we secured subdivision plat approval for a new residential subdivision called Calera, which consists of 155 lots. At June 30, 2008, our remaining unsold developed lots within the Barton Creek Community, included: Calera Drive - 8 lots, Amarra Drive Phase I - 7 lots, Calera Court - 5 lots, Mirador - 2 lots and Wimberly Lane Phase II - 1 lot.

During 2004, we began construction of courtyard homes at Calera Court, the initial phase of the Calera subdivision, which will include 16 home sites on 16 acres. The second phase of Calera, Calera Drive, consisting of 53 single-family lots, many of which adjoin the Fazio Canyons Golf Course, received final plat and construction permit approval in 2005. As of June 30, 2008, only eight lots remained unsold at Calera Drive. Construction of the final phase, known as Verano Drive, which includes 71 single-family lots, began in the first quarter of 2007 and was completed in July 2008.

During 2007, we completed development of Amarra Drive Phase I, the initial phase of the Amarra Drive subdivision. Amarra Drive Phase I includes eight lots, one of which was sold in September 2007, with sizes ranging from approximately one to four acres, some of which are course-side lots on the Fazio Canyons Golf Course and others are secluded lots adjacent to the Nature Conservancy of Texas. In January 2008, we commenced development of Amarra Drive Phase II, which will consist of 35 lots on 51 acres and two townhome tracts on 31 acres.

In the second quarter of 2007, we completed the first phase of the Barton Creek Village. The first phase includes a 22,000-square-foot retail complex. In July 2007, we began construction of a 3,300-square-foot bank building within this 22,000-square-foot retail complex, and it was completed in early 2008. Construction of the second retail complex will begin by 2009.

In 2004, we entered into a contract with a national homebuilder to sell 41 lots within the Wimberly Lane Phase II subdivision in the Barton Creek community. The average purchase price for each of the 41 lots was \$150,400, subject to a six percent annual escalator commencing in December 2004. We sold the last homebuilder lot in January 2008 and have one Wimberly Lane lot remaining for sale.

*Circle C Community.* We are developing the Circle C community based on the entitlements secured in our Circle C settlement with the City. Our Circle C settlement, as amended in 2004, permits development of 1.16 million square feet of commercial space, 504 multi-family units and 830 single family residential lots. Meridian is an 800-lot residential development at the Circle C community. In 2005, the first phase of construction commenced. During the first quarter of 2005, we contracted to sell a total of 494 lots in our Meridian project to three national homebuilders in four phases. Sales for each of the four phases commence upon substantial completion of development for that phase, and continue every quarter until all of the lots have been sold. The first and second phases each consisted of 134 lots. The first phase was substantially completed at the end of 2005. Development of the second phase was substantially completed in March 2006. Development of the 108-lot third phase of Meridian was completed in September 2007. The 118-lot fourth phase commenced in early 2008 and was completed in June 2008.

In 2006, we signed another contract with a national homebuilder for 42 additional lots. Development of those lots commenced in April 2007 and substantial completion occurred in April 2008. Development of the final phase of Meridian, which consists of 57 one-acre lots, is expected to commence in 2009.

We estimate our sales in Meridian will total at least 23 lots for \$1.5 million during the third quarter of 2008.

*Lantana*. Lantana is a partially developed, mixed-use project with remaining entitlements for approximately 1.0 million square feet of office and retail use on 223 acres as of June 30, 2008. Regional utility and road infrastructure is in place with capacity to serve Lantana at full build-out permitted under our existing entitlements.

In 2006, we completed a second 75,000-square-foot office building at 7500 Rialto in response to increased demand for office space within Lantana. As of June 30, 2008, we had leased approximately 97 percent of the space at the original office building and 94 percent of the space at the second office building.



*Deerfield.* In 2004, we acquired the Deerfield property in Plano, Texas, for \$7.0 million. We executed agreements with a national homebuilder, whereby the homebuilder paid us \$1.4 million for an option to purchase all 234 lots over 36 monthly take-downs. The net purchase price for each of the 234 lots was \$61,500, subject to certain terms and conditions. The \$1.4 million option payment was applied against subsequent purchases of lots by the homebuilder after certain thresholds were achieved and was recognized by us as income as lots were sold. In 2005, we executed a revised agreement with the homebuilder, increasing the lot sizes and average purchase price to \$67,150 based on a new total of 224 lots. In January 2008, we sold the final 21 lots for \$1.4 million.

#### **RESULTS OF OPERATIONS**

We are continually evaluating the development potential of our properties and will continue to consider opportunities to enter into significant transactions involving our properties. As a result, and because of numerous other factors affecting our business activities as described herein, our past operating results are not necessarily indicative of our future results.

Summary operating results follow (in thousands):

	Second Quarter			Six Months			15
	 2008 200		2007	2008			2007
Revenues:				_			
Real estate operations	\$ 3,054	\$	6,077	\$	7,170	\$	10,724
Commercial leasing	 1,169		711		2,120		1,381
Total revenues	\$ 4,223	\$	6,788	\$	9,290	\$	12,105
Operating (loss) income	\$ (1,643)	\$	585	\$	(2,650)	\$	1,288
Benefit from (provision for) income taxes	\$ 364	\$	(189)	\$	285	\$	(631)
(Loss) income from continuing operations	\$ (1,168)	\$	417	\$	(1,079)	\$	1,179
Loss from discontinued operations	 (105)	1	(176)		(105)	1	(200)
Net (loss) income	\$ (1,273)	\$	241	\$	(1,184)	\$	979

a. Relates to the revised amount of Texas Margin Tax accrued on Escarpment Village income earned during 2007 (see note 7).

We have two operating segments, "Real Estate Operations" and "Commercial Leasing" (see Note 8). The following is a discussion of our operating results by segment.

#### **Real Estate Operations**

Summary real estate operating results follow (in thousands):

	Second Quarter			Six M	onths
	2008		2007	2008	2007
Revenues:					
Developed property sales	\$	2,358 \$	5,317	\$ 6,262	\$ 8,660
Undeveloped property sales		41	-	41	1,083
Commissions, management fees and other		655	760	867	981
Total revenues		3,054	6,077	7,170	10,724
Cost of sales, including depreciation		(2,700)	(3,444)	(5,964)	(5,059)
General and administrative expenses		(1,631)	(1,587)	(3,056)	(3,308)
Operating (loss) income	\$	(1,277) \$	1,046	<u>\$ (1,850)</u>	<u>\$ 2,357</u>



Developed Property Sales. Property sales for the second-quarter and six-month periods of 2008 and 2007 included the following (revenues in thousands):

	Second Quarter								
		200	8	2007					
	Lots	Lots Revenues		Lots	Reven	ues			
Residential Properties:									
Barton Creek									
Calera Court Courtyard Homes		1	\$	635	-	\$	-		
Calera Drive		-		-	2		809		
Mirador Estate		-		-	2		1,559		
Wimberly Lane Phase II									
Standard Homebuilder		-		-	3		522		
Circle C									
Meridian		22		1,723	20		1,423		
Deerfield		-		-	15		1,004		
Total Residential		23	\$	2,358	42	\$	5,317		

		Six Months									
	2	008	2007								
	Lots	Revenues		Lots	Revenues						
Residential Properties:											
Barton Creek											
Calera Court Courtyard Homes	1	\$	635	-	\$ -						
Calera Drive	-		-	2	809						
Mirador Estate	-		-	2	1,559						
Wimberly Lane Phase II											
Standard Homebuilder	1		265ª	6	1,045						
Circle C											
Meridian	55		3,952	48	3,239						
Deerfield	21		1,410	30	2,008						
Total Residential	78	\$	6,262	88	\$ 8,660						

a. Includes \$0.1 million for homebuilder contract termination fee.

Undeveloped Property Sales. We sold a five-acre tract at Circle C for \$1.1 million during the first six months of 2007.

*Cost of Sales.* Cost of sales totaled \$2.7 million for the second quarter of 2008, \$3.4 million for the second quarter of 2007, \$6.0 million for the first six months of 2008 and \$5.1 million for first six months of 2007, which include ongoing project and marketing costs that are relatively fixed. In addition, most of the sales for the 2008 periods were Circle C lots, which have lower profit margins than Barton Creek lots, therefore resulting in higher costs per unit for the 2008 periods.

In addition, cost of sales for the 2008 periods included \$0.1 million related to the development of the W Austin Hotel & Residences and \$0.4 million related to costs incurred for our proposal for the right to develop a new project in downtown Austin, which was awarded to another contractor. Cost of sales also included reductions for Barton Creek Municipal Utility District (MUD) reimbursements totaling \$0.1 million for the second quarter of 2007, \$0.1 million for the first six months of 2008 and \$1.7 million for the first six months of 2007.

We are projecting fewer lot sales in the next several quarters because of the recent weakness in the United States real estate market.

*General and Administrative Expenses.* General and administrative expenses decreased to \$3.1 million for the first six months of 2008, compared with \$3.3 million for the first six months of 2007, primarily because of lower stock-based compensation costs in 2008.

#### **Commercial Leasing**

Our commercial leasing operating results primarily reflect the activities at 7500 Rialto. As of June 30, 2008, the original office building was 97 percent leased and the second building was approximately 94 percent leased. Rental income increased in the 2008 periods, compared to the 2007 periods, primarily because of an approximate 50 percent increase in the second office building's occupancy from the June 30, 2007 occupancy. In addition, rental income for the first six months of 2008 includes \$0.3 million from the Barton Creek Village, which includes a retail building completed in the second quarter of 2007 and a bank building completed in early 2008.

Summary commercial leasing operating results follow (in thousands):

		Second Quarter			Six Months				
	2008 2007 200		2007		2008 2007		2008		2007
Rental income	\$	1,169	\$ 711	\$	2,120	\$	1,381		
Rental property costs		(923)	(763	)	(1,739)		(1,531)		
Depreciation		(346)	(168	)	(683)		(413)		
General and administrative expenses		(266)	(241	)	(498)		(506)		
Operating loss	\$	(366)	\$ (461	) \$	(800)	\$	(1,069)		

#### **Non-Operating Results**

Interest Expense, Net. Interest expense, net of capitalized interest, totaled \$0.3 million in the second quarter of 2008 and \$0.7 million in the first six months of 2008, compared with less than \$0.1 million in the 2007 periods. The increase in net interest expense in the 2008 periods primarily reflects interest on our higher average debt balance during 2008.

*Interest Income.* Interest income totaled \$0.2 million in the second quarter of 2008, compared with less than \$0.1 million in the second quarter of 2007, and \$1.1 million in the first six months of 2008, compared with \$0.5 million in the first six months of 2007. The increase in interest income primarily reflects interest on our higher average cash balance during 2008. Interest income included interest on Barton Creek MUD reimbursements totaling \$0.6 million in the first six months of 2007.

*Equity in Unconsolidated Affiliate's Income.* We account for our 50 percent interest in our unconsolidated affiliate, Crestview Station, using the equity method. Crestview Station sold its multi-family and commercial properties in the first quarter of 2008, which resulted in our equity in Crestview Station's earnings totaling \$0.8 million for the first six months of 2008.

Benefit from (Provision for) Income Taxes. We had income tax benefit of \$0.4 million for the second quarter of 2008 and \$0.3 million for the first six months of 2008, and income tax provision of \$0.2 million for the second quarter of 2007 and \$0.6 million for the first six months of 2007. The differences between our consolidated effective income tax rates of approximately 23 percent for the second quarter and approximately 20 percent for the first six months of 2008, and the U.S. federal statutory rate of 35 percent primarily was attributable to the current periods' losses, which resulted in nondeductible permanent items and state income taxes having a greater impact on the effective tax rate.

## DISCONTINUED OPERATIONS

On October 12, 2007, we sold the Escarpment Village shopping center, located in Austin, Texas, to Lake Villa, L.L.C. (the Purchaser) for \$46.5 million, before closing costs and other adjustments. The Purchaser paid approximately \$23.0 million in cash at closing and assumed the \$22.4 million principal balance remaining under our loan from Teachers Insurance and Annuity Association of America.

Upon completion of the sale of Escarpment Village, we ceased all involvement with the Escarpment Village shopping center. The results of operations of Escarpment Village, which have been classified as discontinued operations in the consolidated statements of income, previously represented a component of our commercial leasing segment. We earned rental income from Escarpment Village of \$0.9 million in the three months ended June 30, 2007 and \$1.8 in the six months ended June 30, 2007.

In June 2008, we revised the amount of Texas Margin Tax accrued on Escarpment Village income earned during 2007. The revised accrual resulted in \$0.1 million additional tax expense related to 2007, which has been recognized in June 2008. As the results of operations of Escarpment Village have been appropriately classified as discontinued operations, the additional Texas Margin Tax has also been classified as discontinued operations in the accompanying consolidated statements of operations. Our discontinued operations generated a net loss of \$0.2 million in each of the three-month and sixmonth periods ended June 30, 2007.

#### CAPITAL RESOURCES AND LIQUIDITY

#### Comparison of Six-Months 2008 and 2007 Cash Flows

Cash used in operating activities totaled \$2.4 million during the first six months of 2008 and cash provided by operating activities totaled \$5.6 million during the first six months of 2007, including cash used in discontinued operations totaling \$0.1 million during the 2007 period. Compared to the 2007 period, operating cash flows in the first six months of 2008 decreased primarily because of a net loss, partly offset by a \$1.3 million distribution of income from our unconsolidated affiliate, Crestview Station.

Cash used in investing activities totaled \$13.3 million during the first six months of 2008 and \$14.9 million during the first six months of 2007, including \$0.1 million used in discontinued operations. Real estate development expenditures for the first six months of 2008 and 2007 included development costs for properties in the Barton Creek, Lantana and Circle C communities. The first six months of 2008 also included real estate development expenditures for the W Austin Hotel & Residences project and expenditures for commercial leasing properties primarily related to Barton Creek Village. We received Barton Creek MUD reimbursements totaling \$3.8 million in the first six months of 2008 and \$2.6 million in the first six months of 2008. For the second half of 2008, we estimate that our capital expenditures will total approximately \$23 million, including approximately \$20 million of expenditures for the W Austin Hotel & Residences. Per the terms of the operating agreement, we expect the estimated second-half 2008 capital expenditures for the W Austin Hotel & Residences project to be funded by our joint venture partner Canyon-Johnson. We plan to fund the estimated second-half 2008 capital expenditures for our other properties with available cash or borrowings from our \$45.0 million received facility.

Cash provided by financing activities totaled \$9.5 million during the first six months of 2008, which included \$10.7 million of contributions from Canyon-Johnson for the W Austin Hotel & Residences project and \$2.0 million in borrowings from the W Austin Hotel & Residences project construction loan, partly offset by \$2.8 million of financing costs for the W Austin Hotel & Residences project construction loan. In the first six months of 2008, we used \$0.4 million to repurchase shares of our common stock on the open market (see below). Financing activities provided cash of \$12.0 million during the first six months of 2007, which included \$3.0 million of net repayments on our revolving line of credit and \$15.0 million of borrowings under three unsecured term loans, which was primarily used to fund our development activities. In the first six months of 2007, we used \$0.2 million to repurchase shares of our common stock.

In 2001, our Board of Directors approved an open market share purchase program for up to 0.7 million shares of our common stock. During the first six months of 2008, we purchased 15,385 shares for \$0.4 million, a \$27.81 per share average. During the third quarter of 2008 through July 31, 2008, we purchased 1,570 shares for \$35,000, a \$22.29 per share average. A total of 407,406 shares remain available under this program. Our loan agreement with Comerica provides a limit of \$6.5 million for common stock purchases after September 30, 2005 of which \$4.0 million is currently available. The timing of future purchases of our common stock is dependent on many factors including the price of our common shares, our cash flows and financial position, and general economic and market conditions.

#### Credit Facility and Other Financing Arrangements

At June 30, 2008, we had total debt of \$63.4 million, including \$0.3 million of current debt, and total cash and cash equivalents of \$34.7 million. Our debt outstanding at June 30, 2008 consisted of the following:

- \$40.0 million of borrowings outstanding under seven unsecured term loans, including two \$5.0 million loans, two \$8.0 million loans, a \$7.0 million loan and two \$3.5 million loans, all of which will mature in December 2011.
- \$21.4 million of borrowings outstanding under the Lantana Promissory Note, which matures in January 2018.
- \$2.0 million of borrowings outstanding under the W Austin Hotel & Residences project construction loan, which matures in September 2011.

Effective May 30, 2008, we entered into a third modification and extension agreement to extend the maturity and modify the interest rate on our \$45.0 million revolving credit facility. The maturity was extended from May 30, 2009, to May 30, 2010. In addition, the interest rate applicable to amounts borrowed under the facility was modified to an annual rate of either the base rate minus 0.45 percent with a minimum interest rate of 5 percent or the London Interbank Offered Rate plus 2 percent with a minimum interest rate of 5 percent. No amounts were outstanding under this facility at June 30, 2008. For a further discussion of our debt see Note 4 of our 2007 Form 10-K.

## NEW ACCOUNTING STANDARDS

*Fair Value Measurements.* In September 2006, the Financial Accounting Standards Board (FASB) issued Statement of Financial Accounting Standards (SFAS) No. 157, "Fair Value Measurements," which provides enhanced guidance for using fair value to measure assets and liabilities. SFAS No. 157 does not require any new fair value measurements under U.S. GAAP but rather establishes a common definition of fair value, provides a framework for measuring fair value under U.S. GAAP and expands disclosure requirements about fair value measurements. In February 2008, the FASB issued FSP FAS 157-2, which delays the effective date of SFAS No. 157 for nonfinancial assets or liabilities that are not required or permitted to be measured at fair value on a recurring basis to fiscal years beginning after November 15, 2008, and interim periods within those years. Effective January 1, 2008, we adopted SFAS No. 157 for financial assets and liabilities recognized at fair value on a recurring basis. This partial adoption of SFAS No. 157 did not impact our financial assets or liabilities subject to fair value measurement on a recurring basis. We are currently evaluating the impact that the adoption of SFAS No. 157 for nonfinancial assets or liabilities that are not required or permitted to be measured at fair value on a recurring basis. We are currently evaluating the impact that the adoption of SFAS No. 157 for nonfinancial assets or liabilities subject to fair value measurement on a recurring basis. We are currently evaluating the impact that the adoption of SFAS No. 157 for nonfinancial assets or liabilities that are not required or permitted to be measured at fair value on a recurring basis will have on our financial reporting and disclosures.

The Fair Value Option for Financial Assets and Liabilities. In February 2007, the FASB issued SFAS No. 159, "The Fair Value Option for Financial Assets and Liabilities – Including an amendment of FASB Statement No. 115," which permits entities to choose to measure many financial instruments and certain other items at fair value that are not currently required to be measured at fair value. Effective January 1, 2008, we adopted SFAS No. 159, which did not impact our financial reporting and disclosures.

<u>Noncontrolling Interests in Consolidated Financial Statements</u>. In December 2007, the FASB issued SFAS No. 160, "Noncontrolling Interests in Consolidated Financial Statements – an amendment of ARB No. 51," which clarifies that noncontrolling interests (minority interests) are to be treated as a separate component of equity and any changes in the ownership interest (in which control is retained) are to be accounted for as capital transactions. However, a change in ownership of a consolidated subsidiary that results in a loss of control is considered a significant event that triggers gain or loss recognition, with the establishment of a new fair value basis in any remaining ownership interests. SFAS No. 160 also provides additional disclosure requirements for each reporting period. SFAS No. 160 applies to fiscal years beginning on or after December 15, 2008, with early adoption prohibited. This statement is required to be adopted prospectively, except for the following provisions, which are expected to be applied retrospectively: (i) the reclassification of noncontrolling interests to equity in the consolidated balance sheets and (ii) the adjustment to consolidated net income to include net income attributable to both the controlling and noncontrolling interests. We are currently evaluating the impact that the adoption of SFAS No. 160 will have on our financial reporting and disclosures.



<u>The Hierarchy of Generally Accepted Accounting Principles.</u> In May 2008, FASB issued SFAS No. 162, "The Hierarchy of Generally Accepted Accounting Principles," which identifies the sources of accounting and the framework for selecting the principles to be used in the preparation of financial statements of nongovernmental entities that are presented in conformity with U.S. GAAP. SFAS No. 162 is effective 60 days following the SEC's approval of the Public Company Accounting Oversight Board amendments to AU Section 411, "The Meaning of Presenting Fairly in Conformity with Generally Accepted Accounting Principles." The adoption of SFAS No.162 is not expected to result in a change in our current accounting practices.

## CAUTIONARY STATEMENT

Management's Discussion and Analysis of Financial Condition and Results of Operation and Disclosures about Market Risks contains forward-looking statements regarding future reimbursements for infrastructure costs, future events related to financing and regulatory matters, projected capital expenditures, the expected results of our business strategy, and other plans and objectives of management for future operations and activities. Important factors that could cause actual results to differ materially from our expectations include economic and business conditions, business opportunities that may be presented to and pursued by us, changes in laws or regulations and other factors, many of which are beyond our control, and other factors that are described in more detail under "Risk Factors" located in our 2007 Form 10-K.

#### Item 3. Quantitative and Qualitative Disclosures about Market Risk.

There have been no significant changes in our market risks since the year ended December 31, 2007. For more information, please read the consolidated financial statements and notes thereto included in our Annual Report on Form 10-K for the year ended December 31, 2007.

#### Item 4. Controls and Procedures.

(a) <u>Evaluation of disclosure controls and procedures</u>. Our chief executive officer and chief financial officer, with the participation of management, have evaluated the effectiveness of our "disclosure controls and procedures" (as defined in Rules 13a-15(e) and 15(d)-15(e) under the Securities Exchange Act of 1934) as of the end of the period covered by this quarterly report on Form 10-Q. Based on their evaluation, they have concluded that our disclosure controls and procedures are effective as of the end of the period covered by this report.

(b) <u>Changes in internal controls</u>. There has been no change in our internal control over financial reporting that occurred during the second quarter that has materially affected, or is reasonably likely to materially affect our internal control over financial reporting.

#### PART II. - OTHER INFORMATION

## Item 2. Unregistered Sales of Equity Securities and Use of Proceeds.

The following table sets forth shares of our common stock we repurchased during the three-month period ended June 30, 2008.

<b>D</b> : 1	(a) Total Number of Shares	(b) Average Price Paid Per Share		(c) Total Number of Shares Purchased as Part of Publicly Announced	(d) Maximum Number of Shares That May Yet Be Purchased Under
Period	Purchased	Per Share		Plans or Programs <sup>a</sup>	the Plans or Programs <sup>a</sup>
April 1 to 30, 2008	13,260b	\$	27.24 <sup>b</sup>	3,354	412,432
May 1 to 31, 2008	1,656		25.56	1,656	410,776
June 1 to 30, 2008	1,800		23.74	1,800	408,976
Total	16,716	\$	26.70	6,810	

a. In February 2001, our Board of Directors approved an open market share purchase program for up to 0.7 million shares of our common stock. The program does not have an expiration date. Our loan agreement with Comerica provides a limit of \$6.5 million for common stock purchases after September 30, 2005. At June 30, 2008, \$4.0 million remained under the Comerica agreement for purchases of common stock.

b. Includes 9,906 shares (\$27.28 per share) tendered to Stratus to cover the cost of option exercises under the applicable stock incentive plans.

 Table of Contents

 Item 6. Exhibits.

 The exhibits to this report are listed in the Exhibit Index beginning on page E-1 hereof.

# SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

# STRATUS PROPERTIES INC.

By: <u>/s/ John E. Baker</u> John E. Baker Senior Vice President and Chief Financial Officer (authorized signatory and Principal Financial Officer)

Date:

August 11, 2008

# STRATUS PROPERTIES INC. EXHIBIT INDEX

Exhibit		Filed with this		Incorporated by R	eference
Number	Exhibit Title	Form 10-Q	Form	File No.	Date Filed
3.1	Amended and Restated Certificate of Incorporation of Stratus.		10-Q	000-19989	05/17/2004
3.2	Certificate of Amendment to the Amended and Restated Certificate of Incorporation of Stratus, dated May 14, 1998.		10-Q	000-19989	05/17/2004
3.3	Certificate of Amendment to the Amended and Restated Certificate of Incorporation of Stratus, dated May 25, 2001.		10-K	000-19989	03/22/2002
<u>3.4</u>	By-laws of Stratus, as amended as of November 6, 2007.	Х			
4.1	Rights Agreement dated as of May 16, 2002, between Stratus and Mellon Investor Services LLP, as Rights Agent, which includes the Certificates of Designation of Series C Participating Preferred Stock; the Forms of Rights Certificate Assignment, and Election to Purchase; and the Summary of Rights to Purchase Preferred Shares.		8-A	000-19989	05/23/2002
4.2	Amendment No. 1 to Rights Agreement between Stratus Properties Inc. and Mellon Investor Services LLC, as Rights Agent, dated as of November 7, 2003.		8-K	000-19989	11/14/2003
10.1	Third Modification and Extension Agreement by and between Stratus Properties Inc., Stratus Properties Operating Co., L.P., Circle C Land, L.P., Austin 290 Properties, Inc., Calera Court, L.P., Oly Stratus Barton Creek I Joint Venture and Comerica Bank effective May 30, 2008.		8-K	000-19989	07/17/2008
10.2	Second Modification and Extension Agreement by and between Stratus Properties Inc., Stratus Properties Operating Co., L.P., Circle C Land, L.P., Austin 290 Properties, Inc., Calera Court, L.P., and Comerica Bank effective May 30, 2007.		8-K	000-19989	02/08/2008
10.3	Loan Agreement by and between Stratus Properties Inc., Stratus Properties Operating Co., L.P., Circle C Land, L.P., Austin 290 Properties, Inc., Calera Court, L.P., and Comerica Bank dated as of September 30, 2005.		8-K	000-19989	10/05/2005
<u>10.4</u>	Construction Loan Agreement dated May 2, 2008, by and between CJUF II Stratus Block 21 LLC and Corus Bank, N.A.	Х			
<u>10.5</u>	Promissory Note dated May 2, 2008, by and between CJUF II Stratus Block 21 LLC and Corus Bank, N.A.	Х			
10.6	Revolving Promissory Note by and between Stratus Properties Inc., Stratus Properties Operating Co., L.P., Circle C Land, L.P., Austin 290 Properties, Inc., Calera Court, L.P., and Comerica Bank dated as of September 30, 2005.		8-K	000-19989	10/05/2005
10.7	Loan Agreement dated December 28, 2000, by and between Stratus Properties Inc. and Holliday Fenoglio Fowler, L.P., subsequently assigned to an affiliate of First American Asset Management.		10-K	000-19989	03/28/2001

Exhibit		Filed with this		Incorporated by R	eference
Number		Form 10-Q	Form	File No.	Date Filed
10.8	Loan Agreement dated June 14, 2001, by and between Stratus Properties Inc. and Holliday Fenoglio Fowler, L.P., subsequently assigned to an affiliate of First American Asset Management.		10-Q	000-19989	11/13/2001
10.9	Construction Loan Agreement dated June 11, 2001, between 7500 Rialto Boulevard, L.P. and Comerica Bank-Texas.		10 <b>-</b> K	000-19989	03/22/2002
10.10	Modification Agreement dated January 31, 2003, by and between Lantana Office Properties I, L.P., formerly 7500 Rialto Boulevard, L.P., and Comerica Bank-Texas.		10-Q	000-19989	05/15/2003
10.11	Second Modification Agreement dated as of December 29, 2003, to be effective as of January 31, 2004, by and between Lantana Office Properties I, L.P., a Texas limited partnership (formerly known as 7500 Rialto Boulevard, L.P.), as borrower, and Comerica Bank, as lender.		10-К	000-19989	3/30/2004
10.12	Guaranty Agreement dated June 11, 2001, by Stratus Properties Inc. in favor of Comerica Bank-Texas.		10-K	000-19989	03/22/2002
10.13	Loan Agreement dated September 22, 2003, by and between Calera Court, L.P., as borrower, and Comerica Bank, as lender.		10-Q	000-19989	11/14/2003
10.14	Development Agreement dated August 15, 2002, between Circle C Land Corp. and City of Austin.		10-Q	000-19989	11/14/2002
10.15	First Modification Agreement dated March 27, 2006, by and between Stratus 7000 West Joint Venture, as Old Borrower, and CarrAmerica Lantana, LP, as New Borrower, and Teachers Insurance and Annuity Association of America, as Lender.		8-K	000-19989	03/29/2006
10.16	Agreement of Sale and Purchase dated November 23, 2005, by and between Stratus Properties Operating Co., L.P., as Seller, and Advanced Micro Devices, Inc., as Purchaser.		10-Q	000-19989	05/10/2006
10.17	First Amendment to Agreement of Sale and Purchase dated April 26, 2006, by and between Stratus Properties Operating Co., L.P., as Seller, and Advanced Micro Devices, Inc., as Purchaser.		10-Q	000-19989	05/10/2006
10.18	Deed of Trust, Assignment of Leases and Rents, Security Agreement and Fixture Filing dated as of June 30, 2006, by and among Escarpment Village, L.P. and Teachers Insurance and Annuity Association of America.		10-Q	000-19989	08/09/2006
10.19	Promissory Note dated as of June 30, 2006, by and between Escarpment Village, L.P. and Teachers Insurance and Annuity Association of America.		10-Q	000-19989	08/09/2006
10.20	Amended and Restated Loan Agreement between Stratus Properties Inc. and American Strategic Income Portfolio IncII dated as of December 12, 2006.		10-K	000-19989	03/16/2007
10.21	Amended and Restated Loan Agreement between Stratus Properties Inc. and American Select Portfolio Inc. dated as of December 12, 2006.		10-K	000-19989	03/16/2007

Exhibit	ontents	Filed with this	s Incorporated by Reference					
Number	<b>Exhibit</b> Title	Form 10-Q	Form	File No.	Date Filed			
10.22	Loan Agreement between Stratus Properties Inc. and Holliday Fenoglio Fowler, L.P. dated as of December 12, 2006.		10-K	000-19989	03/16/2007			
10.23	Loan Agreement between Stratus Properties Inc. and Holliday Fenoglio Fowler, L.P. dated as of December 12, 2006.		10-K	000-19989	03/16/2007			
10.24	Letter Agreement between Stratus Properties Inc. and Canyon-Johnson Urban Fund II, L.P., dated as of May 4, 2007.		10-Q	000-19989	08/09/2007			
10.25	Loan Agreement between Stratus Properties Inc. and Holliday Fenoglio Fowler, L.P. dated as of June 1, 2007, subsequently assigned to American Select Portfolio Inc., an affiliate of First American Asset Management.		10-Q	000-19989	08/09/2007			
10.26	Loan Agreement between Stratus Properties Inc. and Holliday Fenoglio Fowler, L.P. dated as of June 1, 2007, subsequently assigned to American Strategic Income Portfolio Inc., an affiliate of First American Asset Management.		10-Q	000-19989	08/09/2007			
10.27	Loan Agreement between Stratus Properties Inc. and Holliday Fenoglio Fowler, L.P. dated as of June 1, 2007, subsequently assigned to American Strategic Income Portfolio IncIII, an affiliate of First American Asset Management.		10-Q	000-19989	08/09/2007			
10.28	Purchase and Sale Agreement dated as of July 9, 2007, between Escarpment Village, L.P. as Seller and Christopher Investment Company, Inc. as Purchaser.		8-K	000-19989	10/18/2007			
10.29	Promissory Note dated as of December 14, 2007, between Lantana Office Properties I, L.P., as borrower, and The Lincoln National Life Insurance Company, as lender.		8-K	000-19989	12/14/2007			
10.30*	Stratus' Performance Incentive Awards Program, as amended, effective February 11, 1999.		10-Q	000-19989	05/17/2004			
10.31*	Stratus Properties Inc. Stock Option Plan, as amended and restated.		10-Q	000-19989	05/10/2007			
10.32*	Stratus Properties Inc. 1996 Stock Option Plan for Non-Employee Directors, as amended and restated.		10-Q	000-19989	05/10/2007			
10.33*	Stratus Properties Inc. 1998 Stock Option Plan, as amended and restated.		10-Q	000-19989	05/10/2007			
10.34*	Form of Notice of Grant of Nonqualified Stock Options under the 1998 Stock Option Plan.		10-Q	000-19989	8/12/2005			
10.35*	Form of Restricted Stock Unit Agreement under the 1998 Stock Option Plan.		10-Q	000-19989	05/10/2007			
10.36*	Stratus Properties Inc. 2002 Stock Incentive Plan, as amended and restated.		10-Q	000-19989	05/10/2007			
10.37*	Form of Notice of Grant of Nonqualified Stock Options under the 2002 Stock Incentive Plan.		10-Q	000-19989	08/12/2005			
	E-3							

Exhibit		Filed with this		Incorporated by R	eference
Number	Exhibit Title	Form 10-Q	Form	File No.	Date Filed
10.38*	Form of Restricted Stock Unit Agreement under the 2002 Stock Incentive Plan.		10-Q	000-19989	05/10/2007
10.39*	Stratus Director Compensation.		10 <b>-</b> K	000-19989	03/16/2006
10.40*	Change of Control Agreement between Stratus Properties Inc. and William H. Armstrong III, effective as of January 26, 2007.		8-K	000-19989	01/30/2007
10.41*	Change of Control Agreement between Stratus Properties Inc. and John E. Baker, effective as of January 26, 2007.		10-K	000-19989	01/30/2007
<u>15.1</u>	Letter from PricewaterhouseCoopers LLP regarding the unaudited interim financial statements.	Х			
<u>31.1</u>	Certification of Principal Executive Officer pursuant to Rule 13a-14(a)/15d-14(a).	Х			
<u>31.2</u>	Certification of Principal Financial Officer pursuant to Rule $13a-14(a)/15d-14(a)$ .	х			
<u>32.1</u>	Certification of Principal Executive Officer pursuant to 18 U.S.C. Section 1350.	Х			
<u>32.2</u>	Certification of Principal Financial Officer pursuant to 18 U.S.C. Section 1350.	Х			

Note: Certain instruments with respect to long-term debt of Stratus have not been filed as exhibits to this Quarterly Report on Form 10-Q since the total amount of securities authorized under any such instrument does not exceed 10 percent of the total assets of Stratus and its subsidiaries on a consolidated basis. Stratus agrees to furnish a copy of each such instrument upon request of the Securities and Exchange Commission.

\* Indicates management contract or compensatory plan or arrangement.

E-4

## Exhibit 3.4

## **Stratus Properties Inc.** (formerly FM Properties Inc.)

# **By-Laws**

# ARTICLE I

Name

The name of the corporation is Stratus Properties Inc.

#### ARTICLE II

#### Offices

1. The location of the registered office of the corporation in the State of Delaware is 1209 Orange Street, in the City of Wilmington, County of New Castle, and the name of its registered agent at such address is The Corporation Trust Company.

2. The corporation shall in addition to its registered office in the State of Delaware establish and maintain an office or offices at such place or places as the Board of Directors may from time to time find necessary or desirable.

# ARTICLE III

#### **Corporate Seal**

The corporate seal of the corporation shall have inscribed thereon the name of the corporation and the year (1992) and jurisdiction (Delaware) of its creation. Such seal may be used by causing it or a facsimile thereof to be impressed, affixed, printed or otherwise reproduced.

#### ARTICLE IV

#### Meetings of Stockholders

1. All meetings of the stockholders shall be held at the registered office of the corporation in the State of Delaware, or at any other place as shall be determined, from time to time, by the Board of Directors.

2. The first annual meeting of stockholders shall be held on Monday, May 17, 1993, at eleven o'clock in the forenoon, or on such other date in that year or at such other time as may be determined by resolution of the Board of Directors. In subsequent years the annual meeting of the stockholders shall be held on the Monday immediately preceding the third Tuesday of May at eleven o'clock in the forenoon, or on such other day or at such other time as may be determined from time to time by resolution of the Board of Directors. At each annual meeting of the stockholders they shall elect by plurality vote, by written ballot, the successors of the class of directors whose term expires at such meeting, to hold office until the annual meeting of the stockholders held in the third year following the year of their election and their successors are respectively elected and qualified or until their earlier resignation or removal. Any other proper business may be transacted at the annual meeting.

3. The holders of a majority of the stock issued and outstanding and entitled to vote thereat, present in person or represented by proxy, shall constitute a quorum at all meetings of the stockholders for the transaction of business, except as otherwise expressly provided by statute, by the Certificate of Incorporation or by these By-Laws. If, however, such majority shall not be present or represented at any meeting of the stockholders, the stockholders entitled to vote thereat, present in person or by proxy, shall have power to adjourn the meeting from time to time, without notice other than announcement at the meeting (except as otherwise provided by statute), until the requisite amount of voting stock shall be present. At such adjourned meeting at which the requisite amount of voting stock shall be represented, any business may be transacted which might have been transacted at the meeting as originally notified.

4. At all meetings of the stockholders each stockholder having the right to vote shall be entitled to vote in person or by proxy appointed by an instrument in writing subscribed by such stockholder and bearing a date not more than three years prior to said meeting, unless such instrument provides for a longer period. All proxies shall be filed with the secretary of the meeting before being voted.

5. At each meeting of the stockholders each stockholder shall have one vote, unless otherwise provided in the Certificate of Incorporation, for each share of stock of the corporation having voting power, registered in his name on the books of the corporation at the record date fixed in accordance with these By-Laws, or otherwise determined, with respect to such meeting. Except as otherwise expressly provided by statute, by the Certificate of Incorporation or by these By-Laws, all matters coming before any meeting of the stockholders shall be decided by the vote of a majority of the number of shares of stock present in person or represented by proxy at such meeting and entitled to vote thereat, a quorum being present.

6. Notice of each meeting of the stockholders shall be mailed to each stockholder entitled to vote thereat not less than 10 nor more than 60 days before the date of the meeting. Such notice shall state the place, date and hour of the meeting and, in the case of a special meeting, the purpose or purposes for which the meeting is called.

7. Subject to such rights to call special meetings of stockholders under specified circumstances as may be granted to holders of any shares of Preferred Stock pursuant to the Certificate of Incorporation, special meetings of the stockholders may be called only by the Chairman of the

Board or the President of the corporation, or at the request in writing or by vote of a majority of the Board of Directors, and not by any other persons. Any request for a special meeting made by the Board of Directors shall state the purpose or purposes of the proposed meeting.

- 8. Business transacted at each special meeting shall be confined to the purpose or purposes stated in the notice of such meeting.
- 9. The order of business at each meeting of the stockholders shall be determined by the chairman at such meeting.

10 At an annual meeting of the stockholders, only business shall be conducted as shall have been brought before the meeting (a) by or at the direction of the Board of Directors or (b) by any stockholder of the corporation who complies with the notice procedures set forth in this Section 10. For business to be properly brought before an annual meeting by a stockholder, the stockholder must have given timely notice thereof in writing to the Secretary of the corporation. To be timely, a stockholder's notice must be delivered to the Secretary at the principal executive offices of the corporation not later than the close of business on the 120th day nor earlier than the close of business on the 210th day prior to the first anniversary of the preceding year's annual meeting; provided, however, that in the event that the date of the annual meeting is more than 30 days before or more than 90 days after such anniversary date, notice by the stockholder to be timely must be so delivered not earlier than the close of business on the 120th day prior to such annual meeting and not later than the close of business on the later of the 90th day prior to such annual meeting or the 10th day following the day on which public announcement of the date of such meeting is first made. In no event shall the public announcement of an adjournment of an annual meeting commence a new time period for the giving of a stockholder's notice as described above. A stockholder's notice to the Secretary shall set forth as to each matter the stockholder proposes to bring before the annual meeting (a) a brief description of the business desired to be brought before the annual meeting and the reasons for conducting such business at the annual meeting, (b) the name and address, as they appear on the corporation's books, of the stockholder proposing such business, (c) the class and number of shares of the corporation which are beneficially owned by the stockholder and (d) any material interest of the stockholder in such business. Notwithstanding anything in these By-Laws to the contrary, no business shall be conducted at an annual meeting except in accordance with the procedures set forth in this Section 10. The chairman of an annual meeting shall, if the facts warrant, determine and declare to the meeting that business was not properly brought before the meeting and in accordance with the provisions of these By-Laws, and if he should so determine, he shall so declare to the meeting and any such business not properly brought before the meeting shall not be transacted. Notwithstanding the foregoing provisions of this Section 10, a stockholder seeking to have a proposal included in the corporation's proxy statement shall comply with the requirements of Regulation 14A under the Securities Exchange Act of 1934, as amended (including, but not limited to, Rule 14a-8 or its successor provision).

11. Only persons who are nominated in accordance with the procedures set forth in these By-Laws shall be eligible for election as directors. Nominations of persons for election to the Board of Directors of the corporation may be made at a meeting of stockholders (a) by or at the direction of the Board of Directors or (b) by any stockholder of the corporation entitled to vote for the election of directors at the meeting who complies with the notice procedures set forth in this Section

11. Such nominations, other than those made by or at the direction of the Board of Directors, shall be made pursuant to timely notice in writing to the Secretary of the corporation. To be timely, a stockholder's notice must be delivered to the Secretary at the principal executive offices of the corporation not later than the close of business on the 120th day nor earlier than the close of business on the 210th day prior to the first anniversary of the preceding year's annual meeting; provided, however, that in the event that the date of the annual meeting is more than 30 days before or more than 90 days after such anniversary date, notice by the stockholder to be timely must be so delivered not earlier than the close of business on the 120th day prior to such annual meeting and not later than the close of business on the later of the 90th day prior to such annual meeting or the 10th day following the day on which public announcement of the date of such meeting is first made. In no event shall the public announcement of an adjournment of an annual meeting commence a new time period for the giving of a stockholder's notice as described above. Such stockholder's notice shall set forth (a) as to each person whom the stockholder proposes to nominate for election or reelection as a director all information relating to such person that is required to be disclosed in solicitations of proxies for election of directors, or is otherwise required, in each case pursuant to Regulation 14A under the Securities Exchange Act of 1934, as amended (including such person's written consent to being named in the proxy statement as a nominee and to serve as a director if elected); and (b) as to the stockholder giving the notice (i) the name and address, as they appear on the corporation's books, of such stockholder and (ii) the class and number of shares of the corporation which are beneficially owned by such stockholder. At the request of the Board of Directors any person nominated by the Board of Directors for election as a director shall furnish to the Secretary of the corporation that information required to be set forth in a stockholder's notice of nomination which pertains to the nominee. No person shall be eligible for election as a director of the corporation unless nominated in accordance with the procedures set forth in these By-Laws. The chairman of the meeting shall, if the facts warrant, determine and declare to the meeting that a nomination was not made in accordance with the procedures prescribed by these By-Laws, and if he should so determine, he shall so declare to the meeting and the defective nomination shall be disregarded.

12. Any action required or permitted to be taken by the stockholders of the corporation must be effected at a duly called annual or special meeting of such holders and may not be effected by any consent in writing by such holders.

#### ARTICLE V

#### Directors

1. The business and affairs of the corporation shall be managed under the direction of a Board of Directors which may exercise all such powers and authority for and on behalf of the corporation as shall be permitted by law, the Certificate of Incorporation or these By-Laws.

2. The directors may hold their meetings and have one or more offices, and, subject to the laws of the State of Delaware, keep the stock ledger and other books and records of the corporation, outside said State, at such place or places as they may from time to time determine.

3. Subject to such rights to elect additional directors under specified circumstances as may be granted to the holders of any shares of the Preferred Stock pursuant to the Certificate of Incorporation, the number of directors of the corporation shall be fixed from time to time by the Board of Directors but shall not be less than three. The directors, other than those who may be elected by the holders of any class or series of Preferred Stock, shall be classified, with respect to the time for which they severally hold office, into three classes, designated Class I, Class II and Class III, as nearly equal in number as possible, as determined by the Board of Directors, Class I directors to hold office initially for a term expiring at the annual meeting of stockholders to be held in 1993, Class II directors to hold office initially for a term expiring at the annual meeting of stockholders to be held in 1995, with the members of each class to hold office until their successors are elected and qualified. At each annual meeting of stockholders, the successors of the class of directors whose term expires at that meeting shall be elected to hold office for a term expiring at the annual meeting of stockholders held in the third year following the year of their election.

4. Subject to such rights to elect directors under specified circumstances as may be granted to holders of any shares of the Preferred Stock pursuant to the Certificate of Incorporation, newly created directorships resulting from any increase in the number of directors and any vacancies on the Board of Directors resulting from death, resignation, disqualification, removal or other reason shall be filled solely by the affirmative vote of a majority of the remaining directors then in office, even though less than a quorum of the Board of Directors. Any director elected in accordance with the preceding sentence shall hold office for the remainder of the full term of the class of directors in which the new directorship was created or the vacancy occurred and until such director's successor shall have been elected and qualified. No decrease in the number of directors constituting the Board of Directors shall shorten the term of any incumbent director.

5. Any director may resign at any time by giving written notice of his resignation to the Board of Directors, the Chairman of the Board or the President. Any such resignation shall take effect upon receipt thereof by the Board, the Chairman of the Board or the President, as the case may be, or at such later date as may be specified therein. Any such notice to the Board shall be addressed to it in care of the Secretary.

## ARTICLE VI

#### **Committees of Directors**

By resolution adopted by a majority of the whole Board of Directors, the Board shall designate an Executive Committee and an Audit Committee and may designate one or more other committees as the Board may deem appropriate, each such committee to consist of one or more directors of the corporation. The Executive Committee shall have and may exercise all the powers and authority of the Board in the management of the business and affairs of the corporation (except as otherwise expressly limited by statute) and may authorize the seal of the corporation to be affixed to all papers which may require it. The Audit Committee and each such other committee shall have such of the powers and authority of the Board as may be provided from time to time in resolutions adopted by

a majority of the whole Board. Each committee shall report its proceedings to the Board when required.

## ARTICLE VII

## **Compensation of Directors**

The directors shall receive such compensation for their services as may be authorized by resolution of the Board of Directors, which compensation may include an annual fee and a fixed sum and expenses for attendance at regular or special meetings of the Board or any committee thereof. Nothing herein contained shall be construed to preclude any director from serving the corporation in any other capacity and receiving compensation therefor.

## ARTICLE VIII

#### Meetings of Directors; Action Without A Meeting

1. Regular meetings of the Board of Directors may be held without notice at such time and place as may be determined from time to time by resolution of the Board.

2. Special meetings of the Board of Directors may be called by the Chairman of the Board or by the President on at least 24 hours' notice to each director, and shall be called by the President or by the Secretary on like notice on the request in writing of any director. Except as may be otherwise specifically provided by statute, by the Certificate of Incorporation or by these By-Laws, the purpose or purposes of any such special meeting need not be stated in such notice.

3. At all meetings of the Board of Directors the presence of a majority of the total number of directors shall be necessary and sufficient to constitute a quorum for the transaction of business, and, except as otherwise provided by statute, by the Certificate of Incorporation or by these By-Laws, if a quorum shall be present the act of a majority of the directors present shall be the act of the Board.

4. Any action required or permitted to be taken at any meeting of the Board of Directors or of any committee thereof may be taken without a meeting if all the members of the Board or such committee, as the case may be, consent thereto in writing and the writing or writings are filed with the minutes of proceedings of the board or committee. Any director may participate in a meeting of the Board, or of any committee designated by the Board, by means of a conference telephone or similar communications equipment by means of which all persons participating in the meeting can hear each other, and participation in a meeting pursuant to this sentence shall constitute presence in person at such meeting.

#### ARTICLE IX

#### Officers

1. The officers of the corporation shall be chosen by the Board of Directors and shall be a President, a Secretary and a Treasurer. The Board may also choose a Chairman of the Board, a General Counsel, a Controller, one or more Executive Vice Presidents, Senior Vice Presidents, Vice Presidents, Assistant Secretaries and Assistant Treasurers, and such other officers as it shall deem necessary, who shall hold their offices for such terms and shall exercise such powers and perform such duties as shall be determined from time to time by the Board. Any number of offices may be held by the same person.

2. The Board of Directors, at its first meeting after the annual meeting of stockholders, shall choose a Chairman of the Board and a President from among the directors and shall choose the remaining officers who need not be members of the Board.

3. The salaries of all officers of the corporation shall be fixed by the Board of Directors, or in such manner as the Board may prescribe.

4. The officers of the corporation shall hold office until their successors are chosen and qualified, except that any officer may at any time be removed by the Board of Directors. If the office of any officer becomes vacant for any reason, the vacancy may be filled by the Board.

5. Any officer may resign at any time by giving written notice of his resignation to the Board of Directors, the Chairman of the Board or the President. Any such resignation shall take effect upon receipt thereof by the Board, the Chairman of the Board or the President, as the case may be, or at such later date as may be specified therein. Any such notice to the Board shall be addressed to it in care of the Secretary.

## ARTICLE X

#### Chairman of the Board

The Chairman of the Board shall be the chief executive officer of the corporation and shall preside at meetings of the stockholders and of the Board of Directors. Subject to the supervision and direction of the Board of Directors, he shall be responsible for managing the affairs of the corporation. He shall have general supervision and direction of all of the other officers of the corporation and shall have powers and duties usually and customarily associated with the office of Chairman of the Board and the position of chief executive officer.

## ARTICLE XI

#### President

The President shall be the chief operating officer of the corporation, and he shall have the powers and duties usually and customarily associated with the office of the President and the position of chief operating officer. He shall have such other powers and duties as may be delegated to him by the Chairman of the Board.

# ARTICLE XII

# **Executive Vice Presidents, Senior Vice Presidents and Vice Presidents**

The Executive Vice Presidents, the Senior Vice Presidents and the Vice Presidents shall have such powers and duties as may be delegated to them by the Chairman of the Board.

#### ARTICLE XIII

#### **General Counsel, Secretary and Assistant Secretaries**

1. The General Counsel shall have the powers and duties usually and customarily associated with the position of General Counsel. He shall have such other powers and duties as may be delegated to him by the Chairman of the Board.

2. The Secretary shall attend all meetings of the Board of Directors and of the stockholders, and shall record the minutes of all proceedings in a book to be kept for that purpose. He shall perform like duties for the committees of the Board when required.

3. The Secretary shall give, or cause to be given, notice of meetings of the stockholders, of the Board of Directors and of the committees of the Board. He shall keep in safe custody the seal of the corporation, and when authorized by the Chairman of the Board, the President, an Executive Vice President, a Senior Vice President or a Vice President, shall affix the same to any instrument requiring it, and when so affixed it shall be attested by his signature of by the signature of an Assistant Secretary. He shall have such other powers and duties as may be delegated to him by the Chairman of the Board.

4. The Assistant Secretaries shall, in case of the absence of the Secretary, perform the duties and exercise the powers of the Secretary, and shall have such other powers and duties as may be delegated to them by the Chairman of the Board.

# ARTICLE XIV

#### **Treasurer and Assistant Treasurer**

1. The Treasurer shall have the custody of the corporate funds and securities, and shall deposit or cause to be deposited under his direction all moneys and other valuable effects in the name and to the credit of the corporation in such depositories as may be designated by the Board of Directors or pursuant to authority granted by it. He shall render to the President and the Board whenever they may require it an account of all his transactions as Treasurer and of the financial condition of the corporation. He shall have such other powers and duties as may be delegated to him by the Chairman of the Board.

2. The Assistant Treasurers shall, in case of the absence of the Treasurer, perform the duties and exercise the powers of the Treasurer, and shall have such other powers and duties as may be delegated to them by the Chairman of the Board.

# ARTICLE XV

#### Controller

The Controller shall maintain adequate records of all assets, liabilities and transactions of the corporation, and shall see that adequate audits thereof are currently and regularly made. He shall disburse the funds of the corporation in payment of the just obligations of the corporation, or as may be ordered by the Board of Directors, taking proper vouchers for such disbursements. He shall have such other powers and duties as may be delegated to him by the Chairman of the Board.

# ARTICLE XVI

#### **Certificates of Stock**

The shares of the corporation shall be uncertificated or shall be represented by certificates signed in the name of the corporation. The certificates for shares of stock of the corporation shall be numbered and shall be entered on the books of the corporation as they are issued. The certificated shares shall exhibit the holder's name and number of shares and shall be signed by the Chairman of the Board, the President, an Executive Vice President, a Senior Vice President or a Vice President and by the Treasurer, an Assistant Treasurer, the Secretary or an Assistant Secretary. The signature of any such officers may be facsimile if such certificate is countersigned by a transfer agent other than the corporation or its employee or by a registrar other than the corporation or its employee. In case any officer who has signed or whose facsimile signature has been placed on any such certificate shall have ceased to be such officer before such certificate is issued, then, unless the Board of Directors shall otherwise determine and cause notification thereof to be given to such transfer agent and registrar, such certificate may be issued by the corporation (and by its transfer agent) and registered by its registrar with the same effect as if he were such officer at the date of issue.

# ARTICLE XVII

#### **Transfers of Stock**

1. Upon surrender to the corporation or the transfer agent of a certificate for shares duly endorsed or accompanied by proper evidence of succession, assignment or authority to transfer, it shall be the duty of the corporation to issue a new certificate to the person entitled thereto, cancel the old certificate and record the transaction upon its books. Upon receipt of proper transfer instructions from the registered holder of uncertificated shares such uncertificated shares or certificated shares shall be made to the person entitled thereto and the transaction shall be recorded upon the books of the corporation.

2. Within a reasonable time after the issuance or transfer of uncertificated shares, the corporation shall send, or cause to be sent, to the registered owner thereof a written notice containing the information required to be set forth or stated on certificates pursuant to Sections 151, 156, 202(a) or 218(a) of the Delaware General Corporation Law or a statement that the corporation will furnish without charge to each stockholder who so requests the powers, designations, preferences and relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and rights.

#### ARTICLE XVIII

#### **Fixing Record Date**

In order that the corporation may determine the stockholders entitled to notice of and to vote at any meeting of stockholders or any adjournment thereof, or entitled to express consent in writing to any corporate action without a meeting, or entitled to receive payment of any dividend or other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock, or for any other lawful purpose, the Board of Directors may fix, in advance, a record date which shall not be more than 60 nor less than 10 days before the date of such meeting, nor more than 60 days prior to any other action. Only stockholders of record on the date so fixed shall be entitled to such notice of, and to vote at, such meeting and any adjournment thereof, or entitled to express such consent, or entitled to receive payment of such dividend or other distribution or allotment of rights, or entitled to exercise such rights in respect of change, conversion or exchange, as the case may be, notwithstanding any transfer of stock on the books of the corporation after any such record date fixed as aforesaid.

# ARTICLE XIX

# **Registered Stockholders**

The corporation shall be entitled to treat the holder of record of any share or shares of stock as the holder in fact thereof and, accordingly, shall not be bound to recognize any equitable or



other claim to or interest in such share on the part of any other person, whether or not it shall have express or other notice thereof, save as expressly provided by the laws of the State of Delaware.

### ARTICLE XX

# Checks

All checks, drafts and other orders for the payment of money and all promissory notes and other evidences of indebtedness of the corporation shall be signed by such officer or officers or such other person or persons as may be designated by the Board of Directors or pursuant to authority granted by it.

# ARTICLE XXI

#### **Fiscal Year**

The fiscal year of the corporation shall end on December 31 of each year.

#### ARTICLE XXII

#### **Notices and Waiver**

1. Whenever by statute, by the Certificate of Incorporation or by these By-Laws it is provided that notice shall be given to any director or stockholder, such provision shall not be construed to require personal notice, but such notice may be given in writing, by mail, by depositing the same in the United States mail, postage prepaid, directed to such stockholder or director at his address as it appears on the records of the corporation, or, in default of other address, to such director or stockholder at the General Post Office in the City of Wilmington, Delaware, and such notice shall be deemed to be given at the time when the same shall be thus deposited. Notice of special meetings of the Board of Directors may also be given to any director by telephone or by telex, telegraph or cable and in the latter event the notice shall be deemed to be given at the time such notice, addressed to such director at the address hereinabove provided, shall be transmitted or delivered to and accepted by an authorized telegraph or cable office.

2. Whenever by statute, by the Certificate of Incorporation or by these By-Laws a notice is required to be given, a written waiver thereof, signed by the person entitled to notice, whether before or after the time stated therein, shall be deemed equivalent to notice. Attendance of any stockholder or director at any meeting thereof shall constitute a waiver of notice of such meeting by such stockholder or director, as the case may be, except as otherwise provided by statute.

1	1
1	1

# ARTICLE XXIII

#### Alteration of By-Laws

These By-Laws, including, but not limited to, Section 7 of Article IV and Sections 3 and 4 of Article V, may be altered, amended, changed or repealed at any meeting of the Board of Directors by vote of a majority of the directors present or as otherwise provided by statute, except that, in the case of any amendment, alteration, change or repeal of Section 7 of Article IV or Section 3 or 4 of Article V by the stockholders, notwithstanding any other provision of these By-Laws, the Certificate of Incorporation or any provision of law which might otherwise permit a lesser vote or no vote, the affirmative vote of the holders of 85 percent or more of the outstanding shares of capital stock of the corporation entitled to vote generally in the election of directors shall be required to amend, alter, change or repeal such Section 7 of Article IV or such Section 3 or 4 of Article V.

#### ARTICLE XXIV

# **Indemnification of Corporate Personnel**

The corporation shall indemnify any person who is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise as provided in the Certificate of Incorporation. Expenses incurred by such a director, officer, employee or agent in defending a civil or criminal action, suit or proceeding shall be paid by the corporation as provided in the Certificate of Incorporation. The corporation shall have power to purchase and maintain insurance on behalf of any such person against any liability asserted against him and incurred by him in any such capacity, or arising out of his status as such, whether or not the corporation would have the power to indemnify him against such liability under the provisions of the Certificate of Incorporation provisions of this Article XXIV and the Certificate of Incorporation shall not be deemed exclusive of any other rights to which those seeking indemnification may be entitled under any applicable law, by-law, agreement, vote of stockholders or disinterested directors or otherwise.

The provisions of this Article XXIV and Article EIGHTH of the Certificate of Incorporation shall be deemed to be a contract between the corporation and each person who serves as such director, officer, employee or agent of the corporation in any such capacity at any time while this Article XXIV and Article EIGHTH of the Certificate of Incorporation are in effect. No repeal or modification of the provisions of this Article XXIV and Article EIGHTH of the Certificate of Incorporation nor, to the fullest extent permitted by law, any modification of law shall adversely affect any right or protection of a director, officer, employee or agent of the corporation then existing at the time of such repeal or modification. The provisions of this Article XXIV of the By-Laws of the corporation have been adopted by the stockholders of the corporation.

Exhibit 10.4

# CONSTRUCTION LOAN AGREEMENT

# MADE BY AND BETWEEN

# CJUF II STRATUS BLOCK 21 LLC c/o

Stratus Properties, Inc.

98 San Jacinto, Suite 200

Austin, Texas 78701

AND

COR<u>US</u> BANK, N.A., as Lender

3959 North Lincoln Avenue

Chicago, Illinois 60613

Dated as of May 2, 2008

# Table of Contents

Article	1 INCORPORATION OF RECITALS AND EXHIBITS	2
1.1	Incorporation of Recitals	2
1.2	Incorporation of Exhibits	2
Article	2 DEFINITIONS	2
2.1	Defined Terms	2
2.2	Other Definitional Provisions	14
Article	3 BORROWER'S REPRESENTATIONS AND WARRANTIES	14
3.1	Representations and Warranties	14
3.2	Survival of Representations and Warranties	19
Article	4 LOAN AND LOAN DOCUMENTS	19
4.1	Agreement to Borrow and Lend; Lender's Obligation to Disburse; Excess Disbursements	19
4.2	Loan Documents	21
4.3	Term of the Loan	21
4.4	Prepayments	23
4.5	Required Principal Payments	23
4.6	Receipt of Payments	23
4.7	Termination of Lender's Unfunded Commitment	23
4.8	Lender Default	24
Article	5 INTEREST	24
5.1	Interest Rate	24
Article	6 COSTS OF MAINTAINING LOAN	24
6.1	Increased Costs and Capital Adequacy	24
6.2	Borrower Withholding	25
Article	7 LOAN EXPENSE AND ADVANCES	25
7.1	Loan and Administration Expenses	25
7.2	Loan Fee	26
7.3	Draw Fees	26
7.4	Exit Fee	26
7.5	Lender's Attorneys' Fees and Disbursements	27
7.6	Time of Payment of Fees and Expenses	27
7.7	Expenses and Advances Secured by Loan Documents	27
7.8	Right of Lender to Make Advances to Cure Borrower's Defaults	28
Article	8 NON-CONSTRUCTION REQUIREMENTS PRECEDENT	28
8.1	Non-Construction Conditions Precedent	28
Article	9 CONSTRUCTION REQUIREMENTS PRECEDENT	33
9.1	Construction Documents Required as of Closing	33
9.2	Construction Deliveries Required as of Full Loan Opening.	35
Article	10 BUDGET, CONTINGENCY FUND AND CHANGE ORDERS	37
10.1	Budget	37
10.2	Budget Line Items	38

# i

<u>Page</u>

# Table of Contents

<u>Page</u>

10.3	Contingency Fund	39
10.4	Optional Method for Payment of Interest	40
10.5	Change Orders	40
	I SUFFICIENCY OF LOAN	41
11.1	Loan In Balance	41
11.2	Additional Equity Investment	43
Article 12	2 CONSTRUCTION PAYOUT REQUIREMENTS	43
12.1	Applicability of Sections	43
12.2	Monthly Payouts	43
12.3	Documents to be Furnished for Each Disbursement	44
12.4	Retainages	45
12.5	Disbursements for Materials Stored On-Site	46
12.6	Disbursements for Off-site Materials	46
12.7	Specific Limitation on Disbursements	46
12.8	Disbursements Related to Commercial Space Leases	47
12.9	Delivery of Subcontracts	47
Article 13	3 FINAL DISBURSEMENT FOR CONSTRUCTION	48
13.1	Final Disbursement for Construction	48
Article 14	\$ SALE OF RESIDENTIAL UNITS OR OTHER PORTIONS OF THE PROJECT	49
14.1	Price List Schedule	49
14.2	Sales Agreements	49
14.3	Purchaser Deposits	50
14.4	Residential Unit Sales	51
14.5	Sales Operations and Seller's Obligations	53
14.6	Delivery of Sales Information and Documents	53
14.7	Borrower's Acknowledgment Regarding Buyer Financing	54
14.8	Condominium Regime	54
14.9	Release of Residential Units	55
14.10	Application of Sales Proceeds	57
	5 OTHER COVENANTS	58
15.1	Borrower further covenants and agrees as follows:	58
15.2	Single Purpose Entity Covenants	68
15.3	Authorized Representative	70
Article 16	5 CASUALTIES AND CONDEMNATION	70
16.1	Lender's Election to Apply Proceeds on Indebtedness	70
16.2	Borrower's Obligation to Rebuild and Use of Proceeds Therefor	71
Article 17	7 ASSIGNMENTS BY LENDER AND BORROWER	72
17.1	Assignments and Participations	72
17.2	Prohibition of Assignments and Transfers by Borrower	72
17.3	Prohibition of Transfers in Violation of ERISA	72
17.4	Successors and Assigns	73

ii

<u>Page</u>

Article 18	8 TIME OF THE ESSENCE	73
18.1	Time is of the Essence	73
Article 19	9 EVENTS OF DEFAULT	73
19.1	Events of Default	73
Article 20	0 LENDER'S REMEDIES IN EVENT OF DEFAULT	76
20.1	Remedies Conferred Upon Lender	76
Article 2	1 GENERAL PROVISIONS	77
21.1	Captions	77
21.2	Modification; Waiver	77
21.3	Governing Law	78
21.4	Acquiescence Not to Constitute Waiver of Lender's Requirements	78
21.5	Disclaimer by Lender	78
21.6	Partial Invalidity; Severability	79
21.7	Definitions Include Amendments	79
21.8	Execution in Counterparts	79
21.9	Entire Agreement	79
21.10	Waiver of Damages	79
21.11	Claims Against Lender	79
21.12	Jurisdiction	80
21.13	Set-Offs	80
21.14	Binding Effect	81
21.15	Waiver of Accord and Satisfaction	81
Article 22	2 NOTICES	81
Article 23	3 WAIVER OF JURY TRIAL	83
<u>Illinois C</u>	Collateral Protection Act Notice	viii

# EXHIBITS TO LOAN AGREEMENT

Exhibit A	Legal Description of Land
Exhibit B	Permitted Exceptions
Exhibit C	Title Requirements
Exhibit D	Form of Survey Certification
Exhibit E	Insurance Requirements
Exhibit F	Architect's Certificate
Exhibit G	Initial Budget
Exhibit H	Draw Request Forms
Exhibit I	Partial Plans and Specifications
Exhibit J	Proposed Finish Standards
Exhibit K	Bailment Letter (Warehousemen)
Exhibit L	Bailment Letter (Other Than Warehousemen)
Exhibit M	List of Sales Agreements and Price List Schedule
Exhibit N	Subcontract Delivery Deadline Dates
Exhibit O	Approved Form of Sales Agreement

# Table of Contents

Exhibit P	Materials Purchases Not Subject to Retainage
Exhibit Q	Allowable Tenant Improvements
Exhibit R	Delivery Schedule for Plans and Specifications
Exhibit S	Form of Second Estoppel and Agreement from City of Austin

### CONSTRUCTION LOAN AGREEMENT

Project Commonly Known as

"W Hotel and Residences"

Block 21, Austin, Texas

THIS CONSTRUCTION LOAN AGREEMENT ("Agreement") is made as of May 2, 2008, by and between CJUF II STRATUS BLOCK 21 LLC, a Delaware limited liability company ("Borrower"), and CORUS BANK, N.A., a national banking association, its successors and assigns ("Lender").

# <u>WITNESSETH</u>:

# RECITALS

Borrower is the owner in fee simple of an approximately 76,176 square foot parcel of land commonly known as "Block 21," bounded by A. Second, Third, Guadalupe and Lavaca Streets, City of Austin, County of Travis, State of Texas, and legally described in Exhibit A attached hereto (the "Land"). Borrower proposes to construct on the Land a mixed use project to be known as the "W Hotel and Residences," consisting of a building of thirty-six (36) stories (the "Building") and other facilities containing: (i) one hundred ninety-eight (198) residential condominium units (each, a "Residential Unit") on twenty (20) floors, from floor 18 through floor 37 of the Building, containing at least 272,272 Saleable Square Feet (with each capitalized term used and not defined in these Recitals being defined hereinbelow) of interior space and with interior finished ceiling heights of at least ten (10) feet (outside of areas containing mechanical runs), (ii) a "W" flagged hotel with two hundred fifty-two (252) guest rooms, situated on ten (10) floors, from floor 6 through floor 16 of the Building, to be furnished and managed pursuant to the Hotel Operating Agreement (as hereinafter defined), and containing at least 100,408 square feet of interior room space and 88,212 square feet of hotel operating space, collectively with, on floor 2 through floor 4 of the Building, 9,583 square feet of meeting space. a 8,060 square foot fitness facility, a 9,935 square foot pool and pool deck, and a business center (collectively, the "Hotel"), (iii) on floor 1 through floor 3 of the Building, 18,341 net rentable square feet of retail space (the "Retail Space") and 37,382 net Rentable Square Feet of office space (the "Office Space"), (iv) a live performance venue, on the top three (3) floors of an attached 4-story structure, containing at least 50,336 square feet and a minimum capacity of 2,480 people, with seating for approximately 2,160 people (the "Venue"), (v) 10,995 square feet of storage space, and (vi) a three (3)level subterranean parking garage (the "Parking Garage"), with a direct connection to elevators servicing the Residential Units, and containing parking spaces for at least 480 automobiles (each, a "Parking Space"), of which, 306 Parking Spaces shall be allocated for the Residential Units, and 116 Parking Spaces shall be allocated, collectively, for the Hotel, the Commercial Space, and the Venue. The Residential Units shall have a la carte access to the Hotel amenities on a pay-by-use basis, as provided in the Hotel Operating Agreement.

B. Borrower has applied to Lender for a loan in the aggregate amount of up to One Hundred Sixty-Five Million Dollars (\$165,000,000) to fund construction, development and marketing costs of the Project, and Lender is willing to make the Loan on the terms and conditions hereinafter set forth.

NOW, THEREFORE, in consideration of the mutual covenants and agreements herein contained, the parties hereto agree as follows:

# Article 1

# INCORPORATION OF RECITALS AND EXHIBITS

#### 1.1 Incorporation of Recitals.

The foregoing preambles and all other recitals set forth herein are made a part hereof by this reference.

#### 1.2 Incorporation of Exhibits.

Exhibits A through S to this Agreement, attached hereto, are incorporated in this Agreement and expressly made a part hereof by this reference.

# Article 2

# DEFINITIONS

## 2.1 Defined Terms.

The following terms as used herein shall have the following meanings:

Additional Equity Investment: As such term is defined in Section 11.2.

Affiliate: With respect to a specified person or entity, any individual, partnership, corporation, limited liability company, trust, unincorporated organization, association or other entity that, directly or indirectly, through one or more intermediaries, controls or is controlled by or is under common control with such person or entity, including, without limitation, any general or limited partnership in which such person or entity is a partner.

Agreement: This Construction Loan Agreement.

Allowable Tenant Improvements: As such term defined in Section 12.8.

Appraisal: An MAI certified appraisal of the Project performed in accordance with FIRREA and Lender's appraisal requirements by an appraiser selected and retained by Lender.

Approved Finish Standards: As such term is defined in Section 9.2(f).

Approved Lease: As such term is defined in Section 15.1(m).

Approved Plans and Specifications: As such term is defined in Section 9.2(e).

Architect: BOKA Powell, L.L.C.

Architect's Certificate: A certificate in the form of Exhibit F attached hereto executed by the Architect in favor of Lender.

Authorized Representative: William H. Armstrong, an individual, or such other individual that is designated in accordance with Section 15.3 of this Agreement.

# Available Sources of Funds: As such term is defined in Section 11.1(c).

Bankruptcy Code: Title 11 of the United States Code entitled "Bankruptcy" as now or hereafter in effect, or any successor thereto or any other present or future bankruptcy or insolvency statute.

Bonds: As such term is defined in Section 9.1(d).

Borrower: As such term is defined in the opening paragraph of this Agreement.

Budget: The budget for the Project specifying all costs and expenses of every kind and nature whatsoever to be incurred by Borrower in connection with the Project prior to the Maturity Date, as approved by Lender as set forth in Section 10.1.

Budget Line Item: As such term is defined in Section 10.2.

Building: As such term is defined in Recital A.

Business Day: Any Monday through Friday, excluding days on which Lender is closed for business.

<u>Change Order</u>: Shall mean any of the following: (i) a request for changes in the Approved Plans and Specifications (other than minor field changes involving no extra cost) or for a change to the General Contract Price, (ii) an amendment to the General Contract, (iii) a construction change directive or (iv) a written order for a minor change in the work issued by the architect.

CJUF: Canyon-Johnson Urban Fund II L.P., a Delaware limited partnership.

Closing: The date of the Closing Funding.

Closing Funding: The first disbursement of Loan proceeds in an amount of \$2,000,000, which shall be advanced on or about the date hereof.

<u>Collateral Assignment of Hotel Documents</u>: That certain Assignment of Hotel Documents collaterally assigning Borrower's interests in the Hotel Documents (and related documents) to Lender.

Commercial Space: Together, the Office Space and the Retail Space.

<u>Commitment</u>: Lender's maximum aggregate funding obligation hereunder of up to One Hundred Sixty-Five Million Dollars (\$165,000,000), less any reduction thereof in accordance with the terms of this Agreement.

Completion Date: With respect to the Residential Units, July 25, 2011; with respect to the Hotel, January 7, 2011; with respect to the Venue May 25, 2011; with respect to the

Commercial Space, January 7, 2011; and with respect to the entire Project, the Initial Maturity Date.

<u>Completion and Carveout Guaranty</u>: A guaranty of performance and completion, executed by Guarantor and pursuant to which Guarantor guarantees the lien-free and timely completion of the Project in accordance with all provisions of this Agreement and Borrower's obligation to keep the Loan In Balance and to pay for all cost overruns, subject to the limits stated therein, and guarantees specified non-recourse carve-out obligations.

Condominium Documents: As such term is defined in Section 8.1(s).

Condominium Marketing License Agreement: That certain Condominium Marketing License Agreement dated as of October 26, 2006 by and between Stratus Block 21 Investments, L.P. (predecessor in interest to Borrower), and Starwood Hotels & Resorts Worldwide, Inc.

<u>Construction or construction</u>: The construction and equipping of the Improvements in accordance with the Approved Plans and Specifications, and related improvements required to be performed by Borrower under Sales Agreements (including all off-site improvements reasonably required for use and operation of the Improvements) and the installation of all personal property, fixtures and equipment required for the operation of the Project or required under Sales Agreements.

Construction Disbursement: As such term is defined in Section 7.3.

<u>Construction Schedule</u>: A schedule reasonably satisfactory to Lender, establishing a timetable for completion of the Construction, showing, on a monthly basis, the anticipated progress of the Construction, and showing that the Improvements can be completed on or before the Completion Date and that the Residential Units will be delivered prior to any outside dates, if any, provided for in the Sales Agreements.

Contingency Fund: As such term is defined in Section 10.3.

Contractor's Contingency: As such term is defined in Section 10.3.

<u>Control</u>: As such term is used with respect to any person or entity, including the correlative meanings of the terms "controlled by" and "under common control with," shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management policies of such person or entity, whether through the ownership of voting securities, by contract or otherwise.

Declaration of Condominium: The Master Condominium Declaration and the Residential Condominium Declaration, individually or collectively, as the context shall infer.

Deed of Trust: A construction deed of trust, assignment of rents, security agreement and fixture filing executed by Borrower for the benefit of Lender securing this Agreement, the Note, and all obligations of Borrower in connection with the Loan, granting a first priority lien on Borrower's fee interest in the Project, subject only to the Permitted Exceptions.

Default or default: Any event, circumstance or condition, which, if it were to continue uncured, would, with notice or lapse of time or both, constitute an Event of Default hereunder.

Default Rate: As such term is defined in the Note.

Deficiency Deposit: As such term is defined in Section 11.1(b).

Deposits: The Earnest Money Deposits and the Upgrade Deposits.

Design Professionals: As such term is defined in Section 9.1(a).

Earnest Money Deposits: As such term is defined in Section 14.3 (a).

Environmental Indemnity: An environmental indemnity from Borrower and Guarantor, jointly and severally, indemnifying Lender with regard to all matters related to Hazardous Material and other environmental matters.

Environmental Proceedings: Any environmental proceedings, whether civil (including actions by private parties), criminal, or administrative proceedings, relating to the Project.

Environmental Report: An environmental report prepared at Borrower's expense by a qualified environmental consultant approved by Lender in its sole discretion addressed to Lender (or subject to separate letter agreement permitting Lender to rely on such environmental report), which complies with the USEPA "all appropriate inquiry" rule contained in 40 CRF Part 312.

Equity Investment: As such term is defined in Section 11.2.

ERISA: The Employee Retirement Income Security Act of 1974, as amended, and the regulations promulgated thereunder from time to time.

Escrow Agent: As such term is defined in Section 14.3(a).

Escrow Agreement: As such term is defined in Section 8.1(v).

Event of Default: As such term is defined in Article 19.

Excess Parking Spaces: As such term is defined in Section 14.1.

Exit Fee: As such term is defined in Section 7.4.

Extended Maturity Date: The date that is forty-six (46) months after the date of this Agreement, as the Initial Maturity Date may be extended by Borrower subject to the conditions contained in <u>Section 4.3</u>.

Extension Fee: As such term is defined in Section 4.3(b)(iii).

Extension Option: As such term is defined in Section 4.3(a).

FIRREA: The Financial Institutions Reform, Recovery and Enforcement Act of 1989, as amended from time to time.

Full Loan Opening Date: The date of Full Loan Opening.

<u>Full Loan Opening of Full Opening of the Loan</u>: The second disbursement of Loan proceeds, being the first disbursement of Loan proceeds other than the Closing Funding.

General Contract: As such term is defined in Section 9.1(a).

General Contract Price: As such term is defined in Section 9.1(a).

General Contractor: Austin Building Company.

<u>Governmental Approvals</u>: Collectively, all consents, licenses, and permits and all other authorizations or approvals required from any Governmental Authority for the Construction in accordance with the Approved Plans and Specifications or the sale of the Residential Units.

<u>Governmental Authority</u>: Any federal, state, county or municipal government, or political subdivision thereof, any governmental or quasigovernmental agency, authority, board, bureau, commission, department, instrumentality, or public body, or any court, administrative tribunal, or public utility.

Guarantor: Stratus Properties, Inc., a Delaware corporation.

Guarantor Financial Covenants: The covenants of Guarantor set forth in Section 16 of the Limited Payment Guaranty.

<u>Hard Costs</u>: Any and all costs related to or incurred in connection with the construction of the Project, including, without limitation, the cost of all labor, materials and equipment, but excluding any fees for architectural and engineering services, marketing fees, financing costs, developers' fees and other similar soft fees and costs. The Hard Costs include the items delineated as such on the Budget.

<u>Hazardous Material</u>: Means and includes gasoline, petroleum, asbestos containing materials, explosives, radioactive materials or any hazardous or toxic material, substance or waste which is defined by those or similar terms or is regulated as such under any Law of any Governmental Authority having jurisdiction over the Project or any portion thereof or its use, including: (i) any "hazardous substance" defined as such in (or for purposes of) the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C.A. § 9601(14) as may be amended from time to time, or any so-called "superfund" or "superlien" Law, including the judicial interpretation thereof; (ii) any "pollutant or contaminant" as defined in 42 U.S.C.A. § 9601(33); (iii) any material now defined as "hazardous waste" pursuant to 40 C.F.R. Part 260; (iv) any petroleum, including crude oil or any fraction thereof; (v) natural gas, natural gas liquids, liquefied natural gas, or synthetic gas usable for fuel; (vi) any "hazardous chemical" as defined pursuant to 29 C.F.R. Part 1910; (vii) any mold or fungus that may cause an allergic, toxic or inflammatory response in humans arising from exposure to such mold or fungus in indoor air; and (viii) any other toxic substance or contaminant that is subject to any other Law or

other past or present requirement of any Governmental Authority. Any reference above to a Law, includes the same as it may be amended from time to time, including the judicial interpretation thereof.

Hotel: As such term is defined in Recital A.

Hotel Documents: The Condominium Marketing License Agreement, the Hotel Operating Agreement, and/or the Technical Services Agreement, individually or collectively, as the context may infer.

Hotel Operator: W Hotel Management, Inc., a Delaware corporation, an Affiliate of Starwood Hotel & Resorts Worldwide, Inc.,

Hotel Operating Agreement: That certain W Austin Hotel Operating Agreement by and between Stratus Block 21 Investments, L.P., and Starwood Hotel & Resorts Worldwide, Inc., dated as of October 26, 2006, as amended by First Amendment to Operating Agreement dated January 30, 2008, as such agreement was assigned by Stratus Block 21 Investments, L.P. to Borrower by virtue of that certain Assignment and Assumption Agreement dated as of July 30, 2007; and as such agreement dated as of July 30, 2007.

HUD: United States Department of Housing and Urban Development.

ILSA: The Interstate Land Sales Full Disclosure Act, 42 USC 1701 et. seq., as amended.

<u>Improvements</u>: All of the improvements referred to in <u>Recital A</u> hereto and more particularly described in the Approved Plans and Specifications and any offsite improvements reasonably required to be constructed by Borrower for the use or operation of the improvements described in <u>Recital A</u>.

In Balance or in balance: As such term is defined in Article 11.

Including or including: Means "including, but not limited to".

Indemnified Party: As such term is defined in Section 15.1(t).

Initial Equity Investment: As such term is defined in Section 8.1(a).

Initial Maturity Date: The date that is forty (40) months from the date of this Agreement.

Insurance Policy: As such term is defined in Section 8.1(e).

Interest Rate: As such term is defined in the Note.

Interest Reserve Budget Line Item: As such term is defined in Section 10.4.

Internal Revenue Code: The Internal Revenue Code of 1986, as amended from time to time.

Land: As such term is defined in Recital A.

Laws: Collectively, all federal, state and local laws, statutes, codes, ordinances, orders, rules and regulations, including judicial opinions or precedential authority in the applicable jurisdiction.

Leases: The collective reference to all leases, subleases and occupancy agreements affecting the Project or any part thereof now existing or hereafter executed and all amendments, modifications or supplements thereto approved in writing by Lender.

Lender: As such term is defined in the opening paragraph of this Agreement and including any successor holder of the Loan from time to time.

Lender's Consultant: An independent consulting architect, inspector, and/or engineer designated by Lender in Lender's sole discretion.

Lender's Estimate of Remaining Costs: As such term is defined in Section 11.1(d).

Lender's Remaining Exposure: The sum, at any date, of the outstanding principal balance of the Loan and the Unfunded Commitment.

Limited Payment Guaranty: A guaranty of payment, executed by Guarantor and pursuant to which Guarantor guarantees the repayment of the Loan, in an amount up to \$20,000,000 of the principal amount of the Loan, plus accrued interest thereon (including default interest, if any), and the cost of the enforcement of such guaranty, all in accordance with the terms and provisions more particularly described therein.

List Price: As such term is defined in Section 14.1.

Loan: As such term is defined in Recital B.

Loan Documents: The collective reference to this Agreement, the documents and instruments listed in <u>Section 4.2</u>, and all the other documents and instruments entered into from time to time, evidencing or securing the Loan or any obligation of payment thereof or performance of Borrower's or Guarantor's obligations in connection with the transaction contemplated hereunder, each as amended.

Loan Term: The period of time commencing on the date of this Agreement through and including the date the Loan is repaid in full.

Master Condominium Declaration: That certain Declaration of Condominium Regime for Block 21 Master Condominiums to be recorded against the Project upon completion thereof, which will subdivide the Project into thirteen (13) "Master Units."

<u>Material Adverse Change or material adverse change</u>: If, in Lender's sole and reasonable discretion, the operations or financial condition of a person, entity, or property has changed in a manner likely to impair materially the value of Lender's security for the Loan, prevent timely

repayment of the Loan, or otherwise prevent the applicable person or entity from timely performing any of its material obligations under the Loan Documents.

Maturity Date: As such term is defined in Section 4.3.

Mezzanine Borrower: One or more constituent entities of Borrower, as agreed to between Borrower and Mezzanine Lender.

Mezzanine Lender: A institutional lender experienced in condominium development selected by Borrower and approved in writing by Lender in its reasonable discretion.

Mezzanine Loan: A loan from a Mezzanine Lender to Mezzanine Borrower in an aggregate principal amount such that the Qualifying Portion thereof is equal to or greater than Twenty Million Dollars (\$20,000,000), but no greater than Fifty Million Dollars (\$50,000,000), maturing on a date no earlier than the Maturity Date, and secured by pledges of the membership interests in Borrower.

<u>Mezzanine Loan Documents</u>: The collective reference to the documents and instruments entered into from time to time, evidencing or securing the Mezzanine Loan or any obligation of payment thereof or performance of Mezzanine Borrower's or Guarantor's obligations in connection with the transaction contemplated thereunder, each as amended.

Mezzanine Loan Intercreditor Agreement. A subordination and intercreditor agreement by and between Lender and Mezzanine Lender, prepared by counsel to Lender, in form and substance acceptable to Lender in its reasonable discretion, which provides, without limitation, that (i) the Mezzanine Loan is fully subordinate in right of payment and priority of lien to the Loan; (ii) Mezzanine Lender has no interest in any collateral which is unrelated to the Project and shall not accept any payments of the Mezzanine Loan until Lender is indefeasibly paid in full and Lender has released its Deed of Trust, except interest payments to be funded solely from the Mezzanine Loan interest reserve, if any; (iii) Mezzanine Lender shall not interfere in any manner with the enforcement of Lender's remedies; (iv) Mezzanine Lender shall not enforce any remedies with respect to its collateral or any other collateral (or against Guarantor), except that Mezzanine Lender may foreclose on the pledges of ownership interest in Borrower, provided that (A) any monetary default under the Loan (including any payment due to make the Loan In Balance) is cured by Mezzanine Lender, and (B) Mezzanine Lender delivers to Lender a replacement guaranty substantially similar to the Completion and Carveout Guaranty, and (v) provided that Lender consents to such price reduction in its sole discretion, Mezzanine Lender shall permit Borrower to enter into contracts and thereafter close on any Residential Unit provided that the gross purchase price (exclusive of Upgrades) is greater than or equal to 80% of the List Price for such Residential Unit as specified on the approved Price List Schedule attached hereto as Exhibit M; (vi) Mezzanine Lender shall be authorized (but not required) by the terms of the Mezzanine Loan to make protective advances with respect to the Project and the Loan; (vii) a default under the Loan shall be a default under the Mezzanine Loan; and (viii) Mezzanine Lender shall not be permitted to (A) sell, transfer, or otherwise assign its interest in the Mezzanine Loan, (B) sell or otherwise offer any participations in the Mezzanine Loan, except at par to a Qualified Transferee, or (C) cause or permit the sale, assignment or other transfer of any of the ownership interests in a Mezzanine Lender which is a single purpose entity; in each case,

without the express prior written approval of Lender. Notwithstanding the foregoing, Lender shall permit the transfer of an economic interest of less than fifty percent (50%) in the Mezzanine Loan to a Qualified Transferee, provided (x) the identity, ownership, and financial profile of such Qualified Transferee is disclosed to Lender prior to such transfer, (y) the Loan is In Balance, and (z) there is no Default or Event of Default.

<u>Net Operating Income</u>: For the applicable month, the gross income from the Project less a prorated management fee (not to exceed 4% of annual gross revenues), customary monthly operating expenses, and reasonable prorated tax and insurance reserves; provided, however, for the Hotel, Net Operating Income shall be deemed the amount distributable by Hotel Operator to Borrower pursuant to the Hotel Operating Agreement.

<u>Net Sales Proceeds</u>: The gross sales price paid by any Residential Unit Purchaser for its respective Residential Unit (exclusive of Residential Unit customization items paid for from Upgrade Deposits, but inclusive of all Upgrade Profits and inclusive of all fees and other amounts paid by Residential Unit Purchasers in excess of the purchase price) minus brokerage commissions, title costs, legal fees and other customary closing costs associated with the sale of such Residential Unit that are paid or incurred by Borrower, provided that in calculating Net Sales Proceeds closing costs shall be excluded from this clause to the extent funded from the Loan (rather than being paid from gross sales proceeds).

Non-Disturbance Agreement: That certain subordination, non-disturbance and attornment agreement by and among Hotel Operator, Lender and Borrower.

Note: A promissory note in the amount of One Hundred Sixty-Five Million Dollars (\$165,000,000), executed by Borrower and payable to the order of Lender, evidencing the Loan.

<u>OFAC</u>: As such term is defined in <u>Section 3.1(x)</u>.

Office Space: As such term is defined in Recital A.

Owner's Hard Cost Contingency: As such term is defined in Section 10.3.

Parking Garage: As such term is defined in Recital A.

Parking Space: As such term is defined in Recital A.

Partial Plans and Specifications: Those certain partially completed Plans and Specifications set forth on Exhibit I hereto.

Partial Proposed Finish Standards: Those certain partially completed Proposed Finish Standards set forth on Exhibit J hereto.

Permitted Affiliate Expenses: As such term is defined in Section 12.7.

<u>Permitted Exceptions</u>: Those matters listed on <u>Exhibit B</u> attached hereto, to which title to the Project may be subject at the Closing, and thereafter such other title exceptions as are acceptable to Lender in its sole discretion and approved by Lender in writing.

Person: Any natural person, partnership, limited liability company, corporation, trust, Governmental Authority or other entity.

Plans and Specifications: As such term is defined in Section 9.2(e).

Pre-sale Requirement: As such term is defined in Section 8.1(u).

Price List Schedule: As such term is defined in Section 14.1.

Proceeding: As such term is defined in Section 21.12.

Proceeds: As such term is defined in Section 16.1(a).

<u>Project</u>: The collective reference to (i) the Land, together with all buildings, structures and improvements located or to be located thereon, including the Improvements, (ii) all rights, privileges, easements and hereditaments relating or appertaining thereto, and (iii) all personal property, fixtures and equipment required or beneficial for the operation thereof.

Proposed Finish Standards: As such term is defined in Section 9.2(f).

Qualified Transferee: As such term shall be defined in the Mezzanine Loan Intercreditor Agreement.

Qualifying Portion: As such term is defined in Section 8.1(a).

Qualifying Sales Agreement: As such term is defined in Section 14.4.

Reinvested Proceeds: As such term is defined in Section 14.10(a).

Release Price: As such term is defined in Section 14.9 or Section 14.9A, as applicable for the component of the Project being released.

<u>Remaining Units</u>: Residential Units which have not been conveyed to Residential Unit Purchasers as of the time of determination of the Remaining Units (and, therefore, remain as collateral for the Loan).

<u>Rentable Square Feet</u>: The number of indoor net rentable square feet in any particular portion of the Commercial Space or other space, as measured from the interior of the glass in the exterior walls, the middle of demising walls between rentable spaces and to the public side of any common area walls, but excluding balconies, terraces, hallways, common areas, lobbies, loft space or "loft walls" and the structural walls and areas of exit stairs, elevator shafts, and common mechanical shafts. (One such rentable square foot is referred to in the singular as a "<u>Rentable Square Foot</u>".)

Replacement Lender: As such term is defined in Section 4.8(b).

Required Construction Commencement Date: The date that is thirty (30) days after the date hereof.

<u>Required Permits</u>: Each building permit, environmental permit, utility permit, land use permit and any other permits, approvals or licenses issued by any Governmental Authority that are required in connection the Construction, marketing, sale or operation of the Project.

<u>Residential Condominium Declaration</u>: That certain sub-condominium declaration, entitled the "Subordinate Declaration of Condominium Regime for Block 21 Hotel Residential Condominiums," to be recorded against the Residential Units and Hotel.

Residential Unit: As such term is defined in Recital A. For the sake of clarity, the Hotel is not included in the defined term "Residential Units."

Residential Unit Purchaser: The contract purchaser(s) under each Sales Agreement.

Retail Space: As such term is defined in Recital A.

Retainage: As such term is defined in Section 12.4.

Revised GMP: As such term is defined in Section 9.2(a).

<u>Saleable Square Feet</u>: The number of indoor net saleable square feet in a Residential Unit or other space, as measured from the exterior of the exterior walls, the middle of demising walls between Residential Units and to the public side of any common area walls, but excluding balconies, terraces, common hallways, common mechanical shafts, lobbies, loft space or "loft walls" and the structural walls and areas of exit stairs, elevator shafts, and other common areas. (One such saleable square foot is referred to in the singular as a "<u>Saleable Square Foot</u>".)

Sales Agreement: As such term is defined in Section 14.2.

Sales Report: As such term is defined in Section 14.6.

Soft Cost Contingency: As such term is defined in Section 10.3.

Soft Costs: All costs incurred or to be incurred in connection with the Project, other than the Hard Costs, including, without limitation, interest on the Loan, fees incurred in connection with the Loan, commissions, appraisal fees, architectural and engineering fees, title and recording charges, legal fees, real estate taxes and other impositions and sales and marketing costs. Soft Costs shall include the items delineated as such on the Budget.

Soil Report: A soil test report prepared by a licensed engineer satisfactory to Lender indicating to the satisfaction of Lender that the soil and subsurface conditions underlying the Project will support the Improvements.

State: The state in which the Land is located.

Subcontracts: Subcontracts for labor or materials to be furnished to the Project.

Substantial Completion: The satisfaction of all of the following conditions: (a) the date when the Construction shall have been completed (except for Punch List Items and minor items

which can be fully completed without material interference with the use and operation of the Project) in accordance with the Approved Plans and Specifications; and (b) all material permits and approvals required for the normal use and occupancy of the Project (including a temporary certificate of occupancy) shall have been issued by the appropriate Governmental Authority and shall be in full force and effect to such extent under items (a) and (b) so that Borrower has the absolute right and ability under applicable Laws to convey and deliver Residential Units to the respective Residential Unit Purchasers and open and operate the Hotel.

<u>Technical Services Agreement</u>: That certain Technical Services Agreement dated as of October 26, 2006 by and between Stratus Block 21 Investments, L.P. (predecessor in interest to Borrower), and Starwood Hotels & Resorts Worldwide, Inc.

Tenant: The tenant under a Lease.

<u>Title Insurer</u>: Commonwealth Land Title Insurance Company, or such other title insurance company licensed in the State as may be approved in writing by Lender.

<u>Title Policy</u>: An ALTA 2006 Lender's Title Insurance Policy, or an equivalent Texas Lender's Title Insurance Policy, with extended coverage issued by the Title Insurer insuring the lien of the Deed of Trust as a valid first, prior and paramount lien upon the Project and all appurtenant easements, and subject to no other exceptions other than the Permitted Exceptions and otherwise satisfying the requirements of <u>Exhibit C</u> attached hereto and made a part hereof, all to the extent permitted by the Laws of the State.

<u>Transfer</u>: Any sale, transfer, lease (other than a Lease approved by Lender), conveyance, alienation, pledge, assignment, Deed of Trust, encumbrance, hypothecation or other disposition of (a) all or any portion of the Project or any portion of any other security for the Loan, (b) all or any portion of Borrower's right, title and interest (legal or equitable) in and to the Project or any portion of any other security for the Loan, or (c) any interest in Borrower or any interest in any entity, including, without limitation, Guarantor, which directly or indirectly holds an interest in, or directly or indirectly controls, Borrower.

<u>Unavoidable Delay</u>: Any delay in the construction of the Project, caused by natural disaster, fire, earthquake, hurricanes, tropical storms, floods, war, acts of terrorism, explosion, extraordinary adverse weather conditions, inability to procure or a general shortage of labor, equipment, facilities, energy, materials or supplies in the open market, failure of transportation, strikes or lockouts, or like causes, so long as such cause is not within the reasonable control of Borrower, but in no event to exceed ninety (90) days in the aggregate. In no event shall lack of funds be deemed an Unavoidable Delay.

<u>Unfunded Commitment</u>: The Commitment less all disbursements of the Loan made prior to the date on which the amount of the Unfunded Commitment is being calculated.

Upgrade Deposits: As such term is defined in Section 14.3(c).

Upgrade: As such term is defined in Section 14.3(c).

Upgrade Profit: The amount by which Borrower's costs of providing any upgrades is less than the cost charged to the Residential Unit Purchaser for any upgrades.

Venue: As such term is defined in Recital A.

Waste Management Plan: As such term is defined in Section 9.2(e).

# 2.2 Other Definitional Provisions.

All terms defined in this Agreement shall have the same meanings when used in the Note, Deed of Trust, any other Loan Documents, or any certificate or other document made or delivered pursuant hereto. The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement.

# Article 3

# BORROWER'S REPRESENTATIONS AND WARRANTIES

#### 3.1 Representations and Warranties.

To induce Lender to execute this Agreement and perform its obligations hereunder, Borrower hereby represents and warrants to Lender as follows:

(a) Borrower has good and indefeasible fee simple title to the Project, subject only to the Permitted Exceptions.

(b) Except as previously disclosed to Lender in writing, no litigation or proceedings are pending, or to the best of Borrower's actual knowledge threatened in writing, against Borrower or Guarantor, that could, if adversely determined, be reasonably expected to cause a Material Adverse Change with respect to Borrower, Guarantor or the Project. There are no pending Environmental Proceedings and Borrower has no actual knowledge of any threatened Environmental Proceedings or any facts or circumstances that are reasonably likely give rise to any future Environmental Proceedings.

(c) Borrower is a duly organized and validly existing limited liability company and has full power and authority to execute, deliver and perform all Loan Documents to which Borrower is a party, and such execution, delivery and performance have been duly authorized by all requisite action on the part of Borrower; Borrower has been a single purpose entity in compliance with <u>Section 15.2</u> hereof since its formation.

(d) No consent, approval or authorization of or declaration, registration or filing with any Governmental Authority or nongovernmental person or entity, including any creditor, partner, member or shareholder of Borrower or Guarantor, is required in connection with the execution, delivery and performance of this Agreement or any of the Loan Documents other than the recordation of the Deed of Trust and Declaration of Condominium for the Project and the filing of UCC-1 Financing Statements, except for such consents, approvals or authorizations of or declarations or filings with any Governmental Authority or non-governmental person or entity



where the failure to so obtain would not have a material adverse effect on Borrower or Guarantor or which have been obtained as of any date on which this representation is made or remade.

(e) The execution, delivery and performance of this Agreement, the execution and payment of the Note and the granting of the Deed of Trust and other security interests under the other Loan Documents have not constituted and will not constitute, upon the giving of notice or lapse of time or both, a breach or default under any other agreement to which Borrower or Guarantor is a party or may be bound or affected, or a violation of any law or court order that may affect the Project, any part thereof, any interest therein, or the use thereof.

(f) There is no default under this Agreement or the other Loan Documents, nor any condition that, after notice or the passage of time or both, would constitute a default or an Event of Default under said documents.

(g) (i) No condemnation of any portion of the Project, (ii) no condemnation or relocation of any roadways abutting the Project, and (iii) no proceeding to deny access to the Project from any point or planned point of access to the Project, has commenced or, to Borrower's actual knowledge, is contemplated by any Governmental Authority.

(h) The amounts set forth in the Budget present a full and complete itemization by category of all costs, expenses and fees that Borrower reasonably expects to pay or reasonably anticipates becoming obligated to pay to complete the Construction (including all off-site improvements to be paid for by Borrower), operate the Project and market and sell the Residential Units. Borrower is unaware of any other such costs, expenses or fees that are material and are not covered by the Budget. Borrower further warrants that neither Borrower, Guarantor, nor any of their respective Affiliates are receiving any other payments, distributions, or other consideration directly or indirectly from Borrower, the Project, its seller, contractors or any other party associated with the Project other than the Permitted Affiliate Expenses.

(i) Neither the construction of the Improvements nor the use of the Project when completed in accordance with the Approved Plans and Specifications and the contemplated accessory uses will violate (i) any Laws (including subdivision, zoning, building, environmental protection and wetland protection Laws), or (ii) any building permits, restrictions of record, or agreements affecting the Project or any part thereof. Neither the zoning authorizations, approvals or variances nor any other right to construct or to use the Project is to any extent dependent upon or related to any real estate other than the Land. All Governmental Approvals required for the Construction in accordance with the Approved Plans and Specifications have been obtained or will be obtained prior to Closing (except for those Governmental Approvals that cannot or need not be obtained until a later stage of the Construction or completion of Construction, in which case such Governmental Approvals will be obtained by Borrower on a timely basis and copies will be delivered to Lender on the earliest possible date) and all Laws relating to the Construction and operation of the Improvements have been complied with in all material respects and to Borrower's knowledge, after due inquiry, all permits and licenses, required for the operation of the Project that cannot be obtained until the Construction is completed can be obtained if the Improvements are completed in accordance with the Approved Plans and Specifications.

(j) The Project will have adequate water, gas, if applicable, and electrical supply, storm and sanitary sewerage facilities, other required public utilities, fire and police protection, and means of access between the Project and public highways, and none of the foregoing will be foreseeably delayed or impeded by virtue of any requirements under any applicable Laws.

(k) No brokerage fees or commissions are payable by or to any person in connection with this Agreement or the Loan to be disbursed hereunder, other than to Holliday Fenoglio Fowler, who will be paid by Borrower.

(I) All financial statements and other information previously furnished by Borrower or Guarantor to Lender in connection with the Loan are true, complete and correct in all material respects and fairly present the financial conditions of the subjects thereof as of the respective dates thereof and do not fail to state any material fact necessary to make such statements or information not misleading, and no Material Adverse Change with respect to Borrower or Guarantor has occurred since the respective dates of such statements and information. Neither Borrower nor Guarantor has any material liability, contingent or otherwise, not disclosed in such financial statements and that all charges payable with respect to the Project are current and not in default. Except as previously disclosed in writing to Lender, neither Borrower, nor Guarantor, nor any officer or director of Borrower, nor any equity owner of Borrower or Guarantor, or any of Borrower's or Guarantor's respective Affiliates (excluding investors in CJUF and any shareholders of Guarantor other than those that are considered "insiders" under SEC regulations): (i) has ever been the subject of any criminal proceedings (other than minor traffic violations); (ii) has ever been the owner, whether directly or indirectly, to a deed in lieu of foreclosure; or (iv) is currently a party to any material pending litigation or administrative proceedings, or subject to any judicial or non-judicial orders or consent agreements.

(m) Except as disclosed in any Environmental Report delivered by Borrower to Lender prior to the date hereof, (i) to Borrower's actual knowledge, the Project is in a safe condition, and, except for small quantities of Hazardous Materials lawfully used in the ordinary course of construction, maintenance and operation of the Project, is free of all Hazardous Material and is in compliance with all applicable Laws; (ii) except for small quantities of Hazardous Materials lawfully used in the ordinary course of construction, maintenance and operation of the Project, neither Borrower nor, to the actual knowledge of Borrower, any other person or entity, has ever caused or permitted any Hazardous Material to be placed, held, located or disposed of on, under, at or in a manner to affect the Project, or any part thereof, and the Project has never been used (whether by Borrower or, to the actual knowledge of Borrower, by any other person or entity) for any activities involving, directly or indirectly, the use, generation, treatment, storage, transportation, or disposal of any Hazardous Material; (iii) neither the Project nor Borrower is subject to any remedial obligations under any applicable Laws pertaining to health or the environment; and (iv) to the actual knowledge of Borrower, there are no underground tanks, vessels, or similar facilities for the storage, containment or accumulation of Hazardous Materials of any sort on, under or affecting the Project.

(n) For all purposes the Project may be mortgaged, conveyed and otherwise dealt with as an independent parcel and is a separate real estate tax parcel.

(o) Borrower and its agents have not entered into any Leases, subleases or other arrangements for occupancy of space within the Project (other than Sales Agreements that permit occupancy by the Residential Unit Purchasers following closing thereunder and the Hotel Operating Agreement).

(p) When the Construction is completed substantially in accordance with the Approved Plans and Specifications, no building or other improvement will encroach upon any property line, building line, setback line, side yard line or any recorded or visible easement (or other easement of which Borrower is aware or has reason to believe may exist) in violation thereof.

(q) The Loan is not being made for the purpose of purchasing or carrying "margin stock" within the meaning of Regulation T, U or X issued by the Board of Governors of the Federal Reserve System, and Borrower agrees to execute all instruments necessary to comply with all the requirements of Regulation U of the Federal Reserve System.

(r) Borrower is not a party in interest to any plan defined or regulated under ERISA, and the assets of Borrower are not "plan assets" of any employee benefit plan covered by ERISA or Section 4975 of the Internal Revenue Code.

(s) Borrower is not a "foreign person" within the meaning of Section 1445 or 7701 of the Internal Revenue Code.

(t) Borrower uses no trade name other than (i) its actual name set forth herein and (ii) "W Hotel and Residences." The principal place of business of Borrower is as stated in <u>Article 22</u>.

(u) Borrower's place of organization is Delaware.

(v) Except as set forth in Exhibit M, there are no Sales Agreements to purchase Residential Units. The Sales Agreements listed on Exhibit M are in full force and effect and such Sales Agreements are not subject to any rights of rescission. Borrower hereby represents that Exhibit M is a true, accurate and complete schedule of all Sales Agreements and sets forth: (i) the name of Residential Unit Purchaser, (ii) the Residential Unit being purchased, (iii) any upgrades, (iv) any Upgrade Deposit, (v) the purchase price, and (vi) the Earnest Money Deposit. All Sales Agreements are (or when entered into, and after expiration of statutory rescission periods, will be) Qualifying Sales Agreements. No event of default, or any event that, with the passage of time or the giving of notice, or both, would constitute an event of default, has occurred pursuant to the terms of any of the Sales Agreements on the part of Borrower or, to Borrower's actual knowledge, the other parties thereto. No Residential Unit Purchaser under the Sales Agreements listed on Exhibit M has terminated its respective Sales Agreement and there are no side agreements with any Residential Unit Purchasers modifying any of the terms of the Sales Agreements or otherwise.

(w) All Sales Agreements are exempt from or will comply with the requirements of ILSA, and Laws of the State (and any applicable local Laws), so that (i) the sale of the Residential Units is lawful and will not be subject to interruption due to a violation of Laws, (ii) no Sales Agreement is terminable under any of such Laws (other than the termination rights contained in such Sales Agreement), and (iii) neither Borrower nor the Project will be subject to any civil or criminal penalties by reason of failure to comply with such Laws. The marketing and sale of Residential Units by Borrower (and any marketing or sales of Residential Units) is, and at all times has been, in compliance with all Laws pertaining to the sale of condominiums (and/or residential real estate generally). All consents and approvals needed for the sale of Residential Units under applicable federal, state and local Laws have been received and remain in full force and effect.

(x) The Hotel Documents are in full force and effect. No event of default, or any event that, with the passage of time or the giving of notice, or both, would constitute an event of default, has occurred pursuant to the terms of the Hotel Documents, either on the part of Borrower or, to Borrower's actual knowledge, the other parties thereto. Hotel Operator, or any other party to any Hotel Document, has not sent to Borrower any notices of default under any Hotel Document, nor has Hotel Operator, or any other party to any Hotel Document, sent to Borrower any other written notices of a material nature. The Hotel Documents have not been amended (except as set forth in the definition of Hotel Documents). There are no other agreements, written or oral, with Hotel Operator, Starwood Hotels & Resorts Worldwide, Inc., or any Affiliates of the foregoing, that supplement or modify any of the terms of any of the Hotel Documents or otherwise.

(y) Neither Borrower, Guarantor or any other person owning an interest in Borrower is (or will be) a person with whom Lender is restricted from doing business with under regulations of the Office of Foreign Asset Control ("<u>OFAC</u>") of the Department of the Treasury of the United States of America (including, those persons named on OFAC's Specially Designated and Blocked Persons list) or under any statute, executive order (including the September 24, 2001 Executive Order Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism), or other governmental action and is not and shall not knowingly engage in any dealings or transactions or otherwise be associated with such persons. In addition, Borrower hereby agrees to provide Lender with any additional information that Lender deems reasonably necessary from time to time in order to ensure compliance with all applicable Laws concerning money laundering and similar activities.

(z) Borrower shall have complied, in all respects, with the provisions of the USA PATRIOT Act of 2001, as applicable to Borrower and the Project.

(aa) The Project complies and, when constructed, shall comply with all requirements of that certain Declaration of Restrictive Covenants by the City of Austin, dated December 15, 2006 recorded in the official public records of Travis County, Texas as Instrument No. 2006240877.

(bb) All statements set forth in the Recitals are true and correct in all material respects.

#### 3.2 Survival of Representations and Warranties.

Borrower agrees that all of the representations and warranties set forth in <u>Section 3.1</u> and elsewhere in this Agreement are true in all material respects as of the date hereof, will be true in all material respects at Closing and, except for matters that have been disclosed by Borrower and approved by Lender in writing, will be true in all material respects at all times thereafter (including at Full Loan Opening) until the Loan has been repaid and Borrower's obligations hereunder have been satisfied in full. Each request for a disbursement under the Loan Documents shall constitute a reaffirmation of such representations and warranties, as deemed modified in accordance with the disclosures made and approved as aforesaid, as of the date of such request. It shall be a condition precedent to the Closing Funding and each subsequent disbursement that each of said representations and warranties is true and correct in all material respects as of the date of such requested disbursement. In addition, at Lender's request, Borrower shall reaffirm such representations and warranties in writing prior to each disbursement hereunder.

# Article 4

# LOAN AND LOAN DOCUMENTS

# 4.1 Agreement to Borrow and Lend; Lender's Obligation to Disburse; Excess Disbursements.

Subject to the terms, provisions and conditions of this Agreement and the other Loan Documents, Borrower agrees to borrow from Lender and Lender agrees to lend to Borrower the Loan, for the purposes and subject to all of the terms, provisions and conditions contained in this Agreement. If Lender consists of more than one party, the obligations of each such party with respect to the amount it has agreed to loan to Borrower shall be several (and not joint and several) and each lending party's obligations shall be limited to its proportionate share of the Loan and of each advance.

(a) The maximum aggregate principal amount of the Loan to be funded hereunder shall be the lesser of (i) the Commitment, or (ii) the total costs associated with the Project (as described in the Budget) less the required Equity Investment.

(b) Lender agrees, upon Borrower's compliance with and satisfaction of all conditions precedent to the Closing Funding set forth in this Agreement and provided (i) the Loan is In Balance, (ii) no Material Adverse Change has occurred and is continuing with respect to Borrower, Guarantor or the Project, (iii) no material casualty to the Project has occurred that has not been repaired and there is no existing or threatened condemnation or taking which could cause a Material Adverse Change with respect to the Project and (iv) no Default or Event of Default has occurred and is continuing hereunder, to make the Closing Funding.

(c) After the Closing Funding, Borrower shall be entitled to receive further successive disbursements of the proceeds of the Loan in accordance with <u>Articles 8, 9, 12</u> and <u>13</u> following compliance with all conditions precedent thereto set forth in this Agreement, provided that (i) the Loan remains In Balance, (ii) Borrower has complied with all conditions precedent to disbursement from time to time set forth in this Agreement including the requirements of <u>Section</u>

<u>3.2</u> and <u>Articles 8, 9, 12</u> and <u>13</u>, (iii) no Material Adverse Change has occurred and is continuing with respect to Borrower, Guarantor or the Project, (iv) no material casualty to the Project has occurred that has not been repaired or is not being repaired in accordance with Article 16 hereof, and there is no existing or threatened condemnation or taking which could cause a Material Adverse Change with respect to the Project and (v) no Event of Default and no material Default has occurred and is continuing hereunder or under any other Loan Document. Lender shall make commercially reasonable efforts to fund such subsequent disbursements within ten (10) Business Days after receipt of all of the documents required under this Agreement, including a draw request together with all items listed in <u>Section 12.3</u>.

(d) To the extent that Lender may acquiesce in noncompliance with any requirements set forth in this Agreement precedent to the Closing Funding, the Full Opening of the Loan, or any subsequent disbursement of Loan proceeds, such acquiescence shall not constitute a waiver by Lender, and Lender may at any time after such acquiescence require Borrower to comply with all such requirements.

(e) All payments by Borrower on account of the Loan shall be made as such amounts become due or are declared due pursuant to the terms of this Agreement and the other Loan Documents. All payments shall be made without deduction, defense, setoff or counterclaim as follows:

For payments made by Regular Mail: Corus Bank N.A. P.O. Box 102865 Atlanta, Georgia 30368-2865 For payments made by Federal Express: Corus Bank N.A. - #102865 Lockbox Mail Department Georgia Operations Center 100 South Crest Drive Stockbridge, Georgia 30281 For payments made by Wire Transfer and ACH: SUNTRUST BANK, ATLANTA ABA 061000104 TO CREDIT ACCOUNT 1000008140328 ACCOUNT NAME: CORUS BANK NA FOR FURTHER CREDIT TO: Corus Bank Loan No. 67920-11152

#### 4.2 Loan Documents.

Borrower agrees that it will, on or before the date hereof, execute and deliver or cause to be executed and delivered to Lender the following documents in form and substance acceptable to Lender:

(a) The Note.

- (b) The Deed of Trust.
- (c) The Completion and Carveout Guaranty.
- (d) The Limited Payment Guaranty.
- (e) The Environmental Indemnity.
- (f) The Collateral Assignment of Hotel Documents.

(g) A collateral assignment, to the extent assignable, of construction documents, including, without limitation, the General Contract, all architecture, Design Professional and engineering contracts, Plans and Specifications, permits, licenses, approvals and development rights, together with consents to the assignment and continuation agreements from the General Contractor, the architect, real estate broker and other parties reasonably specified by Lender.

(h) A collateral assignment, to the extent assignable, of all Sales Agreements, Earnest Money Deposits, Upgrade Deposits and all other documents relating to the establishment of a condominium regime at the Project.

(i) Such UCC financing statements as Lender determines are advisable or necessary to perfect or notify third parties of the security interests intended to be created by the Loan Documents.

(j) A collateral assignment, to the extent assignable, of any management contract entered into with respect to the Project.

(k) Such other documents, instruments or certificates as Lender and its counsel may reasonably require, including such documents as Lender in its reasonable discretion deems necessary or appropriate to effectuate the terms and conditions of this Agreement and the Loan Documents, and to comply with the laws of the State.

# 4.3 Term of the Loan.

(a) All principal, interest and other sums due under the Loan Documents shall be due and payable in full on the Initial Maturity Date, provided that Borrower shall have the option to extend the Maturity Date for one (1) additional six-month period (the "<u>Extension Option</u>"), provided Borrower has satisfied the conditions set forth in subparagraph (b) below. All references herein to the "<u>Maturity Date</u>" shall mean the Initial Maturity Date, or, in the event Borrower satisfies the conditions to the exercise of the Extension Option in accordance with this

Section 4.3, the "Maturity Date" shall mean the Extended Maturity Date. Interest only payments shall continue to be due and payable on the first of the month according to the Interest Rate then in effect on the outstanding principal balance of the Loan during the extension period.

(b) Borrower may only exercise the Extension Option upon satisfying the following conditions:

(i) The Project has been completed in accordance with the Approved Plans and Specifications in all material respects, the Project is lien free (other than liens for the benefit of Lender and Permitted Exceptions), and certificates of occupancy shall have been issued by the appropriate Governmental Authority for all of the Project sufficient, among other things, to allow operation of the Hotel and Venue, leasing and occupancy of the Commercial Space, the conveyance and occupancy of the Residential Units, and the use and occupancy of the Parking Garage and all Parking Spaces, and to allow satisfaction of item (vii) below (any certificate of occupancy for the Commercial Space may be conditioned upon the completion of tenant finish improvements);

(ii) The Hotel is open and operating in accordance with the Hotel Documents, and the Hotel Operator, on behalf of Starwood Hotel & Resorts Worldwide, Inc., has accepted and confirmed the completion of the Hotel in accordance with the Hotel Documents;

(iii) Borrower shall have delivered to Lender written notice of such election no earlier than seventy-five (75) days and no later than thirty (30) days prior to the Initial Maturity Date;

(iv) Borrower shall pay, together with its written notice of such election, an extension fee (the "Extension Fee") equal to 0.5% of Lender's Remaining Exposure at the time the Extension Option is requested;

(v) If there is a Mezzanine Loan, the Mezzanine Lender extends the maturity date of the Mezzanine Loan to a date no earlier than the Extended Maturity Date;

(vi) The Loan has not matured and no Event of Default or material Default has occurred and is continuing under the Loan Documents;

(vii) Following the full funding of the Reinvested Proceeds pursuant to <u>Section 14.10(b)</u> hereof, Borrower has closed and conveyed, pursuant to Qualifying Sales Agreements and each at no less than 90% of the applicable List Price, Residential Units in an amount sufficient to generate aggregate gross sales proceeds of at least \$55,000,000 (apart from and in addition to the gross sales proceeds which generated the Reinvested Proceeds), and Borrower shall have paid to Lender the Release Prices for each such Residential Unit sold; and

(viii) Borrower has funded the Additional Equity Investment, as and to the extent required hereunder.

(c) Notwithstanding the above provisions of this Section 4.3, in the event that (x) Borrower does not repay the Loan in full on or before the Initial Maturity Date and (y) the Maturity Date is not extended pursuant to the provisions of Section 4.3(b) (whether because

Borrower did not request the extension or Borrower requested the extension but did not qualify for such extension) then the Maturity Date shall nevertheless be extended from the Initial Maturity Date to a date that is thirty (30) days after the Initial Maturity Date and the Extension Fee shall be due and payable from Borrower to Lender on the Initial Maturity Date. The Deed of Trust will not be released until the Extension Fee and all other amounts due under the Loan have been paid in full. In such event, Borrower shall not be entitled to any further extensions of the Maturity Date. No such extension shall occur if the Loan has been accelerated prior to the Initial Maturity Date.

(d) If Borrower timely sends written notice of Borrower's election to exercise the Extension Option in accordance with <u>Section 4.3(b)</u>, Lender, no later than twenty (20) days prior to the Initial Maturity Date, shall notify Borrower ("<u>Lender's Response Notice</u>") whether, with respect to the Extension Option, the conditions set forth in Section 4.3(b) have been satisfied and accordingly the Maturity Date has been extended to the Extended Maturity Date.

(e) If Lender fails to timely send Lender's Response Notice, but Borrower has not qualified for such Extension Option, then Lender shall not thereby be deemed to have waived the Extension Fee payable under subparagraph 4.3(c); provided, however, that in such circumstance, if (x) Borrower repays the Loan in full prior to its receipt of Lender's Response Notice or (y) Borrower repays the Loan in full no later than the date twenty (20) days after its receipt of Lender's Response Notice, then no Extension Fee shall be owed under subparagraph 4.3(c).

#### 4.4 Prepayments.

Borrower shall have the right to make prepayments of the Loan in accordance with and subject to the terms of the Note.

#### 4.5 Required Principal Payments.

All principal shall be paid on or before the Maturity Date.

#### 4.6 Receipt of Payments.

All payments received by Lender prior to or at 3:00 p.m. (Chicago time) on a Business Day shall be credited to Borrower on the day of receipt; all payments received after 3:00 p.m. (Chicago time) on a Business Day shall be deemed received on the next succeeding Business Day.

# 4.7 Termination of Lender's Unfunded Commitment.

Upon the repayment in full of the outstanding principal balance of the Loan, Lender's obligation to fund the Unfunded Commitment shall thereupon terminate and Lender shall have no further obligation to fund Loan proceeds hereunder. If, in Lender's sole discretion, Lender agrees (in writing) that the Commitment shall not so terminate, all Release Prices for the sale of Residential Units or any other part of the Project shall be escrowed with Lender until such time as Lender's obligation to fund the Commitment expires or is terminated and the Loan has been repaid.

#### 4.8 Lender Default.

If Lender is prevented from funding the Unfunded Commitment by reason of bankruptcy or insolvency proceedings or an order from regulatory authorities in connection with Lender's insolvency or failure to meet regulatory requirements, then Lender shall notify Borrower and, so long as no Event of Default exists, Borrower shall have the right to do either of the following no later than ninety (90) days after receipt of such notice:

(a) Borrower may prepay the Loan in full without the payment of any Prepayment Charge (as such term is defined in the Note) or the remaining Exit Fee; or

(b) Borrower may cause another financial institution experienced at making similar construction loans and reasonably acceptable to Lender (the "<u>Replacement Lender</u>") to agree to fund the Unfunded Commitment. Pursuant to a standard Assignment and Assumption Agreement, Lender shall assign to Replacement Lender, without recourse, and Replacement Lender shall assume, Lender's rights and obligations with respect to the Unfunded Commitment and whereupon Lender shall be released therefrom. Lender and Replacement Lender shall execute a co-lender agreement in form and substance reasonably acceptable to Lender and Replacement Lender, providing, among other things, for the priority of the Unfunded Commitment to the previously advanced Loan proceeds.

# Article 5

# INTEREST

# 5.1 Interest Rate.

The Loan shall bear interest as set forth in the Note. Interest shall be paid on the Loan when and as set forth in the Note.

# Article 6

# COSTS OF MAINTAINING LOAN

#### 6.1 Increased Costs and Capital Adequacy.

(a) Borrower recognizes that the cost to Lender of maintaining the Loan or any portion thereof may fluctuate, and Borrower agrees to pay Lender additional amounts to compensate Lender for any increase in its actual costs incurred in maintaining the Loan or any portion thereof outstanding or for the reduction of any amounts received or receivable from Borrower as a result of any change after the date hereof in any applicable Law, regulation or treaty, or in the interpretation or administration thereof, or by any domestic or foreign court, (i) changing the basis of taxation of payments under this Agreement and/or the Note to Lender (other than taxes imposed on all or any portion of the overall net income or receipts of Lender), or (ii) imposing, modifying or applying any reserve (other than a loan loss reserve), special deposit or similar requirement against assets of, deposits with or for the account of, credit extended by, or any other acquisition of funds for loans by Lender (which includes the Loan or any applicable portion thereof) or (iii) imposing on Lender any other condition affecting the Loan, provided that the result of the foregoing is to increase the cost to Lender of maintaining the

Loan or any portion thereof or to reduce the amount of any sum received or receivable from Borrower by Lender under the Loan Documents.

(b) If the adoption after the date hereof of any Law, rule, regulation or guideline regarding capital adequacy, or any change after the date hereof in any of the foregoing, or in the interpretation or administration thereof by any domestic or foreign Governmental Authority, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by Lender with any request or directive regarding capital adequacy (whether or not having the force of law) of any such authority, central bank or comparable agency, has the effect of reducing the rate of return on Lender's capital to a level below that which Lender would have achieved but for such application, adoption, change or compliance, then, from time to time Borrower shall pay to Lender such additional amounts as will compensate Lender for such reduction with respect to any portion of the Loan outstanding.

(c) Any amount payable by Borrower under subsection (a) or subsection (b) of this <u>Section 6.1</u> shall be paid within ten (10) Business Days of receipt by Borrower of a certificate signed by an authorized officer of Lender setting forth the amount due and the basis for the determination of such amount, which statement shall be conclusive and binding upon Borrower absent manifest error. Lender shall also provide to Borrower copies of any applicable invoices, bills, demands or statements of account. Failure on the part of Lender to demand payment from Borrower for any such amount attributable to any particular period shall not constitute a waiver of Lender's right to demand payment of such amount for any subsequent or prior period.

# 6.2 Borrower Withholding.

If by reason of a change in any applicable Laws occurring after the date hereof, Borrower is required by Law to make any deduction or withholding in respect of any taxes (other than taxes imposed on or measured by the net income of Lender or any franchise tax imposed on Lender), duties or other charges from any payment due under the Note to the maximum extent permitted by law, the sum due from Borrower in respect of such payment shall be increased to the extent necessary to ensure that, after the making of such deduction or withholding, Lender receives and retains a net sum equal to the sum that it would have received had no such deduction or withholding been required to be made.

Article 7

# LOAN EXPENSE AND ADVANCES

# 7.1 Loan and Administration Expenses.

Borrower unconditionally agrees to pay all reasonable expenses of the Loan, including all amounts payable pursuant to <u>Sections 7.2</u> and <u>7.3</u> and any and all other fees owing to Lender pursuant to the Loan Documents, and also including, without limiting the generality of the foregoing, all recording, filing and registration fees and charges, Deed of Trust, intangible or documentary taxes, escrow charges, title charges, all insurance premiums, title insurance premiums and other charges of the Title Insurer, printing and photocopying expenses, survey fees and charges, cost of certified copies of instruments, cost of premiums on the Title Policy, charges of the Title Insurer or other escrowee for administering disbursements, all reasonable

fees and disbursements of Lender's Consultant, all appraisal fees, insurance consultant's fees, investigator's fees, environmental consultant's fees, reasonable travel related expenses and all reasonable costs and expenses incurred by Lender in connection with the determination of whether or not Borrower has performed the obligations undertaken by Borrower hereunder or has satisfied any conditions precedent to the obligations of Lender hereunder. Borrower shall pay the airfare and other reasonable travel expenses for each officer or analyst of Lender who inspects the Project as part of Lender's due diligence. The amount charged for airfare shall be the lesser of (i) the actual cost thereof incurred by Lender, or (ii) \$1,200 per person per visit. Borrower agrees to pay all brokerage, finder or similar fees or commissions payable in connection with the transactions contemplated hereby and shall indemnify, defend, and hold Lender harmless against all claims, liabilities, costs and expenses (including attorneys' fees and expenses) incurred in relation to any such claim by broker, finder or similar person alleging to have dealt with Borrower in connection with this transaction.

### 7.2 Loan Fee.

Borrower shall pay to Lender on or before the date of this Agreement a loan fee in the amount of One Million Six Hundred Fifty Thousand Dollars (\$1,650,000), which includes an application fee of Two Hundred Fifty Thousand Dollars (\$250,000) (which was previously received by Lender), a commitment fee in the amount of Two Hundred Fifty Thousand Dollars (\$250,000) (which was previously received by Lender), and a closing fee in the amount of One Million One Hundred Fifty Thousand Dollars (\$1,150,000). Such fees are fully earned and non-refundable. Lender may deduct the closing fee and any costs from the Closing Funding.

# 7.3 Draw Fees.

The first disbursement of funds for Hard Costs associated with construction and each such disbursement thereafter shall be referred to as a "<u>Construction Disbursement</u>." Borrower shall pay to Lender at the time of each Construction Disbursement a draw fee in the amount of Five Thousand Dollars (\$5,000) per disbursement as compensation to Lender for its costs in reviewing and processing each Construction Disbursement. Borrower hereby authorizes Lender to retain such fee from each such disbursement without further direction from Borrower.

# 7.4 Exit Fee.

Borrower shall pay to Lender a fee equal to the amount of \$1,650,000 (the "Exit Fee"), which Exit Fee shall be due and payable to Lender upon the earlier of (i) payment in full of the Loan for any reason (which shall be deemed to include Borrower's cancellation or termination of the Loan prior to any funding after the Closing Funding) or (ii) maturity of the Loan, whether by acceleration or otherwise. Prior to payment in full or maturity of the Loan, the Exit Fee shall be payable at the rate of the full Exit Fee for the release of the Hotel in accordance with the terms of this Agreement, or in installments of \$20,000 per Residential Unit at the closing of the sale of any Residential Unit sold in accordance with the terms of this Agreement (such amounts to be paid out of the Release Price paid to Lender in accordance with <u>Section 14.9</u> until such time as Borrower has paid the Exit Fee in full). Any unpaid portion of the Exit Fee shall be due and payable on the first to occur of the acceleration of the Loan, the Maturity Date or payment in full of the Loan for any reason if prior to the Maturity Date.

### 7.5 Lender's Attorneys' Fees and Disbursements.

Borrower agrees to pay Lender's reasonable attorneys fees and disbursements incurred in connection with this Loan both before and after the date hereof (and whether or not the Loan closes), including (i) the preparation of this Agreement, any intercreditor agreements and the other Loan Documents and the preparation of the closing binders, (ii) the disbursement and administration of the Loan and (iii) the enforcement of the terms of this Agreement and the other Loan Documents. The reasonable legal fees and disbursements to be paid by Borrower under this Section 7.5 and the reasonable legal fees and disbursements to be paid by Borrower under this Section 7.5 and the reasonable legal fees and disbursements to be paid by Borrower under all other applicable provisions of this Loan Agreement and the other Loan Documents shall include the reasonable fees and expenses of Lender's inside counsel charged at a rate not higher than the rate charged by Lender's outside counsel for an attorney with equivalent experience. Without limiting the generality of the foregoing, Borrower hereby acknowledges and agrees that Lender will incur additional expenses, including attorneys fees and disbursements, in connection with any Mezzanine Loan that Borrower proposes to obtain in accordance with this Agreement, and Borrower hereby covenants to pay, at such time and in such manner requested by Lender, all such reasonable expenses of Lender incurred for its review of the proposed and final Mezzanine Loan Documents, drafting and negotiating an intercreditor agreement, and all other reasonable expenses of Lender incurred for its review of the proposed and final Mezzanine Loan, regardless of whether such Mezzanine Loan closes during the term of the Loan.

## 7.6 Time of Payment of Fees and Expenses.

Borrower shall pay all expenses and fees incurred by Lender in connection with the Loan as of the date of this Agreement at Closing. At Closing, Lender may pay from the Closing Funding all reasonable Loan expenses and all fees payable to Lender and not previously paid. Lender may require the payment of outstanding fees and expenses as a condition to any disbursement of the Loan. Lender is hereby authorized, without any specific request or direction by Borrower, to make disbursements from time to time in payment of or to reimburse Lender for all reasonable Loan expenses and fees (whether or not, at such time, there may be any undisbursed amounts of the Loan allocated in the Budget for the same). Lender shall provide to Borrower copies of all applicable invoices relating to such expenses and fees. Lender may charge any such disbursement to any applicable Budget Line Item or, if in Lender's judgment there is no available source of funds in the Budget, may require Borrower to pay excess expenses from Borrower's own funds. If the actual amount of charges are not ascertainable at Closing, then Lender may charge the same, at its option, at Full Loan Opening or to the Soft Cost Contingency or other applicable Budget Line Item, and Borrower shall pay upon demand any excess monies due.

## 7.7 Expenses and Advances Secured by Loan Documents.

Any and all advances or payments made by Lender under this <u>Article 7</u> from time to time, and any amounts expended by Lender pursuant to <u>Section</u> <u>20.1(a)</u>, shall, as and when advanced or incurred, constitute additional indebtedness evidenced by the Note and secured by the Deed of Trust and the other Loan Documents and shall bear interest at the rate then applicable under the Note (including the Default Rate, when applicable).

### 7.8 Right of Lender to Make Advances to Cure Borrower's Defaults.

In the event that Borrower fails to perform any of Borrower's covenants, agreements or obligations contained in this Agreement or any of the other Loan Documents (and applicable grace or cure periods have expired, unless in Lender's judgment an emergency or other exigent circumstance exists, in which case Lender need not wait for such period to expire), Lender may (but shall not be required to) perform any of such covenants, agreements and obligations, and any reasonable amounts expended by Lender in so doing and shall constitute additional indebtedness evidenced by the Note and secured by the Deed of Trust and the other Loan Documents and shall bear interest at the Default Rate. Lender may expend any such amounts even if such advance would result in the balance owing to Lender exceeding the stated amount of the Note.

## Article 8

## NON-CONSTRUCTION REQUIREMENTS PRECEDENT

#### 8.1 Non-Construction Conditions Precedent.

Borrower agrees that Lender's obligation to make the Loan is conditioned upon Borrower's delivery, performance and satisfaction of the following conditions precedent in form and substance satisfactory to Lender in its reasonable discretion (except as otherwise expressly set forth herein) prior to (or at the time of) Closing, with the exception of the conditions listed in Section 8.1(a), Section 8.1(u), Section 8.1(w), Section 8.1(y)(b) and Section 8.1(z)(ii) and (iii) hereunder that must be satisfied at such other time as is set forth therein:

(a) Equity: Borrower shall have provided evidence reasonably satisfactory to Lender prior to Full Loan Opening (which, for the avoidance of doubt, shall exclude the Closing Funding) that Borrower's equity invested in the Project for costs in the Budget is not less than \$124,412,805.92 in cash (the "Initial Equity Investment"), of which, Guarantor or its Affiliates shall contribute at least \$49,165,122.37, and CJUF shall contribute the balance; provided, however, that Lender shall permit \$1,500,000 (which amount shall be credited pro rata to the respective contributions of Guarantor and CJUF) of the Initial Equity Investment to be deferred until on or after the repayment in full of the Loan, provided that the Developer Fee Budget Line Item is paid only in strict accordance with Section 12.7 hereof. Such Initial Equity Investment must be used to pay Project costs contained in the Budget and approved by Lender, with evidence of payment of the Initial Equity Investment to be delivered to Lender prior to Full Loan Opening. In all events, any equity contribution shall be subordinate to the rights of Lender and general unsecured creditors of Borrower. Borrower may not be indebted to any Person for any equity contribution. Borrower shall provide Lender with documentation identifying all equity Investment, provided same appear in the Project Budget and such expenses (which actual cost and expenses shall be credited against the Initial Equity Investment, provided same appear in the Project Budget and such expension), such documentation to be in form and substance acceptable to Lender, in its sole discretion).

Borrower's Initial Equity Investment was increased by \$1,505,000 (the "Equity Increase") to a total of \$124,412,805.92 so as to permit the creation of the "Sales Commissions Paid During Construction" Budget Line Item in the amount of \$1,505,000. (\$460,179 of this Budget Line

Item has already been funded and the remainder is to be funded at a rate of approximately \$35,000 per month as an advance against brokerage commissions for Unit sales.) Notwithstanding anything to the contrary contained herein, so long as no Event of Default exists, at the time of Unit closings up to fifty percent (50%) of the brokerage commissions payable upon each such closing may be used to return to Borrower the Equity Increase. The foregoing shall not reduce any Release Price payable to Lender.

If Borrower obtains a Mezzanine Loan with the consent of Lender and otherwise in accordance Section 15.1(u) hereof, the Initial Equity Investment may be reduced by the Qualifying Portion of such Mezzanine Loan; provided, however, that such reduction in the Initial Equity Investment shall only be applicable to such portion of the Initial Equity Investment that has not been contributed at the time of the closing of the Mezzanine Loan, such that no part of the Initial Equity Investment may be recouped by Borrower or other contributor, and Borrower hereby acknowledges and agrees that, as a result, less than all of the reduction in the Initial Equity Investment may be realized. The "Qualifying Portion" of the Mezzanine Loan shall mean the portion available for borrowing to pay costs in the Budget attached as Exhibit G hereto (e.g., the Qualifying Portion shall not include items such as interest and/or fees and expenses payable with respect to the Mezzanine Loan). The Qualifying Portion must be fully funded prior to Full Loan Opening.

Borrower's Equity Investment requirement shall be increased as a result of any changes to the Budget and as necessary to maintain the Loan In Balance as described in <u>Section 11.1</u>. Lender shall not be required to disburse any proceeds to reimburse Borrower for its Equity Investment unless Lender has determined that Borrower has invested amounts in excess of its required Equity Investment. In the event that Borrower, or such other Person on behalf of Borrower, invests such funds that render the total actual equity investment to be in excess of the required Equity Investment (excluding the Additional Equity Investment or any equity deposited as a result of the Loan being not In Balance), the Commitment shall be permanently reduced by such excess investment, unless Borrower delivers to Lender a written request for a refund of such excess within thirty (30) days of such excess investment, but in no event shall such a request be required sooner than thirty (30) days after the date hereof.

(b) Fees and Expenses: Borrower shall have paid all of Lender's fees and expenses as required by Article 7 or elsewhere in this Agreement, to the extent due and payable.

(c) <u>Title and Other Documents</u>: Borrower shall have furnished to Lender a Commitment for the Title Policy with the premiums for such Title Policy paid in full, and a pro forma Title Policy, with coverage effective as of Closing, together with legible copies of all title exception documents cited in the Commitment for the Title Policy and all other legal documents affecting the Project or the use thereof. Borrower shall provide to Lender the final original Title Policy within fifteen (15) days following Closing. The pro forma Title Policy and the Title Policy shall be subject only to the Permitted Exceptions. Any exception for the rights of Residential Unit Purchasers shall only be permissible if the Title Insurer insures such rights are subordinate to the Deed of Trust.

(d) Survey: Borrower shall have furnished to Lender an ALTA/ACSM "Class A" Land Title Survey of the Project prepared by a licensed surveyor satisfactory to Lender. Said

survey shall be dated no earlier than ninety (90) days prior to the date hereof, shall be made (and certified to have been made) as set forth in Exhibit D attached hereto and made a part hereof. Such survey shall be sufficient to permit issuance of the Title Policy in the form required by this Agreement. Such survey shall include the legal description of the Land.

(e) <u>Insurance Policies</u>: Borrower shall have furnished to Lender, prior to the date hereof, certificates evidencing that insurance coverages are in effect with respect to the Project and Borrower, in accordance with the Insurance Requirements attached hereto as <u>Exhibit E</u> and incorporated herein by reference as if fully set forth herein (or such other insurance coverages reasonably acceptable to Lender), for which the premiums have been prepaid, and with endorsements satisfactory to Lender. On or before the date that is thirty (30) days after the date hereof, Borrower shall provide a copy of the insurance policy with respect to the Project and Borrower in accordance with <u>Exhibit E</u> (the "<u>Insurance Policy</u>").

(f) <u>No Litigation</u>: Borrower shall have furnished evidence that no litigation or proceedings shall be pending or threatened that is reasonably likely to cause a Material Adverse Change with respect to Borrower, Guarantor or the Project, other than litigation that is disclosed in writing to Lender prior to the Loan closing and is acceptable to Lender, in its sole discretion.

(g) <u>Utilities</u>: Borrower shall have furnished to Lender (by way of utility letters or otherwise) evidence establishing to the reasonable satisfaction of Lender that the Project, when constructed, will have adequate water supply, storm and sanitary sewerage facilities, telephone, gas (if applicable), electricity, fire and police protection, means of ingress and egress to and from the Project and public highways and any other required public utilities and that the Project is benefited by insured easements as may be required for any of the foregoing.

(h) <u>Attorney Opinions</u>: Borrower shall have furnished to Lender an opinion from counsel for Borrower and Guarantor, in a form satisfactory to Lender, covering due authorization, execution and delivery and enforceability of the Loan Documents and also containing such other legal opinions as Lender shall reasonably require, with customary assumptions and qualifications.

(i) <u>Appraisal</u>: Lender shall have obtained, at Borrower's expense, an Appraisal acceptable to Lender in all respects, in Lender's sole discretion. The Appraisal of the Project performed by CB Richard Ellis of Austin, dated March 5, 2008, is hereby accepted by Lender. A copy of such Appraisal will be supplied to Borrower upon request, subsequent to Closing.

(j) <u>Searches</u>: Borrower shall have furnished to Lender current bankruptcy, federal tax lien and judgment searches and searches of all Uniform Commercial Code financing statements filed in each place UCC Financing Statements are to be filed hereunder for Borrower and Guarantor, demonstrating the absence of materially adverse claims.

(k) <u>Financial Statements</u>: Tax Returns: Borrower shall have furnished to Lender current annual financial statements of Borrower and Guarantor, each in form and substance and certified by such individual as acceptable to Lender. Borrower shall have furnished to Lender any federal and state tax returns of Borrower or Guarantor for the past two (2) years. Borrower and Guarantor shall have signed and delivered to Lender an Internal Revenue Service Tax Return

Verification Form (IRS Form 4506-T). Borrower and Guarantor shall provide such other additional financial information as Lender reasonably requires, including financial statements of income and expenses for the Project and tax returns for all entities reporting the income and expenses on the Project.

(1) Price List Schedule. Borrower shall have furnished to Lender the Price List Schedule, as approved by Lender.

(m) <u>Other Agreements</u>: Borrower shall have delivered to Lender executed copies of any marketing, brokerage and development agreements entered into by Borrower in connection with the Construction and/or the sale of Residential Units at, or any other part of, the Project, each of which Lender shall have approved in Lender's reasonable discretion.

(n) Flood Hazard: Lender has received evidence that the Project is not located in an area designated by the Secretary of Housing and Urban Development as a special flood hazard area, or flood hazard insurance acceptable to Lender in its reasonable discretion.

(o) <u>Zoning</u>: The Title Policy shall include an ALTA 3.1 zoning endorsement modified for plans and specifications, or, if not available in the State, Borrower shall have furnished to Lender a legal opinion that provides the same zoning compliance assurance as an ALTA 3.1 zoning endorsement subject to customary qualifications, clarifications and assumptions.

(p) <u>Organizational Documents</u>: Borrower shall have furnished to Lender proof satisfactory to Lender of authority, formation, organization and good standing in the state of its incorporation or formation and, if applicable, qualification as a foreign entity in good standing in the state of its incorporation or formation, of all corporate, partnership, trust and limited liability company entities (including Borrower and Guarantor) executing any Loan Documents, whether in their own name or on behalf of another entity. Borrower shall also provide an organizational chart as well as certified resolutions in form and content satisfactory to Lender, authorizing execution, delivery and performance of the Loan Documents, and such other documentation as Lender may require to evidence the authority of the persons executing the Loan Documents.

(q) No Default; No Material Adverse Change; No Condemnation; etc.: There shall be no Default or continuing Event of Default by Borrower hereunder; there shall have not occurred a Material Adverse Change in the financial condition of Borrower or Guarantor or the condition of the Project that has not been cured or satisfied; and neither the Project nor any part thereof shall have suffered any material casualty or be subject to any existing or threatened condemnation or taking by eminent domain proceeding or otherwise.

(r) <u>Easements</u>: Borrower shall have furnished to Lender all easements reasonably required for the construction, maintenance or operation of the Project, and such easements shall be insured by the Title Policy.

(s) <u>Condominium Documents</u>: Borrower shall have delivered to Lender a copy of all proposed and/or executed condominium plats, declarations, agreements regarding cost sharing, filings, escrow agreements and other documents pertaining to the establishment of a condominium regimes at the Project or relating to the Project's compliance with all applicable

local, state and federal Laws relating to condominiums (collectively, the "<u>Condominium Documents</u>"), including, without limitation, the draft Master Condominium Declaration and Residential Condominium Declaration, all of which Condominium Documents shall be acceptable to Lender in its sole and absolute discretion. Lender shall be satisfied with the sufficiency of the services and infrastructure provided and the adequacy of their funding.

(t) [Intentionally Omitted]

(u) Lease. Borrower shall have delivered to Lender the form of Lease to be used for the leases of space in the Retail Space and in the Office Space.

(v) <u>Pre-sales</u>. Borrower shall have delivered to Lender, prior to Closing, Qualifying Sales Agreements (in full force and effect) for at least 55 Residential Units comprising at least 70,761 Saleable Square Feet together with Parking Spaces for such applicable Residential Units at prices greater than or equal to the List Price for each and every such Residential Unit, constituting, in the aggregate, gross sales of greater than or equal to \$45,184,250 (collectively, the "<u>Pre-sale Requirement</u>"). The mix of Residential Unit types and locations within the Project must be reasonably acceptable to Lender.

(w) Earnest Money and Upgrade Deposits. All existing Earnest Money Deposits and Upgrade Deposits shall have been deposited by the Escrow Agent, or by Lender to the extent allowed by Law, in accounts in accordance with <u>Section 14.3</u>. If an Escrow Agent is being used, Borrower shall have delivered to Lender a copy of the escrow agreement pursuant to which the Earnest Money Deposits are being held (the "<u>Escrow Agreement</u>"), together with a letter from the Escrow Agent agreeing to deliver such deposits to Lender when and as Borrower has a right to receive such deposits for application in accordance with <u>Section 14.3</u> and <u>Section 14.9</u> below. Prior to Full Loan Opening, Borrower shall have deposited with either the Escrow Agent or Lender, as applicable, cash Earnest Money Deposits equal to at least ten percent (10%) of the contract price for each Residential Unit included in the Pre-sale Requirement.

(x) <u>Required Condominium Approvals</u>. Borrower shall have furnished to Lender evidence satisfactory to Lender that Borrower has received all approvals required for the condominium subdivision of the Project pursuant to the Declaration of Condominium, the sale or marketing of the Residential Units under the requirements of ILSA, applicable laws of the State and any applicable local Laws. Borrower also shall have furnished to Lender as filed copies of Borrower's HUD Property Report and any additional reports required by the State of Texas or City of Austin.

(y) <u>Operating Agreements</u>. Borrower shall have provided to Lender (a) the executed Hotel Documents, Non-Disturbance Agreement and an estoppel letters addressed to Lender with respect to the Hotel Documents, and (b) before Full Loan Opening, (x) the management agreement for the management and operation of the Venue, together with a subordination, estoppel and collateral assignment thereof and (y) an amendment to the Hotel Operating Agreement extending the outside completion date thereunder to January 7, 2012; all of the foregoing in form and substance reasonably satisfactory to Lender

(z) <u>City Estoppel and Agreement</u>. (i) Prior to Closing, Borrower shall have provided to Lender an Estoppel Certificate and Agreement from the City of Austin, a Texas home rule city and municipal corporation ("City") in form and substance satisfactory to Lender addressing certain matters relative to (A) that certain Declaration of Restrictive Covenants dated as of December 15, 2006, by the City, recorded in the official public records of Travis County, Texas as Instrument No. 2006240877; and (B) Special Warranty Deed dated as of December 15, 2006, by the City, in favor of Stratus Block 21 Investments, L.P., a Texas limited partnership, recorded in the official public records of Travis County, Texas as Instrument No. 2006240878; (ii) prior to Full Loan Opening, a direction by Borrower to the City to pay any repurchase price payable under such deed to Lender (to be applied to the indebtedness under the Loan Documents in such order as Lender shall elect, with any excess payable to Borrower), which direction is accepted by the City and (iii) prior to Full Loan Opening, Borrower shall have provided to Lender a Second Estoppel Certificate and Agreement from the City in all material respects in the form of <u>Exhibit S</u> or otherwise in form and substance satisfactory to Lender.

(aa) <u>Patriot Act</u>: Borrower shall have provided Lender with proof that Borrower has complied in all respects with the provisions of the USA PATRIOT Act of 2001, as applicable, including without limitation, furnishing to Lender proof that Borrower has taken all action necessary to comply with Section 326 of such Act.

(bb) <u>Additional Documents</u>. Borrower shall have furnished to Lender such other materials, documents, papers or requirements regarding the Project, Borrower or Guarantor as Lender shall reasonably request.

### Article 9

## CONSTRUCTION REQUIREMENTS PRECEDENT

### 9.1 Construction Documents Required as of Closing.

Borrower shall cause to be furnished to Lender and to Lender's Consultant the following, in form and substance satisfactory to Lender in its reasonable discretion (except as otherwise expressly set forth herein), and Lender shall have approved the following in its reasonable discretion (except as otherwise expressly set forth herein), prior to (and at the time of) Closing (or such later time as is specified in any subparagraph hereof) as additional conditions to Lender's obligations to make any disbursements of the Loan:

(a) Fully executed copies of the following shall be delivered: (i) prior to the Closing, a general contract with the General Contractor pertaining to the construction of the Project (the "<u>General Contract</u>"), which must be acceptable to Lender in all respects in Lender's sole discretion, with a guaranteed maximum price not to exceed \$186,889,641 (the "<u>General Contract Price</u>"), and the General Contract shall not be entered into until after the Deed of Trust is recorded; (ii) [intentionally omitted]; (iii) copies of all other direct contracts to be entered into by Borrower for construction, purchase of materials or furniture, fixtures or equipment; and (iv) all contracts with the architects, engineers, third-party owner's representatives and other design professionals (the "<u>Design Professionals</u>") acceptable to Lender in all respects in Lender's reasonable discretion. None of General Contractor and any Design Professionals shall be an Affiliate of Borrower or of Guarantor, and all shall be of acceptable credit quality, as determined

by Lender in its sole discretion. If Lender shall determine that the General Contractor is not of acceptable credit quality, Lender may require, in its sole discretion, certain guaranties or other assurances from the parent entity or entities or Affiliates of General Contractor. Lender hereby approves the General Contract and approves Austin Building Company as General Contractor, subject to the execution and delivery of a guaranty of the General Contract, in form and substance satisfactory to Lender in its sole discretion, from General Contractor's parent company, Austin Commercial, L.P., a Delaware limited partnership. General Contractor may not be replaced without Lender's prior written consent. No Design Professional may be replaced without Lender's prior written consent, subject to its reasonable discretion;

(b) A schedule of values, as included in the General Contract;

(c) [Intentionally Omitted];

(d) In lieu of Bonds, Lender agrees to accept subguard insurance with terms, limits, and endorsements (including a Financial Interest Endorsement naming Lender) acceptable to Lender in its reasonable discretion covering all Subcontracts;

(e) Copies of each of the Required Permits, except for those Required Permits that cannot be issued until a later stage or completion of Construction, in which event such Required Permits will be obtained by Borrower on a timely basis in accordance with all recorded maps and conditions and applicable building, land use, zoning and environmental codes, statutes and regulations and will be delivered to Lender promptly thereafter;

(f) The Partial Plans and Specifications and the Partial Proposed Finish Standards, each of which Lender approves as of the date hereof;

- (g) The Construction Schedule;
- (h) The Soil Report;

(i) The Environmental Report, which shall, at a minimum, (i) demonstrate the absence of any existing or potential Hazardous Material contamination or violations of environmental Laws at the Project, except as acceptable to Lender in its sole and absolute discretion, (ii) include the results of all sampling or monitoring to confirm the extent of existing or potential Hazardous Material contamination at the Project, including the results of leak detection tests for each underground storage tank located at the Project, if any, (iii) describe response actions appropriate to remedy any existing or potential Hazardous Material or underground storage tanks from the Project was completed in accordance with applicable Laws, (v) confirm whether or not the Land is located in a wetlands district, and (vi) comply with the USEPA "all appropriate inquiry" rule contained in 40 C.F.R. Part 312. Borrower shall also have caused to be furnished to Lender any environmental disclosure statement required pursuant to the law of the State;

(j) A report from Lender's Consultant that contains an analysis of the Approved Plans and Specifications, the Budget, the Construction Schedule, the General Contract, all subcontracts then existing and the Soil Report. Such report shall be solely for the benefit of

Lender and shall contain (i) an analysis satisfactory to Lender demonstrating the adequacy of the Budget to complete the Project and (ii) a confirmation that the Construction Schedule is realistic. Lender's Consultant shall monitor construction of the Project and shall visit the Project at least one (1) time each month, and shall certify as to amounts of construction costs for all requested fundings; each report of Lender's Consultant is for the sole benefit of Lender and Lender shall not be bound by any recommendation or conclusion of Lender's Consultant;

- (k) Each Architect's Certificate;
- (l) The Budget, as approved by Lender pursuant to Article 10 hereof;

(m) Original executed consents, in form and substance satisfactory to Lender, of the General Contractor, the Architect, and any other Design Professional to the Collateral Assignment of Construction Documents; and

(n) Such other papers, materials and documents as Lender may reasonably require with respect to the Construction, the Project, Borrower, or Guarantor.

## 9.2 Construction Deliveries Required as of Full Loan Opening.

Prior to Full Loan Opening, in addition to fully satisfying the conditions set forth above in Section 9.1, Borrower shall also satisfy the following conditions:

(a) General Contractor shall have affirmed in writing the validity of the General Contract and described any Change Orders or pending Change Orders as of the Full Loan Opening. As of Full Loan Opening, General Contractor shall provide to Lender the then-current guaranteed maximum price based upon the 100% complete and final Approved Plans and Specifications (and 100% buyout of Subcontracts based upon such 100% complete and final Approved Plans and Specifications (and 100% buyout of Subcontracts based upon such 100% complete and final Approved Plans and Specifications of Allowances as set forth in this subsection (the "<u>Revised GMP</u>"). After giving effect to such Revised GMP and Change Orders, if the Loan is not In Balance, Borrower shall contribute any additional equity required to put the Loan In Balance (including, without limitation, the available amount under the Owner's Hard Cost Contingency being equal to or greater than 4% of the remaining Project hard costs). An additional condition to Full Loan Opening shall be that no more than two percent (2%) of the total General Contract may be an "Allowance" (as such term is defined under the AIA A201-1997 General Conditions of the Contract for Construction) (excluding general conditions, General Contractor's fees and the Contractor's Contingency) and such allowances shall be subject to Lender's reasonable approval;

(b) An initial sworn statement of the General Contractor, approved by Borrower, and Lender covering all work done and to be done, together with lien waivers covering all work and materials for which payments have been made by Borrower prior to the Full Loan Opening;

(c) An update of the consultant's report set forth in <u>Section 9.1(j)</u> above;

(d) Executed contract(s) for all owner direct cost items signed by the applicable suppliers of such items, including, without limitation, those for the purchase of the furniture, fixtures and equipment for the Project.

(e) In addition to the Partial Plans and Specifications addressed in <u>Section 9.1(f)</u> above, full and complete detailed plans and specifications for the Improvements in duplicate, prepared by the Architect (the "<u>Plans and Specifications</u>"). Without limiting the foregoing, such Plans and Specifications must be predicated on, among other things, the Partial Plans and Specifications approved by Lender, and the characteristics in the Recitals of this Agreement and the floor layouts and elevations substantially similar to those which have been previously submitted to Lender. Lender must also be satisfied, in its reasonable discretion, that the specifications, project manual, floor plans, mechanical, electrical, plumbing, fire protection and life safety, structural and site plans are sufficient and appropriate for the Project. Borrower shall deliver packages of such Plans and Specifications to Lender in accordance with the delivery schedule attached hereto as <u>Exhibit R</u>. Lender shall review each such package of Plans and Specifications and Specifications required under <u>Exhibit R</u>, for Lender's final review and approval in its reasonable discretion, as more specifically addressed in Section 9.1(f) above, at least sixty (60) days prior to the expected day of Full Loan Opening. Upon Lender's written approval of such final package of Plans and Specifications, such final Plans and Specifications shall be the "<u>Approved Plans and Specifications</u>". Borrower shall have obtained Lender's approval of the Approved Plans and Specifications proved Plans and Specifications shall be permitted without Lender's prior written approval;

(f) Borrower shall have furnished to Lender and Lender's Consultant, for Lender's approval, detailed Plans and Specifications based upon the Partial Proposed Finish Standards and setting forth a detailed description and quality level for materials used for the type of construction, the façade and the finishes of each Residential Unit, and throughout the Hotel, and of all fixtures and personal property that will be included in the standard price of the individual Residential Units, and throughout the Hotel, including, without limitation, items such as floor coverings, wall coverings, electrical systems, lighting plans, HVAC systems, bathroom and kitchen fixtures and countertops, cabinetry, appliances and furniture (collectively, "<u>Proposed Finish Standards</u>"). Lender must be satisfied, in its reasonable discretion, that the quality level of the Proposed Finish Standards is comparable to other condominium developments or hotels, as applicable, in the same price range and located in Austin, Texas, that the Proposed Finish Standards are consistent with the Plans and Specifications submitted to Lender, and that the Hotel finish standards are in accordance with the Hotel Documents. Borrower shall have delivered the final Proposed Finish Standards for Lender's final review and approval in its reasonable discretion at least sixty (60) days prior to Full Loan Opening, and Lender shall review and approve or comment on same within a reasonable period of time following its receipt thereof. Borrower shall have obtained Lender's approval of such Proposed Finish Standards prior to Full Loan Opening, in which event such standards shall constitute the "<u>Approved Finish Standards</u>." No material changes to the Approved Finish Standards shall be permitted without the prior written approval by Lender, to be determined in Lender's reasonable discretion. Borrower shall finish all Residential Units and the Hotel to the Approved Finish Standards as part of the Construction required hereunder;

(g) Borrower has provided evidence to Lender that all of the work, other than that certain water filtration system, contemplated by that certain Waste Management Plan dated

March 4, 2008 and prepared by Terracon Consultants, Inc., and any amendments or updates thereto (the "<u>Waste Management Plan</u>"), has been completed or otherwise addressed to Lender's reasonable satisfaction, such evidence to include, but not be limited to, a written statement from an authorized representative of Terracon Consultants, Inc., affirming that the recommendations of such Waste Management Plan were complied with and no further actions, other than the ongoing use and maintenance of the water filtration system during the operation of the Project, are required to address the conditions at the Project site that were disclosed in any of Environmental Reports delivered to Lender.

(h) Hotel Operator, and, as may be required by the Hotel Documents, Starwood Hotels & Resorts Worldwide, Inc., shall have approved the Approved Plans and Specifications and Approved Finish Standards; and

(i) To the extent not previously delivered at Closing, the Required Permits, and updates thereof, which shall include the final full building permits for the Project no later than thirty (30) days prior to Full Loan Opening.

## Article 10

#### BUDGET, CONTINGENCY FUND AND CHANGE ORDERS

### 10.1 Budget.

Disbursement of the Loan shall be governed by the Budget for the Project, in form and substance acceptable to Lender in Lender's reasonable discretion. Borrower shall only be entitled to disbursements that are in accordance with the Budget. The Budget shall specify the amount of cash equity invested in the Project, and all costs and expenses of every kind and nature whatsoever to be incurred by Borrower in connection with the Project. Costs associated with Residential Unit sales (including but not limited to broker's commissions, closing and escrow costs) may be paid from proceeds resulting from any Residential Unit closing to the extent the gross sales price paid by any Residential Unit Purchaser for its respective Residential Unit exceeds the applicable Release Price; otherwise, such closing costs shall be paid in cash by Borrower. The Budget shall include, in addition to the Budget Line Items described in Section 10.2 below, the Contingency Fund described in Section 10.3 below and amounts satisfactory to Lender for Hard Costs, Soft Costs and other reserves reasonably acceptable to Lender. The construction trade line items contained in the General Contract (or in the schedule of values) shall each be deemed a Budget Line Item for purposes of this Agreement. The Budget is attached hereto as Exhibit G and made a part hereof. Except as set forth in this Agreement, all changes to the Budget shall in all respects be subject to the prior written approval of Lender, which approval shall be granted or withheld in Lender's reasonable discretion. Borrower shall promptly notify Lender of any anticipated changes in the line items of the Budget that, if approved, would result in a net increase in the total amount of the Budget and Borrower shall not enter into any agreement that would increase the total amount of the costs in the Budget without Lender's prior written consent. In the event the total Project costs are less than the final Budget, Borrower shall have no right to borow the balance of the Loan not needed for Pr

### 10.2 Budget Line Items.

(a) The Budget shall include as line items ("Budget Line Items"), to the extent determined to be applicable by Lender in its reasonable discretion, the cost of all labor, materials, equipment, fixtures and furnishings needed for the completion of the Construction, and all other costs, fees and expenses relating in any way whatsoever to the Construction of the Improvements, marketing and sales costs, commissions, operating deficits, real estate taxes, and all other sums due in connection with Construction and operation of the Project, the Loan, and this Agreement. Each line item in the trade breakdown of the General Contract shall be considered a separate Budget Line Item for all purposes of this Agreement whether or not separately shown as Budget Line Items on the Budget attached hereto. Borrower agrees that all Loan proceeds disbursed by Lender shall be used only for the Budget Line Items for which such proceeds were disbursed, except as reallocated in accordance with this Agreement or otherwise permitted by Lender in its reasonable discretion. The Budget shall not contain any line items payable to Borrower, Guarantor or any Affiliate of either Borrower or Guarantor and Borrower and Guarantor shall not pay or cause to be paid any Loan proceeds to any Affiliate of either, except for the Permitted Affiliate Expenses. The Budget shall include as Budget Line Items, outside of Owner's Hard Cost Contingency and the Soft Cost Contingency, (i) all fees and sums payable to Hotel Operator or its Affiliates pursuant to the Hotel Documents, including without limitation, (A) the marketing assistance fee of \$1,250 per Residential Unit payable in twenty-four (24) equal monthly installments commencing upon the effective date of the Condominium Marketing License Agreement, (B) pre-opening services budget of \$2,536,085, (C) pre-opening information technology budget of \$1,261,604 (which shall be included within the TV Lease and Technology Line Item), (D) pre-opening inventories budget of \$3,276,000, (E) FF&E of \$15,000 per Hotel key, (F) technical services fees of \$513,000, consisting of \$1,250 per Hotel key, plus \$1,000 per Residential Unit, and (G) reimbursable expenses under the Technical Services Agreement up to \$5,000 per Hotel key and initial working capital budget of \$1500 per Hotel key, but excluding the Condominium Licensing Fee of 4.5% of gross sales revenue per Residential Unit, which shall be payable to Hotel Operator from sales proceeds, as opposed to the Budget, and (ii) all costs contemplated by the Waste Management Plan.

(b) Borrower shall have the right to reallocate cost savings effected by a final Change Order or other appropriate final documentation to other Budget Line Items subject to (x) Lender's prior written consent, in its reasonable discretion, and (y) the limits contained in this <u>Section 10.2</u> and <u>Section 10.3</u> of this Agreement. No reallocations shall be permitted to or from the Interest Reserve Budget Line Item, Developer Fee Budget Line Item or for any amounts payable to Borrower, Guarantor or any Affiliates of Borrower or Guarantor. If there is a savings in Hard Costs upon completion of the construction work contemplated by the General Contract, as determined by Lender in its reasonable discretion, savings may be reallocated to Owner's Hard Cost Contingency (except to the extent a portion of such savings are paid to the General Contractor pursuant to the terms of the General Contract). Notwithstanding the foregoing in the immediately preceding sentence and subject to Lender's reasonable approval, Borrower may reallocate any final savings in Hard Costs, which accrue prior to completion of the Project, to the Owner's Hard Cost Contingency for use by Borrower for other Hard Costs. If the total and final expenditures for any Soft Cost Budget Line Item are less than the amount provided for in the Budget, then the savings may be reallocated to the Soft Cost Contingency. In the event that the final costs of the Project

are less than the total amount of sources of funds in the Budget (including the Loan), the amount of the cost savings shall not be available for borrowing (e.g., as a return of equity).

(c) Except as reallocated in accordance with this Agreement or otherwise permitted by Lender in its reasonable discretion, Lender shall not be obligated to disburse any amount for any category of costs set forth as a Budget Line Item that is greater than the amount set forth for such category in the applicable Budget Line Item. Borrower shall pay as they become due all amounts set forth in the Budget with respect to costs to be paid for by Borrower.

## 10.3 Contingency Fund.

The Budget shall contain Budget Line Items available to Borrower for (i) payment of Hard Costs, separate and apart from the General Contract (the "<u>Owner's Hard Cost Contingency</u>") and (ii) for payment of Soft Costs (the "<u>Soft Cost Contingency</u>." together, the Owner's Hard Cost Contingency and the Soft Cost Contingency constitute the "<u>Contingency Fund</u>"). At Closing, the funds in the (i) Owner's Hard Cost Contingency Budget Line Item must be in an amount equal to \$8,000,000, and at no time shall the Owner's Hard Cost Contingency Budget Line Item must be in an increase in the initial contractual amount of such General Contract, and (b) the Soft Cost Contingency Budget Line Item must be in an amount equal to \$750,000. At Full Loan Opening, the remaining unallocated Owner's Hard Cost Contingency must equal at least 4% of the unpaid cost of all construction hard cost Budget Line Items.

Lender shall not be obligated to disburse or reallocate all or any part of the Contingency Fund, except as set forth herein. The Owner's Hard Cost Contingency may be used by Borrower to pay the cost of Change Orders (which have been approved by Lender or as to which Lender's approval is not required under <u>Section 10.5</u>) or reallocated to other Hard Costs (in accordance with this Agreement) on a <u>pro rata</u> basis over the course of Construction, according to the percentage of construction trade items (which shall equal the amount of the General Contract Price less general conditions, the General Contractor's fee and any Contractor's Contingency) expended by Borrower and completed to such date. Funds from the Soft Cost Contingency may be reallocated to pay other Soft Costs, excluding the Developer Fee Budget Line Item (and any other amounts payable to Borrower, Guarantor, or any of their Affiliates) and the Interest Reserve Budget Line Item (i) without Lender's approval on a pro rata basis such that the cumulative percentage of such reallocated funds is equal to or less than the percentage of completion of the Project, and (ii) otherwise subject to Lender's prior written approval, in its reasonable discretion.

The Budget Line Item for amounts payable to the General Contractor under the General Contract may also include a contingency amount that is available to the General Contractor (the "<u>Contractor's Contingency</u>"). This is in addition to the Contingency Fund, which is intended to afford protection to Lender, and any Contractor's Contingency may not be counted towards the amounts required to be contained in the Contingency Fund by the preceding paragraphs of this <u>Section 10.3</u>. The Contractor's Contingency shall be used in accordance with the terms of the General Contract.

### 10.4 Optional Method for Payment of Interest.

For Borrower's benefit, the Budget includes a Budget Line Item for interest on the Loan (the "Interest Reserve Budget Line Item"). (Neither the Interest Reserve Budget Line Item nor any other part of the Loan shall be available to pay interest on any Mezzanine Loan.) Borrower hereby authorizes Lender from time to time, for the mutual convenience of Lender and Borrower, to disburse Loan proceeds to pay all the then accrued interest on the Note when due, regardless of whether Borrower shall have specifically requested a disbursement of such amount. Any such disbursement, if made, shall be added to the outstanding principal balance of the Note and shall, when disbursed, bear interest pursuant to the Note. Borrower may borrow the Interest Reserve Budget Line Item to pay interest on the Loan upon qualifying for such disbursement by satisfying the applicable requirements of <u>Section 4.1(c)</u> hereof. Without limiting the foregoing, or the provisions of <u>Article 11</u> below, once the Interest Reserve Budget Line Item has been expended, Borrower shall pay interest from its own funds. In the event the Project, during any month during the term of the Loan, achieves a positive Net Operating Income, such Net Operating Income shall be applied to the payment of interest before the funds in the Interest Reserve Budget Line Item are used to pay such interest. If the Net Operating Income exceeds the monthly interest then due, the excess shall continue to be held in the operating account for the Project established pursuant to <u>Section 15.1(ii)</u>, and such excess shall be applied to pay the interest accrued on the Loan for the following month.

### 10.5 Change Orders.

Borrower agrees that no changes will be made in the Approved Plans and Specifications (or in such Plans and Specifications as have been approved by Lender from time to time) without the prior written approval of Lender in Lender's reasonable discretion, and Borrower must notify Lender in writing of any such change at least seven (7) days prior to date such change order will be implemented, provided, however, that Borrower may make changes to the Approved Plans and Specifications without Lender's approval if (a) Borrower notifies Lender in writing of such change in no event later than the next construction draw following Borrower's agreement to the changes; (b) Borrower obtains the approval of all parties whose approval is legally required, including any affected Residential Unit Purchaser, sureties, and any Governmental Authority to the extent approval from such parties is required; (c) the structural integrity of the Improvements is not impaired; (d) no change in architectural appearance is effected; (e) the performance of the mechanical, electrical, and life safety systems of the Improvements of the Project and as a result of such change there would be no diminution in value or impairment of the marketability of the Project as described in the Approved Plans and Specifications; (g) the cost of, or reduction in cost, resulting from any such change does not exceed \$200,000; and (h) the Loan will remain In Balance (after giving effect to any reallocation of the Contingency Fund or other Budget Line Items that Borrower has qualified for pursuant to this <u>Article 10</u>). Changes in the scope of construction work or to any construction related contract must be documented with a Change Order on the AIA Form G 701 or equivalent form. Borrower shall not agree to or implement any Change Order without the prior written consent of Hotel Operator (or its Affiliate) where such approval is required under any of the Hotel Documents.

### Article 11

## SUFFICIENCY OF LOAN

### 11.1 Loan In Balance.

(a) The Loan is required to be In Balance at all times. The Loan shall be "<u>In Balance</u>" only when both (i) the Available Sources of Funds equal or exceed Lender's Estimate of Remaining Costs and (ii) each Budget Line Item is sufficient to pay the costs such Budget Line Item was established to pay for (including, with respect to the Interest Reserve Budget Line Item interest in the Loan through the Maturity Date), all as determined by Lender from time to time in its reasonable discretion in accordance with the provisions of this <u>Article 11</u>. The determination by Lender at any time that the Loan is In Balance shall not preclude Lender from determining at a later time that the Loan is not In Balance. Lender shall not be obligated to make any disbursements unless the Loan is In Balance.

(b) Borrower agrees that if Lender determines in its reasonable discretion that the Loan is not In Balance, regardless of how such condition may have been caused, Borrower shall within ten (10) days after written request by Lender (and in any event prior to any further disbursement of the Loan) deposit the deficiency with Lender ("Deficiency Deposit"). The Deficiency Deposit shall be first exhausted for costs of the Project before any further disbursements of the Loan shall be made. Disbursement of a Deficiency Deposit shall be subject to the same conditions precedent and the same requirements as apply to a disbursement of the Loan pursuant to this Agreement. Any Deficiency Deposit deposited with Lender shall be added to and made a part of the Equity Investment of Borrower in the Project, and no interest shall be paid to Borrower with respect to any such amounts. Borrower pledges to Lender any Deficiency Deposit (and Borrower's rights in any such account and interest earned thereon) as security for the Loan.

(c) (i) The "<u>Available Sources of Funds</u>" shall mean the Unfunded Commitment, adjusted as provided below, plus any remaining unexpended Deficiency Deposit, less the remaining balance of the Contingency Fund. Prior to Full Loan Opening, the unfunded Initial Equity Investment shall be an Available Source of Funds. Borrower may seek to increase the Available Sources of Funds (or the amount of any Budget Line Item) by requesting that Lender reallocate cost savings in another Budget Line Item or a portion of the Contingency Fund to pay for cost increases in Budget Line Items, and Lender shall grant or deny such request in accordance with the provisions of <u>Sections 10.2(b)</u> or <u>10.3</u>. Lender may, so long as it is entitled to do so under <u>Sections 10.2(b)</u> or <u>10.3</u>, deny a request for reallocation of cost savings or a portion of the Contingency Fund, even if as a result thereof Borrower would be required to make a Deficiency Deposit to maintain the Loan In Balance.

(ii) The following specific adjustments shall be made to the Available Sources of Funds for this transaction: (i) Upgrade Deposits may be counted as an Available Source of Funds but only up to the amount of costs of upgrades included in Lender's Estimate of Remaining Costs, and (ii) tax escrow deposits may be counted as an Available Source of Funds to the extent real estate taxes are included in Lender's Estimate of Remaining Costs.

(d) "Lender's Estimate of Remaining Costs" shall mean the amount which Lender estimates in its reasonable discretion is necessary to pay for all Hard Costs and Soft Costs of the Project which have not yet been paid, including, without limitation, the following: (i) all costs of the Construction in accordance with the Approved Plans and Specifications; (ii) all costs required to complete the preparation of the Residential Units and the Hotel in accordance with the Approved Finish Standards, including, without limitation, all finishes of Residential Units required to be paid for by Borrower under Sales Agreements or reasonably anticipated for unsold Residential Units; (iii) marketing and sales costs for the Residential Units (other than closing costs which are paid from gross sales proceeds or by Borrower pursuant to <u>Section 10.1</u>); (iv) marketing, sales, and operation costs for the Hotel, Commercial Space and Venue; (v) costs of furniture, fixtures and equipment for the Hotel, Commercial Space and Venue; (vi) any work to be completed by Borrower in accordance with the Budget; (vii) the amount required to pay interest on the Loan through the Maturity Date; (viii) all expenses payable or reimbursable to Lender under the terms of this Agreement; (ix) all real estate taxes, insurance premiums and operating costs of the Project (in excess of any Net Operating Income which Lender estimates will be available for payment of such costs); (x) all amounts needed for tenant allowances, tenant improvements and leasing commissions with respect to executed Leases and with respect to unleased space at the Project; (xi) all other amounts of any type or nature incurred or expected to be incurred in connection with the acquisition, Construction, marketing and sale of the Project or in order for Borrower to comply with the Loan Documents or the requirements of Governmental Authorities.

(e) In determining Lender's Estimate of Remaining Costs (or the amount of any Budget Line Item), Lender shall be entitled to take into account all conditions, facts and circumstances related to the Project or the Loan then existing, and all other considerations which Lender, in its reasonable discretion, determines are relevant to, or reasonably likely to have an impact upon, any of the amounts included in Lender's Estimate of Remaining Costs. By way of example and not limitation, Lender shall have the right, in making Lender's Estimate of Remaining Costs (or determining the sufficiency of any Budget Line Item), to consider in such manner and to such extent as Lender determines is appropriate in its reasonable discretion: (i) all existing and proposed modifications to the Approved Plans and Specifications (whether or not Lender has the right to approve the same), whether the same are proposed by Borrower or by a contractor, (ii) all existing construction contracts and purchase orders or, in those instances where construction contracts or purchase orders have not yet been let, written bids from responsible contractors, tradesmen and material suppliers acceptable to Lender in its reasonable discretion, or Lender's estimate of such costs, (iii) all Change Orders and pending Change Orders, (iv) all claims by any contractors or suppliers for increased or additional amounts, including all claims by the General Contractor), (vi) whether any savings in a Budget Line Item have been demonstrated to Lender's estimated to Lender's estimate deformed and permanent and whether any proposed reallocation of such savings to pay other costs satisfies the limitations and restrictions in <u>Section 10.2(b)</u> of this Agreement (and any applicable restrictions or limitations in the General Contract), and (vii) the effect of actual or anticipated delays, whether or not permitted by the terms of this Agreement.

### 11.2 Additional Equity Investment.

The Budget has been prepared and approved with the expectation that Eight Million Dollars (\$8,000,000) in Project costs set forth in such Budget will be funded by Reinvested Proceeds available pursuant to <u>Section 14.10</u>. Such anticipated Eight Million Dollars (\$8,000,000) of Reinvested Proceeds shall be an Available Source of Funds for determining whether the Loan is In Balance, provided that if less than Eight Million Dollars (\$8,000,000) of Reinvested Proceeds has been generated and is on deposit with Lender upon the earliest to occur of the following: (A) the Initial Maturity Date, (B) the occurrence of an Event of Default, and (C) the date upon which the Unfunded Commitment (exclusive of the Interest Reserve Budget Line Item) is insufficient to fund a draw request for Project Costs then due, Borrower, in addition to the Initial Equity Investment, shall deposit with Lender additional equity in an amount equal to (i) Eight Million Dollars (\$8,000,000), less (ii) the aggregate amount of Reinvested Proceeds actually received by Lender to date (the "Additional Equity Investment," and collectively with the Initial Equity Investment and any additional equity investment that may be required by Lender the "Equity Investment"). Borrower shall deposit any required Additional Equity Investment with Lender within ten (10) days of written request by Lender (and, in any event, any further disbursement of the Loan). Provided Borrower contributes such Additional Equity Investment as and when required pursuant hereto and provided further that no Default or Event of Default then exists, on or following such time as the full amount of the Reinvested Proceeds are contributed pursuant to <u>Section 14.10(b)</u> hereof, Borrower may, as part of its draw package, request that Lender reimburse Borrower in the amount of the Additional Equity Investment.

# Article 12

## CONSTRUCTION PAYOUT REQUIREMENTS

### 12.1 Applicability of Sections.

The provisions contained in this Article 12 shall apply to the Full Opening of the Loan and to all disbursements of proceeds during Construction.

## 12.2 Monthly Payouts.

After the Full Opening of the Loan, further disbursements shall be made during Construction from time to time as the Construction progresses, but no more frequently than once in each calendar month, except to pay interest from the Interest Reserve Budget Line Item in accordance with Section 10.4 or as Lender may otherwise permit in its sole discretion. Hard Costs shall be disbursed as incurred, less the Retainage. The General Contractor's fee shall be disbursed on a pro rata basis according to the percentage of construction trade items (excluding general conditions, Contractor's fees and the Contractor's Contingency) completed. All disbursements of Loan proceeds must be approved by Lender in its reasonable discretion based on costs expended as contained in the Budget. Lender shall have no obligation to disburse funds if an Event of Default or material Default has occurred and is continuing. Lender reserves the right to at any time to make disbursements into an escrow to be subsequently disbursed to Borrower by the Title Insurer.

Lender, through Lender's Consultant and/or its own employees, may perform site inspections to confirm that progress on the Project conforms to the sworn statements, and that the Project is progressing within the Budget and the Approved Plans and Specifications. Such inspections and confirmations are solely for the benefit of Lender and may not be relied upon by Borrower. Upon such confirmation, and upon receipt of a title company's commitment to issue a date down endorsement insuring the funds about to be disbursed, the Title Insurer shall disburse money to Borrower or General Contractor pursuant to the escrow agreement, or, after any default, directly to subcontractors or other payees. Provided Lender receives complete and orderly draw requests, together with the appropriate lien waivers, it shall make every commercially reasonable effort to fund such requests within ten (10) Business Days. No disbursements for costs arising under the General Contract for a particular component of the Project shall be made later than ninety (90) days after the applicable Completion Date for such component.

## 12.3 Documents to be Furnished for Each Disbursement.

As a condition precedent to each disbursement of the Loan proceeds (including the initial disbursement at the Full Opening of the Loan), Borrower shall furnish or cause to be furnished to Lender the following documents covering each disbursement, in form and substance reasonably satisfactory to Lender:

(a) A completed draw request in the form attached as Exhibit H hereto and made a part hereof, including a Borrower's Sworn Statement executed by an Authorized Representative and a completed standard AIA Form G702 and Form G703 (or similar forms acceptable to Lender) signed by the General Contractor and certified by the Architect, together with General Contractor's sworn statements and unconditional waivers of lien, and all Design Professionals', consultants', vendors', contractors', subcontractors', material suppliers' and laborers' waivers of lien (except that partial lien waivers from subcontractors and Design Professionals may be delivered on a trailing thirty (30) day basis, except for the final disbursement to each subcontractor) substantiating each line item of the draw request and covering all work, paid with the proceeds of the prior draw requests, together with such invoices, contracts or other supporting data as Lender may reasonably require to evidence that all costs for which disbursement is sought have been incurred, to be followed by lien waivers for such draw no later than the subsequent draw request, so long as the Title Insurer provides Lender with a date down endorsement to the Title Policy as provided in <u>Section 12.3(f)</u> below;

(b) A sworn owner's statement detailing the names of all suppliers, vendors, consultants and contractors with whom Borrower has contracted, amounts of contracts, amounts paid to date, and amounts of current payment and balances due, broken down in a consistent manner with the Budget;

(c) A Contractor's statement detailing the names and addresses of all suppliers, contractors and its own forces contracted to perform work including amounts of contracts, amounts paid to date, change orders, amounts of current payment, Retainage, and balances due;

(d) The Contractor's application and certification for payment executed by Borrower's lead Design Professional;

(e) Paid invoices or other evidence reasonably satisfactory to Lender that fixtures and equipment, if any, have been paid for and are free of any lien or security interest therein;

(f) A date down endorsement to the Title Policy issued to Lender covering the date of disbursement and showing the Deed of Trust as a first, prior and paramount lien on the Project subject only to the Permitted Exceptions and real estate taxes that have accrued but are not yet due and payable and particularly that nothing has intervened to affect the validity or priority of the Deed of Trust. If the Title Policy is subject to a pending disbursement endorsement, the amount of such endorsement must be increased to the full amount funded;

(g) A report from Lender's Consultant (who shall be retained solely at Borrower's expense) that contains an analysis satisfactory to Lender demonstrating the adequacy of the Budget to complete the Project, a confirmation that Construction is proceeding in accordance with the Construction Schedule and the Approved Plans and Specifications, and a certification as to amounts of Construction costs for the applicable requested funding, provided, however, that the opinion of Lender's Consultant shall not be binding on Lender;

(h) Copies of any proposed or executed Change Orders on the standard AIA G701 form that have not been previously furnished to Lender;

- (i) Copies of all construction contracts (including subcontracts) that have been executed since the last disbursement;
- (j) Verification of expenditures for Soft Costs;
- (k) All Required Permits not previously delivered to Lender;
- (1) Upon commencement of sales, the Sales Report; and
- (m) Such other instruments, documents and information as Lender or the Title Insurer may reasonably request.

#### 12.4 Retainages.

At the time of each disbursement of Loan proceeds, ten percent (10%) of the draw request for Hard Cost Line Items in the Budget (excluding the General Contractor's fee line item and the General Contractor's general conditions) shall be withheld from the amount disbursed to the various contractors, subcontractors and material suppliers for costs of the Construction (the "<u>Retainage</u>"). Once Lender determines, in its sole discretion, that any given Subcontract is 50% complete and all such work thereunder has been approved by Lender and Lender's Consultant, no further Retainage for such Subcontract will be required. Once Lender determines, in its sole discretion, that any given Subcontract is fully complete and all work thereunder has been approved by Lender and Lender's Consultant, any Retainage payable under such Subcontract shall be released. The remaining Retainage shall be disbursed when Borrower has satisfied the requirements set forth in <u>Article 13</u> below. Retainage may be witheld from tenant improvement work to be performed by Borrower on the same basis. Notwithstanding the foregoing, Lender shall not withhold a Retainage for purchases of those materials set forth on <u>Exhibit P</u> attached hereto.

### 12.5 Disbursements for Materials Stored On-Site.

Any requests for disbursements that in whole or in part relate to materials, equipment or furnishings that Borrower owns and that are not incorporated into the Improvements as of the date of the request for disbursement, but are to be temporarily stored at the Project, shall be made in an aggregate amount not to exceed \$3,000,000 with respect to materials covered by the General Contract plus \$1,000,000 for materials under owner direct contracts other than the General Contract (such amounts are subject to increase at Lender's reasonable discretion upon request of Borrower). Any such request must be accompanied by evidence satisfactory to Lender that (a) such stored materials are included within the coverages of insurance policies carried by Borrower, (b) the ownership of such materials is vested in Borrower free of any liens and claims of third parties, and Lender has a purported security interest in such stored materials; (c) such materials are properly protected against theft or damage, (d) Lender's Consultant has viewed and inspected the stored materials, and (e) in the opinion of Lender's Consultant the stored materials are physically secured and can be incorporated into the Project within forty-five (45) days.

## 12.6 Disbursements for Off-site Materials.

Lender shall make disbursements for materials stored off-site, in which event all of the requirements of Section 12.5 shall be applicable to such disbursement, provided that the following additional requirements must have been complied with: (a) such stored materials are stored in a bonded public warehouse or another facility acceptable to Lender in its sole discretion; (b) Borrower has caused any warehouseman (as defined in Section 7-102 of the Uniform Commercial Code) that possesses, holds or controls the stored materials to execute (i) a bailment letter in the form of Exhibit K and (ii) a non-negotiable warehouse receipt covering such stored materials in form sufficient to enable Lender to have perfected security interest therein and (c) Borrower has caused any party other than a warehouseman that possesses, holds or controls the stored materials to execute and deliver to Lender a bailment letter in the form of Exhibit L. At any one time, the maximum amount of disbursements for materials stored offsite shall not exceed \$3,000,000 with respect to materials covered by the General Contract plus \$3,000,000 for materials under owner direct contracts other than the General Contract (such amounts are subject to increase at Lender's reasonable discretion upon request of Borrower).

## 12.7 Specific Limitation on Disbursements.

No amounts in the Budget shall be disbursed to Borrower, Guarantor, any direct or indirect beneficial owner of an interest in Borrower, Guarantor or any Affiliate of any of the foregoing, except for (a) the Developer Fee Budget Line Item equal to \$6,000,000 (the "<u>Permitted Affiliate Expenses</u>"), which shall be paid, provided no Event of Default exists and the Loan is In Balance as follows: (i) \$4,500,000, to be paid in equal monthly installments of \$112,500 per month over Loan Term (or until such earlier time as such amount has been fully disbursed), and (ii) the remaining balance of \$1,500,000 to be paid after the repayment in full of the Loan; and (b) a payment to Guarantor of up to \$500,000 for a employee salary reimbursement for OCIP administration on behalf of Borrower. Borrower shall have no right to borrow any portion of the developer's fee referred to in clause (a)(ii) above from Lender.

All disbursements of Permitted Affiliate Expenses are subject to the same conditions precedent as apply to disbursements generally under this Agreement.

### 12.8 Disbursements Related to Commercial Space Leases.

Loan proceeds may be used to fund leasing commissions and tenant improvements with respect to Approved Leases, provided (a) the maximum cost to be funded with respect to each Commercial Space Lease shall not exceed (i) for tenant improvements, the dollar amount per Rentable Square Foot set forth on Exhibit Q attached hereto with respect to the type of space being leased (the "<u>Allowable Tenant Improvements</u>"), and (ii) for leasing commissions, \$21.00 per Rentable Square Foot, and (b) the undisbursed amount in the (i) Tenant Improvements Budget Line Item must be sufficient to pay for the Allowable Tenant Improvements for all Commercial Space for which tenant improvements have not been paid, and (ii) the undisbursed amount in the Commercial Leasing Commissions Budget Line Item must equal at least \$21 times the number of unleased Rentable Square Feet in the Commercial Space. Borrower shall fund any excess as the first dollars to be funded for any Commercial Space, provided that cost savings in (i) tenant improvements for any one Commercial Space Lease may be reallocated to tenant improvements with respect to any other Commercial Space Lease. In addition to any items to be provided by Borrower pursuant to <u>Section 12.3</u> hereof, and as a condition precedent to the first advance for any tenant improvements, copies of all proposed contracts with trades or subcontractors relating to the completion of the tenant improvements, and schedule for the completion of the tenant improvements and information which will demonstrate that the tenant improvements will be completed in accordance with the Lease and the Budget. Borrower shall also promptly provide to Lender proof of payment of all leasing commissions.

## 12.9 Delivery of Subcontracts. »

Borrower shall deliver to Lender by the deadlines specified in <u>Exhibit N</u> those certain fully-executed Subcontracts for the trades and divisions set forth on <u>Exhibit N</u>, which Subcontracts shall be in form and substance satisfactory to Lender in its reasonable discretion. The receipt of Subcontracts by the applicable deadline as set forth in <u>Exhibit N</u> shall be a condition to funding any then-pending or forthcoming draw request.

## FINAL DISBURSEMENT FOR CONSTRUCTION

### 13.1 Final Disbursement for Construction.

Lender will advance to Borrower the final disbursement for the cost of the Construction (including retainages not yet disbursed) when the following conditions have been complied with, provided that all other conditions in this Agreement for disbursements have been complied with:

(a) The Improvements have been fully completed and equipped substantially in accordance with the Approved Plans and Specifications free and clear of materialmen's liens and security interests and are ready for occupancy;

(b) Borrower shall have furnished or caused to be furnished to Lender "all risks" casualty insurance and windstorm and flood insurance each in form and amount and with companies reasonably satisfactory to Lender in accordance with the requirements contained herein;

(c) Borrower shall have furnished to Lender copies of all licenses and permits required by any Governmental Authority having jurisdiction for the occupancy of the Improvements and the operation thereof, including a final certificate of occupancy from the municipality in which the Project is located or a temporary certificate of occupancy, so long as (i) Residential Units may still be sold and occupied, (ii) the Hotel is open and operating in accordance with the Hotel Documents, (iii) the Commercial Space may still be rented and/or sold and occupied, and (iv) the Venue is open and operating in accordance with the management agreement reviewed and approved by Lender in accordance herewith;

(d) Borrower shall have furnished a plat of survey covering the completed Improvements in compliance with Section 8.1(d);

(e) All fixtures, furnishings, furniture, equipment and other property required for the operation of the Project shall have been installed free and clear of all liens and security interests, except in favor of Lender;

(f) Borrower shall have furnished to Lender copies of all final waivers of lien (other than the liens that Borrower is contesting and has insured or bonded over in accordance with <u>Section 15.1(f)</u> of this Agreement), and sworn statements from contractors, subcontractors and material suppliers and an affidavit from the General Contractor in accordance with the mechanic's lien law of the State or as otherwise established by Lender;

(g) Borrower shall have furnished to Lender a certificate from the Architect or other evidence satisfactory to Lender in Lender's sole discretion dated at or about the Completion Date stating that (i) the Improvements have been completed substantially in accordance with the Approved Plans and Specifications, and (ii) the Improvements as so completed comply with all applicable Laws in all material respects; and

(h) Lender shall have received a certificate from Lender's Consultant for the sole benefit of Lender that the Improvements have been satisfactorily completed substantially in accordance with the Approved Plans and Specifications.

Borrower shall comply with and satisfy each of the foregoing final disbursement conditions contained in this <u>Section 13.1</u> within ninety (90) days after the Completion Date.

### Article 14

## SALE OF RESIDENTIAL UNITS OR OTHER PORTIONS OF THE PROJECT

## 14.1 Price List Schedule.

Borrower has submitted to Lender for its written approval a Price List Schedule (the "<u>Price List Schedule</u>") stating, at a minimum, a detailed breakdown of each condominium Residential Unit's size, type, location and gross sales price ("<u>List Price</u>") for each Residential Unit (including, but not limited to, a weighted average gross sales price (excluding Parking Spaces and Upgrades) of not less than \$620 per Saleable Square Foot for each Residential Unit). The approved Price List Schedule is attached to this Agreement as <u>Exhibit M</u>. No changes to the approved Price List Schedule shall be permitted without Lender's prior written approval, in Lender's sole and reasonable discretion. The List Prices for each Residential Unit include the price of one (1) Parking Space per bedroom in such Residential Unit, up to two (2) Parking Spaces per Residential Unit, except for the penthouse Residential Units, which are each allocated three (3) Parking Spaces but does not include storage space. Of the 306 Parking Spaces allocated to the Residential Units collectively, 300 Parking Spaces shall be included in the List Prices of such Residential Units as set forth above, and 6 Parking Spaces (the "<u>Excess Parking Spaces</u>") shall not be included in the List Price of any Residential Unit and shall be available for sale at an additional charge.

#### 14.2 Sales Agreements.

Each Residential Unit shall be sold under a written agreement (the "<u>Sales Agreement</u>") in the form of the sales agreements attached hereto as <u>Exhibit</u> <u>O</u>, or on a form otherwise approved by Lender, in its sole discretion in all material respects conforming to all Laws, including those requiring disclosures to prospective and actual buyers. Lender has previously approved the form of the Sales Agreement to be used for such Residential Unit sales, which form of sales contract is attached hereto as <u>Exhibit O</u>. No Residential Unit may be leased, sold or conveyed under any lease, conditional sales contract or other arrangement where Borrower retains a deferred portion of the purchase price or any residual or contingent interest in the Residential Unit, including any purchase money security interest, without the express prior written consent of Lender in each instance. All agreements (each, also a "<u>Sales Agreement</u>") with respect to the sale of any portion of the Project other than the Residential Units shall be subject to the prior written approval of Lender in its reasonable discretion. Borrower shall deliver to Lender a copy of each executed Sales Agreement within thirty-one (31) days of its execution. Borrower shall promptly enforce the obligations of the purchasers under each Sales Agreement upon default by any such purchaser, except as lender may otherwise agree in writing in its reasonable discretion.

## 14.3 Purchaser Deposits.

(a) On or before Closing (as to earnest money deposits received prior to Closing), and promptly after receiving any additional deposits made by the Residential Unit Purchasers from time to time (as to earnest money deposits received following Closing), Borrower shall deposit with Armbrust & Brown, L.L.P. and Heritage Title Company of Austin, Inc., jointly and severally, as escrow agent (or any successor agent approved by Lender in writing in its sole discretion) (together, jointly and severally, the "Escrow Agent"), in accordance with the Escrow Agreement, a copy of which has been delivered to Lender, and shall cause Escrow Agent to maintain in one or more accounts titled in the name and subject to the control of Escrow Agent all earnest money deposits paid by Residential Unit Purchasers under the Sales Agreements (the "Earnest Money Deposits"). At Lender's option, Borrower shall require Escrow Agent to use Lender as the depository bank for all deposits, and in such case, the deposits shall be placed in an interest bearing account. Such account(s) shall bear interest at a rate comparable to the rate provided by Escrow Agent.

(b) Borrower shall require all Residential Unit Purchasers to make cash Earnest Money Deposits of at least ten percent (10%) of the gross sales price at the time of contract execution. If Earnest Money Deposits are held by an Escrow Agent, Borrower shall cause Escrow Agent to deliver to Lender within seven (7) days of the end of each calendar month a statement indicating the amount of funds on deposit representing Earnest Money Deposits, together with information on the date of deposit, and to which Residential Unit all such deposits apply. Borrower shall not accept any non-cash Earnest Money Deposits or Upgrade Deposits. Borrower shall not be permitted to use or apply the Earnest Money Deposits prior to the closing of the sale of a Residential Unit, whereupon the Earnest Money Deposit shall be considered part of the Net Sales Proceeds and applied as set forth in <u>Section 14.9</u>.

(c) If a Residential Unit Purchaser desires to alter the Approved Construction Finish Standards, which results in additional costs to Borrower to complete a Residential Unit (an "<u>Upgrade</u>"), Borrower shall require such Residential Unit Purchaser to deposit funds with Lender prior to the commencement of any work relating to such Upgrade (the "<u>Upgrade Deposit</u>"). Borrower shall require Residential Unit Purchasers to make Upgrade Deposits equal to one hundred percent (100%) of the cost to Borrower of such Upgrade. All Upgrade or Residential Unit customization costs shall be subject to Lender's approval if such costs for any Residential Unit are in excess of Forty Dollars (\$40.00) per Saleable Square Foot. Loan proceeds shall not be available for payment of costs incurred in connection with Upgrades or customization items, the sole responsibility for such costs being with Borrower. So long as permitted under applicable law, Lender shall permit Borrower to use Upgrade Deposits to perform such Upgrades. To the extent that the amount of Upgrade Deposit on deposit with Lender is not sufficient to pay in full the costs of any use Upgrades when due, Borrower shall promptly deposit with Lender an additional Equity Investment equal to the costs of the Upgrade. Such deposit will be required at the time of the execution of the Sales Agreement containing the Upgrade. Lender shall also be entitled to inspect any Upgrade work, and shall be entitled to review and approve reasonable supporting documentation with respect to such Upgrade work, in a manner commensurate with other Hard Cost disbursements before approving payment therefor.

(d) The Deposits shall be segregated from other funds and shall be held, applied or returned, as applicable, in accordance with the terms of the respective Sales Agreement and applicable Laws. To the extent Borrower becomes entitled to retain any Deposits that have not been previously applied in accordance herewith (i.e. upon the forfeiture of any deposit by a Residential Unit Purchaser), such amounts shall be paid to Lender and applicable to the Loan. To the extent that a Residential Unit Purchaser becomes entitled to return of its Deposits under its Sales Agreement or under applicable Laws, so long as the Upgrade Deposit has not already been spent by Borrower, Borrower shall be entitled to notify either Lender or the Escrow Agent, as applicable, and withdraw from the escrow account such Deposit for return to the Residential Unit Purchaser, with a copy of such notice to be simultaneously sent to Lender if Deposits are held by an Escrow Agent. The Escrow Agent may not be changed without the prior written consent of Lender. Notwithstanding the foregoing, Lender hereby acknowledges and agrees that Borrower's rights to the Deposits are subject to the rights of the Residential Unit Purchasers to such Deposits as set forth in the Sales Agreements, the Escrow Agreement and under applicable Law, and that Lender or the Escrow Agent, as the case may be, may be obligated to return the Deposits to such Residential Unit Purchasers when and as so required even if an Event of Default exists.

(e) Borrower hereby grants to Lender a security interest in all of Borrower's right, title and interest in and to the Deposits and all accounts holding any such deposits from time to time. Upon request by Lender, Borrower shall promptly provide to Lender such documentation as Lender determines is necessary to confirm and perfect such security interest. Such security interest is subject to the rights of Residential Unit Purchasers in and to such Deposits in accordance with the terms of their respective Sales Agreements and applicable Laws.

## 14.4 Residential Unit Sales.

Borrower may enter into Sales Agreements and sell Residential Units without Lender's prior written consent only if:

(a) A Sales Agreement is executed with the Residential Unit Purchaser, the form and substance of which does not materially differ from the form of sales contract attached hereto as <u>Exhibit O</u> and approved by Lender in writing and that otherwise conforms to the requirements of this Agreement, and Borrower has not entered into any side agreements relating to the purchase of the Residential Units with any Residential Unit Purchaser.

(b) The Sales Agreement requires full payment in cash not later than closing, with no purchase money financing provided by Borrower, and the sale of the Residential Unit is for a base price (including one (1) Parking Space per bedroom in such Residential Unit, up to two (2) Parking Spaces per Residential Unit, except for the penthouse Residential Units, which are allocated three (3) Parking Spaces per Residential Unit) that is greater than or equal to 90% of such Residential Unit's List Price set forth in the Price List Schedule, excluding from such base sale price any Upgrades and any seller concessions or discounts;

(c) The Sales Agreement requires non-refundable earnest money of at least ten percent (10%) of the gross sales price to be deposited with Lender or the Escrow Agent in accordance with Section 14.3 hereof;

(d) The Residential Unit Purchaser is not affiliated with Borrower or Guarantor, and no more than two Residential Units may be purchased by any single Residential Unit Purchaser or its Affiliates without Lender's prior approval, except that up to nine (9) Residential Units may be sold to Affiliates of Borrower or Guarantor;

(e) The Sales Agreement must not be cancelable or contain any unexpired contingencies except as set forth in Exhibit O;

(f) The Sales Agreement must not include a Residential Unit completion, delivery, or closing date which gives the Residential Unit Purchaser the ability to cancel or terminate the Sales Agreement. Notwithstanding the foregoing, the Sales Agreement may include an outside estimated Project completion date that is no earlier than one (1) year following the completion date for the floor upon which such Residential Unit is to be located, as stated in the construction schedule attached to the General Contract, plus extensions for Unavoidable Delay;

(g) The Sales Agreement clearly identifies any Upgrade required to be provided by Borrower, as seller, which Upgrade and Upgrade Deposit shall comply with the requirements hereof and all Laws;

(h) The Sales Agreement is not assignable by the Residential Unit Purchaser, except to an Affiliate of the Residential Unit Purchaser;

(i) [Intentionally Omitted];

(j) Subject to subsection (k) below, the Sales Agreement stipulates that, in the event that the Residential Unit Purchaser fails to perform thereunder, Borrower shall be entitled to the full amount of the Earnest Money Deposit and Upgrade Deposit and can sue the Residential Unit Purchaser to collect any unpaid portion of such Earnest Money Deposit; and

(k) The Sales Agreement complies with ILSA through a HUD filing, not through an exemption or exception, and complies with all other local and state regulations applying to condominium sales. Lender shall have received an executed acknowledgement by each Residential Unit Purchaser stipulating that prior to the date of the Sales Agreement acceptance, Residential Unit Purchaser received a copy of the Property Report which has been accepted for filing by HUD. The Sales Agreement must not be dated earlier than the effective date on which HUD accepted the Statement of Record. In addition, Lender shall have approved the specific terms of the form contract, which shall contain provisions required by ILSA including:

(i) A statement that the Residential Unit Purchaser may revoke the Sales Agreement for a period of seven (7) days from the date of signing, or if applicable state law provides for a longer period, the longer period shall be stated.

(ii) A statement that if the Residential Unit Purchaser has not received the ILSA Property Report prior to executing the Sales Agreement, the Sales Agreement shall be revocable for two (2) years from the date the Sales Agreement was signed by Residential Unit Purchaser.

(iii) A provision which states that if the Residential Unit Purchaser defaults under the terms of the Sales Agreement, the Residential Unit Purchaser shall receive written notice of default, and shall have a period of twenty (20) days from the date of the notice to cure the default.

(iv) Damages payable to Borrower under the Sales Agreement must conform to the limitations in ILSA, which generally limit damages to the greatest of 15% liquidated damages or actual damages.

"Qualifying Sales Agreement" shall mean (i) an agreement meeting all of the conditions required by this Agreement, and (ii) those certain Sales Agreements delivered to Lender prior to the date of this Agreement, despite the failure of such Sales Agreements to comply with the requirements of Sections 14.4(d), (e) (f) and (h) above. Borrower shall not, without the prior written consent of Lender, enter into any Sales Agreement that is not a Qualifying Sales Agreement, or any other documentation pertaining to the sale of any portion of the Project; provided, however, that Borrower may enter into a Qualifying Sales Agreement subject to a commercially reasonable financing contingency for up to sixty (60) days provided such Qualifying Sales Agreements will not be deemed Qualifying Sales Agreements until the expiration or satisfaction of the Deed of Trust contingency. Borrower shall not terminate a Qualifying Sales Agreement for any reason other than the default of the Residential Unit Purchaser thereunder without Lender's prior written consent.

# 14.5 Sales Operations and Seller's Obligations.

Borrower shall at all times maintain adequate marketing capability, and shall timely perform all obligations required to be performed by it under each Sales Agreement.

## 14.6 Delivery of Sales Information and Documents.

Within ten (10) days after the end of each month, Borrower shall deliver to Lender a sales report (the "<u>Sales Report</u>") showing (a) each Residential Unit under contract for sale, including the date of the contract, the Residential Unit number, the base contract price plus the charge for Upgrades, the amount of the Earnest Money Deposit and, if applicable, the Upgrade Deposit, and the date expected to close; and (b) for each Residential Unit closed, the Residential Unit number, the base contract price plus the charge for Upgrades, the date closed, and the Net Sales Proceeds paid to Lender; and (c) for each Sales Agreement cancelled by a Residential Unit Purchaser, the Residential Unit number, the date cancelled, the reason the contract was cancelled by the Residential Unit Purchaser and a statement regarding whether the Earnest Money Deposit was returned to the Residential Unit Purchaser or retained by Borrower (and remitted to Lender). Borrower shall also promptly deliver to Lender such other sales information and documents that Lender from time to time may reasonably request, including operating statements, all new Sales Agreements, and notice of or information regarding any claimed breach or disavowal of buyer's or seller's obligations under any one or more Sales Agreements.

### 14.7 Borrower's Acknowledgment Regarding Buyer Financing.

Borrower acknowledges that Lender is not committed to provide any financing to or for the buyers of any individual Residential Units.

#### 14.8 Condominium Regime.

Borrower shall promptly, and in any event (a) prior to the closing of the first purchase and sale of any Residential Unit, cause the Condominium Documents to be filed or recorded, as appropriate, and (b) prior to Lender's execution of any partial release of the Deed of Trust, cause the Master Condominium Declaration to be recorded. Prior to recording or filing any documents, Borrower shall submit to Lender for Lender's final review and reasonable approval, executed copies of the proposed Condominium Documents, which, once approved, shall be filed or recorded, as applicable. If such documents have been submitted to and approved by Lender prior to the Full Loan Opening Date, any changes or additions to such documents from those submitted to Lender prior to the Full Loan Opening Date shall be subject to Lender's consent, which consent shall not unreasonably be withheld. Borrower shall on an ongoing basis comply with all requirements of ILSA and all other applicable Laws relating to the sale or marketing of the Residential Units. Without limiting the generality of the foregoing, Borrower shall (i) timely file its Annual Report of Activity with HUD each year during the term hereof; (ii) pay any fees in connection therewith before the due date; (iii) deliver financial statements to HUD within one hundred and twenty (120) days after the close of Borrower's fiscal year; and (iv) amend the HUD Property Report to reflect any change in any material representation of a material fact required in the Statement of Record within fifteen (15) days of learning of the change. Borrower shall deliver to Lender (simultaneously with delivery to the applicable governmental authority) copies of all filings and notices delivered to HUD in accordance with ILSA, and Borrower shall, within five (5) Business Days after receipt thereof, deliver to Lender all notices, requests, correspondence and demands delivered by HUD, the State of Texas or the City of Austin to Borrower. Borrower shall not amend the HUD Property Report or the City or State Property Report without Lender's prior written consent. Provided that no Event of Default exists, Lender, prior to the initial conveyance of Residential Units, shall execute such subordinations of the Deed of Trust and other documents as are needed to permit the recordation or filing of such documents. Prior to such execution by Lender and as a condition thereto, Borrower shall furnish to Lender a copy of (a) a notice sent by Borrower to the condominium association informing the association of Lender's rights under the declaration as a beneficiary under the Deed of Trust and furnishing the association with Lender's address for notice purposes, (b) resignations of all Borrower-appointed directors and officers, which Lender may deliver at any time an Event of Default exists, and (c) such other documents in connection with the establishment of such condominium regime as Lender reasonably requests.

Borrower shall not permit control of any condominium association to be turned over to the Residential Unit owners more than thirty (30) days prior to the date that is required by Texas law.

### 14.9 Release of Residential Units.

(a) At Borrower's request upon the closing of a Residential Unit sale, Lender shall issue a partial release of the Residential Unit from the Deed of Trust, so long as all of the following conditions are satisfied at the time of, and with respect to, the partial release:

(i) The Master Declaration of Condominium and the Residential Condominium Declaration have each been recorded in the public records of Travis County, Texas;

(ii) No Event of Default exists;

(iii) Such sale is pursuant to a Qualifying Sales Agreement;

(iv) Lender has been paid, in immediately available funds, a Release Price (defined below) for the Residential Unit in an amount determined as set forth below to be applied to the Loan as set forth in <u>Section 14.10</u>;

(v) Lender receives a copy of the closing statement and applicable Residential Unit release no later than 9:00 a.m. (Chicago time) at least one (1) Business Day prior to any Residential Unit sale; and

(vi) All escrow, closing and recording costs have been paid at no expense to Lender.

(b) The execution of a Sales Agreement shall not by itself satisfy the conditions for release of the Residential Unit that is being sold; those conditions must be satisfied in full at the time the Residential Unit is to be released.

(c) If Lender does not require satisfaction of all of the conditions described above before releasing one or more Residential Units, that alone shall not be a waiver of such conditions, and Lender reserves the right to require their satisfaction in full before releasing any further Residential Units from the Deed of Trust.

(d) The "<u>Release Price</u>" for a Residential Unit (including one (1) Parking Space per bedroom in such Residential Unit, up to two (2) Parking Spaces per Residential Unit, except for the penthouse Residential Units, which are each allocated three (3) Parking Spaces ) shall be the greatest of (x) 100% of Net Sales Proceeds from the sale of the applicable Residential Unit and associated Parking Spaces (including any additional charge or fee in excess of the purchase price payable to Borrower or its Affiliates), (y) 90% of the gross sales proceeds, exclusive of any extras and/or Upgrades, plus 100% of any Upgrade Profits for such Residential Unit, and (z) 90% of the Sales Agreement contract sales price (excluding the selling price of any extras and/or Upgrades, and without accounting for sales commissions paid from the Project Budget), plus 100% of any Upgrade Profits for such Residential Unit.

(e) Borrower shall pay all reasonable costs and expenses associated with the sale of the Residential Unit, including title expenses, reasonable legal fees, brokerage and sales commissions and other closing costs, from the portion of the sales price in excess of such

Release Price and, if such excess is insufficient for such purpose, shall pay such excess costs from its own funds.

(f) If requested by Lender, Borrower shall also deposit with Lender (to the extent not required to be deposited in condominium association bank accounts) all amounts deducted or set aside for real estate taxes or retained for assessments or working capital (and pledges its interest in such account(s) to Lender). Lender shall hold such amount subject to the rights of Residential Unit Purchasers and the condominium association therein, provided, however, Lender shall permit Borrower to use the funds deposited in such accounts for their legally required purposes.

## 14.9A Sales of Parking, Storage, Commercial Space or Venue.

(a) The Hotel may be released from the Deed of Trust during the term of the Loan in accordance herewith. The Venue and the Commercial Space may be subject to release from the Deed of Trust, provided that, in addition to the satisfaction of the other conditions of this Section 14.9A, the Venue and the Commercial Space must be released concurrently. Borrower shall provide for Lender's prior review and written approval any agreement for the sale of any Excess Parking Space, Storage, Hotel, and/or Venue and Commercial Space, and Borrower shall obtain Lender's consent thereto prior to executing such sale agreement. Lender's consent and approval under this paragraph shall not be unreasonably withheld or delayed.

(b) At Borrower's request upon the closing of a sale of the Hotel, the Commercial Space or Venue, Lender shall issue a partial release of the portion of the Project from the Deed of Trust, so long as all of the following conditions are satisfied at the time of, and with respect to, the partial release:

(i) The Declaration of Condominium has been recorded in the public records of Travis County, Texas;

(ii) All rights of the declarant under the Declaration of Condominium shall remain with Borrower following the sale and release;

(iii) No Event of Default exists;

(iv) With respect to the Venue and Commercial Space only, Borrower shall have closed and conveyed, pursuant to Qualifying Sales Agreements, and Lender shall have received, Release Prices for such number of Residential Units totaling or exceeding 137,500 Saleable Square Feet of space; and

(v) Such other conditions as Lender shall reasonably require.

(c) The "Release Price" for:

(i) Any Excess Parking Space shall be the greater of (x) 100% of the Net Sales Proceeds for such Parking Space(s), and (y) \$20,000 per Parking Space;

(ii) The storage space shall be the greater of (x) 100% of the Net Sales Proceeds for such storage space, and (y) \$20 per square foot of such storage space;

(iii) The Venue and the Commercial Space, together, shall be the greater of: (x) 100% of the Net Sales Proceeds or proceeds of the thirdparty re-financing of such area, and (y) \$20,000,000; and

(iv) The Hotel shall be the greatest of: (x) 100% of the Net Sales Proceeds or proceeds of the third-party re-financing of the Hotel, (y) 70,000,000, and (z) an amount, to be applied as a payment of principal, that, after payment of the Exit Fee and the escrow of Reinvested Proceeds, would be sufficient to reduce Lender's Remaining Exposure to: (A) 100, multiplied by (B) the Saleable Square Feet in the Remaining Units.

### 14.10 Application of Sales Proceeds.

Except as otherwise expressly provided herein, all Release Prices received by Lender in connection with the sale of a Residential Unit or other component of the Project addressed in <u>Section 14.9A</u> above shall be applied by Lender as follows (regardless of any contrary order of payment specified by Borrower):

(a) First, with respect to Release Prices from Residential Units, to the payment of any Exit Fee then due, or, with respect to the Release Price from the Hotel, to the payment of the full Exit Fee;

(b) Second, the first \$8,000,000 of aggregate Release Prices collected pursuant to closings that otherwise satisfy the terms and provisions of this Agreement (collectively, the "<u>Reinvested Proceeds</u>") shall be held by Lender and shall be used by Lender (before funding from the Loan proceeds) to fund un-depleted line items in the Budget (except for the interest payable on the Loan, which shall be paid from the Interest Reserve Budget Line Item); provided, however, Lender shall not be required to disburse any Reinvested Proceeds unless: (i) the Initial Equity Investment has been contributed in accordance with Section 8.1(a) hereof, (ii) the Additional Equity Investment, if any, has been contributed, and (iii) all conditions set forth in Section 4.1(c) are satisfied. Borrower hereby pledges to Lender all Reinvested Proceeds held by Lender from time to time as additional security for the Loan.

(c) Third, to the payment of principal then due and owing under the Note, until the amounts due thereunder have been paid in full; and

(d) Finally, to any other amounts payable to Lender under the Loan Documents, including any costs and expenses of Lender.

Notwithstanding the foregoing, (i) Lender may elect in its sole discretion to apply Release Prices proceeds first to payment of accrued interest (but shall never be obligated to do so, even if doing so would cure a default in the payment of interest) and (ii) if an Event of Default exists, Lender may apply all amounts received to the indebtedness under the Loan Documents in such order as Lender may elect in its sole discretion.

# OTHER COVENANTS

### 15.1 Borrower further covenants and agrees as follows:

(a) <u>Full Opening of Loan on or Prior to Full Loan Opening Date</u>. Borrower shall cause all conditions precedent set forth in <u>Article 8</u> and <u>Section 9.1</u> of this Agreement, except those which are expressly required to be satisfied by a later date, to be satisfied on or before May 15, 2008. Borrower shall cause all conditions precedent to the Full Opening of the Loan to be complied with, and Borrower shall qualify for Full Loan Opening, no later than January 1, 2010. If Borrower fails to satisfy timely either of such requirements, an Event of Default shall exist and Lender may terminate its commitment to fund the Loan by written notice to Borrower.

(b) Compliance with Condominium Documents. Once the Declaration of Master Condominium and/or Residential Condominium Declaration are recorded, and subject to the provisions of Section 82.112(b) of the Texas Uniform Condominium Act, Borrower shall pay all general and special assessments for common charges and expenses and insurance premiums made against or relating to the Residential Units owned by Borrower or otherwise payable by Borrower under the Condominium Documents as the same shall become due and payable and prior to delinquency, and not later than the tenth (10<sup>th</sup>) day of each month provide to Lender evidence of such payments, and in the event Borrower shall fail to make such payments as the same become due and payable and prior to delinquency, Lender may from time to time at its option, but without any obligation to do so and without notice or demand upon Borrower, make such payments, and all expenses paid by Lender for such purpose, including, without limitation, attorneys' fees, shall be added to the outstanding principal amount of the Loan and shall be payable on demand and bear interest at the Default Rate until repaid. Borrower shall not (and shall not permit any Borrower appointed directors), without the prior written consent of Lender (not to be unreasonably withheld or delayed), give any consent or perform any action in furtherance of any material modification or amendment of the Condominium Documents, including any modifications or amendments to the Condominium Documents which would permit a Residential Unit Purchaser to rescind its Sales Agreement under Texas Laws. Borrower shall comply with all of the material terms, covenants and conditions on its part to be performed under the Condominium Documents, as the same shall be in force and effect from time to time; provided, however, that if Borrower fails to cure such non-compliance within any applicable cure periods provided in the Condominium Documents, Lender may from time to time at its option, but without any obligation to do so, cure or remedy any such default by Borrower (Borrower hereby authorizing Lender to enter upon the Project as may be necessary for such purposes), and all reasonable expenses paid by Lender for such purpose, including, without limitation, reasonable attorneys' fees, shall be added to the outstanding principal balance of the Loan and shall be payable on demand and bear interest at the Default Rate until repaid. Borrower shall deliver to Lender a true and complete copy of each and every notice of default, if any, received by Borrower with respect to Borrower under any of the Condominium Documents or applicable law regarding the condominium. Borrower shall not (and shall not permit any Borrower appointed director to), without the prior written consent of Lender, exercise any right it may have to vote for (x) the expenditure of insurance proceeds (which are governed by Article 16 below) or condemnation awards for the repair or restoration of the Project or (y) any additions or improvements to the common elements of the Project.

(c) <u>Construction of Improvements</u>. The Improvements shall be constructed and fully equipped in a good and workmanlike manner with materials of high quality, substantially in accordance with the Approved Plans and Specifications (or in accordance with any changes therein that may be approved in writing by Lender or as to which Lender's approval is not required). Such construction and equipping shall be commenced and completed, as applicable, and shall be prosecuted with due diligence and continuity in substantial accordance with the Construction Schedule, the requirements of all Sales Agreements and the Operating Agreement. Without limiting the foregoing, Borrower shall commence construction by the Required Commencement Date, and shall substantially complete (i) the entire Project no later than the Completion Date applicable to the entire Project, and (ii) each particular portion of the Project no later than the Completion Date applicable to extension for any reason, except that Lender may elect, but shall be under no obligation, to extend such dates at its sole discretion, but such election shall not constitute a waiver of Lender's right to enforce any other Completion Date or a waiver of any of Lender's remedies and shall not constitute any custom or course of dealing between the parties hereto.

(d) <u>Payment for Work</u>. Borrower agrees to fully pay and discharge when due and payable all claims for labor done and material and services furnished in connection with the construction of the Project and to take all other steps to forestall the assertion of claims against the Project or the Loan.

(e) Inspection by Lender. Borrower shall reasonably cooperate with Lender in arranging for inspections by representatives of Lender of the progress of the Construction from time to time including an examination of (i) the Improvements, (ii) all materials to be used in the Construction, (iii) all plans and shop drawings that are or may be kept at the construction site, (iv) any contracts, bills of sale, statements, receipts or vouchers in connection with the Improvements, (v) all work done, labor performed, and materials furnished in and about the Improvements, (vi) all books, contracts and records with respect to the Improvements, and (vii) any other documents relating to the Improvements or the Construction. Borrower shall cooperate with Lender's Consultant to enable him to perform his functions hereunder and will promptly comply with Lender's requirements and remove any defect regarding the Construction of the Improvements or the progress thereof.

(f) <u>Materialmen's Liens and Contest Thereof</u>. Borrower shall not suffer or permit any materialmen's lien claims to be filed or otherwise asserted against the Project or any funds due to the General Contractor, and shall promptly discharge the same in case of the filing of any claims for lien or proceedings for the enforcement thereof, provided, however, that Borrower shall have the right to contest in good faith by appropriate legal proceeding and with reasonable diligence the validity of any such lien or claim, provided that Borrower posts a statutory lien bond over such lien. Lender shall not be required to make any further disbursements of the proceeds of the Loan until any materialmen's lien claims have been removed or bonded over, and Lender may, at its option, restrict disbursements to reserve sufficient sums to pay 125% of the face amount of the lien unless and until the lien is removed or bonded over.

(g) Settlement of Materialmen's Lien Claims. If Borrower shall fail promptly either (i) to discharge any such lien, or (ii) to post a statutory lien bond over such lien, in each case, within thirty (30) days after the filing of the lien, Lender may, at its election (but shall not be

required to), procure the release and discharge of any such claim and any judgment or decree thereon and, further, may in its sole discretion effect any settlement or compromise of the same, or may furnish such security or indemnity to the Title Insurer, and any amounts so expended by Lender, including premiums paid or security furnished in connection with the issuance of any surety company bonds, shall be deemed to constitute disbursement of the proceeds of the Loan hereunder. In settling, compromising or discharging any claims for lien, Lender shall not be required to inquire into the validity or amount of any such claim.

(h) <u>Proceedings</u>. In addition to (and not in lieu of Borrower's covenants set forth in <u>Section 15.1(f)</u>), if any action, claim or proceeding affecting title to the Land, the construction of the Project, Borrower, Guarantor, the rights of Lender under any Loan Document, or otherwise affecting the Project, be filed or commenced, then at the request of Lender, Borrower shall appear in and defend, at Borrower's sole cost and expense, any such action or proceeding and, if applicable, Borrower shall insure or bond over such action, claim or proceeding in accordance with <u>Section 15.1(f)</u>. If Borrower fails to appear and defend any such action or proceeding, then Lender may commence, intervene in, and defend actions or proceedings affecting the Project or the transactions contemplated herein, and compromise or settle any claim or controversy pertaining thereto, employing legal counsel acceptable to Lender to defend such claims at Borrower's sole cost, unless due to the willful misconduct of Lender. Lender shall not be liable to Borrower for any action, error, mistake, omission or delay pertaining to the actions described in this Section or any damages resulting thereform, unless due to the willful misconduct of Lender. Any cost incurred by Lender under this Section shall be deemed to be expenses of the Loan payable by Borrower pursuant to <u>Article 7</u> of this Agreement.

(i) <u>Insurance</u>. Borrower shall cause insurance policies to be maintained in compliance with <u>Exhibit E</u>, or such other insurance requirements as may be reasonably required by Lender, at all times. Borrower shall provide Lender, prior to the date hereof, a certificate(s) demonstrating appropriate insurance coverage(s), which shall demonstrate insurance coverage that meets or exceeds the requirements on <u>Exhibit E</u>. Borrower shall timely pay all premiums on all insurance policies required hereunder, and as and when additional insurance is required, from time to time, during the progress of Construction, and as and when any policies of insurance may expire, furnish to Lender, premiums prepaid, additional and renewal insurance policies with companies, coverage and in amounts reasonably satisfactory to Lender in accordance with <u>Exhibit E</u>.

(j) Payment of Taxes. Borrower shall pay all real estate taxes and assessments and charges of every kind upon the Project before the same become delinquent. Borrower may use Loan proceeds for such purpose to the extent available in the Budget for such purpose upon meeting all conditions precedent set forth in this Agreement to any such disbursement. Borrower shall have the right to pay such tax under protest or to otherwise contest any such tax or assessment, but only if (i) such contest has the effect of preventing the collection of such taxes so contested and also of preventing the sale or forfeiture of the Project or any part thereof or any interest therein, (ii) Borrower has notified Lender of Borrower's intent to contest such taxes, and (iii) Borrower has deposited security in form and amount reasonably satisfactory to Lender, in its reasonable discretion, and has increased the amount of such security so deposited promptly after Lender's request therefor. If Borrower fails to commence such contest or, having commenced to contest the same, and having deposited such security required by Lender for its full amount, shall

thereafter fail to prosecute such contest in good faith or with due diligence, or, upon adverse conclusion of any such contest, shall fail to pay such tax, assessment or charge, Lender may, at its election (but shall not be required to), pay and discharge any such tax, assessment or charge, and any interest or penalty thereon, and any amounts so expended by Lender shall be deemed to constitute disbursements of the Loan proceeds hereunder (even if the total amount of disbursements would exceed the face amount of the Note). Borrower shall furnish to Lender evidence that taxes are paid at least five (5) days prior to the last date for payment of such taxes and before imposition of any penalty or accrual of interest.

(k) <u>Tax Escrow Accounts</u>. After the earlier to occur of (i) an Event of Default or (ii) the Budget Line Item for real estate taxes having been exhausted (in either case, a "<u>Triggering Event</u>"), Borrower shall make monthly tax escrow deposits in the amount of one-twelfth (1/12) of one hundred ten percent (110%) of the annual real estate taxes as reasonably estimated by Lender, such deposit to be held in an interest bearing escrow account held by Lender in Lender's name and under its sole dominion and control. (If an Event of Default occurs and Lender requires real estate tax escrow deposits to be made, but such Event of Default is thereafter cured, then, so long as no other Triggering Event exists, Lender shall not require further real estate tax deposits to be made.) If at any time Lender determines in its sole discretion that the amount of the monthly escrow payments made pursuant to this <u>Section 15.1(k)</u> are not sufficient to pay in full the next installment of real estate taxes then due, then upon written notice from Lender of the amount of any expected deficiency (and regardless of whether a Triggering Event then exists), Borrower shall then deposit funds equal to such amount with Lender. All payments deposited in the escrow account, and all interest accruing thereon, are pledged as additional collateral for the Loan. Notwithstanding Lender's holding of the escrow account, nothing herein shall obligate Lender to pay any real property taxes with respect to any portion of the Project at any time an Event of Default exists.

(1) <u>Personal Property</u>. All of Borrower's personal property, fixtures, attachments and equipment delivered upon, attached to or used in connection with the Construction or the operation of the Project shall always be located at the Project and shall be kept free and clear of all liens, encumbrances and security interests, other than as otherwise permitted under the Loan Documents.

(m) Leasing Restrictions. (i) Borrower shall not enter into any Leases pertaining to the Project without Lender's prior written approval in its sole discretion. Borrower shall provide Lender with a copy of any proposed Lease no less than ten (10) days prior to the proposed execution date of such Lease. Lender's approval of any Lease for the Commercial Space shall be predicated upon, among other things: (A) current tenant financial information in sufficient detail to assess the experience and credit worthiness of the tenant, (B) credit worthiness of tenant, (C) form and content of the Lease, including, among other things, tenant's obligation to provide, at least annually, its financial information to Borrower, (D) tenant's agreement to enter into an acceptable subordination, non-disturbance and attormment agreement, and (E) a minimum (7) seven year term, without any right to cancel prior to five (5) years; provided, however, Lender shall approve a five (5) year Lease term, but the tenant improvement funding permitted for such Lease shall be reduced proportionately. Except as otherwise approved by Lender, each Lease must provide for a minimum triple net rent of \$35 per Rentable Square Foot at all times throughout the term of the Loan, and, for purposes of calculating such rent, the total value of free

rent, non-standard lease concessions and tenant improvements in excess of the budgeted amount will be amortized at a per annum rate of ten percent (10%) over the full term of the Lease, including any free rent period. The resulting amount will be deducted from the lowest stated rent at any point in the term of the Lease, and that figure must be equal to or greater than the minimum rent noted above. Furthermore, Borrower will in no event offer more than six (6) months of free rent, tenant improvements in excess of twenty percent (20%) more than the budgeted amount, or other non-standard concessions unacceptable to the Lender.

(ii) Borrower shall not execute any Lease without written acknowledgement of Lease approval by the Lender, except that Leases of Commercial Space covering not more than 1,500 Rentable Square Feet individually, and 8,000 Rentable Square Feet in the aggregate, and otherwise satisfying the leasing requirements set forth above may be entered into without Lender's consent. Lender shall endeavor to provide (i) preliminary approval or disapproval of any proposed Lease within five (5) Business Days of the receipt of all pertinent information needed for approval, and (ii) final approval or disapproval of any proposed lease transaction within ten (10) Business Days of the receipt of final proposed lease documentation and any other information needed for its decision, and Lender's failure to approve a Lease within such ten (10) Business Day period shall be deemed its disapproval thereof.

(iii) A Lease, which has been approved by the Lender (or as to which Lender's approval is not required under the terms of clause (ii)), is hereinafter referred to as an "<u>Approved Lease</u>." Approval of a Lease shall not create a presumption that the Loan is In Balance as the result of excess tenant improvements or leasing commissions as set forth in <u>Section 12.8</u> hereof. To the extent that the Loan is not In Balance due to tenant improvements or leasing commissions in excess of the amount allowed pursuant to <u>Section 12.8</u>, Borrower shall deposit additional equity within ten (10) days following the execution of the Lease. An Approved Lease shall not be amended or modified (in any material respect) or terminated without the Lender's prior written consent.

(iv) Borrower shall not accept any rental payment under any Approved Lease in advance of its due date, other than acceptance of a prepayment of the first month's rent upon the execution of an Approved Lease.

(n) <u>Defaults Under Leases</u>. Borrower will not suffer or permit any material breach or default to occur in any of Borrower's obligations under any of the Leases, nor suffer or permit the same to terminate by reason of any failure of Borrower to meet any requirement of any Lease, including those requirements with respect to any time limitation within which any of Borrower's work is to be done or the space is to be available for occupancy by the lessee. Without limiting the generality of the foregoing, a default by Borrower under the terms of an Approved Lease that would allow the tenant to withhold or delay the payment of rent, or to terminate or cancel the Approved Lease, shall constitute a Default hereunder. Borrower shall notify Lender promptly in writing in the event a tenant commits a material default under a material Lease.

(o) Lender's Attorneys' Fees for Enforcement of Agreement. In case of any Event of Default hereunder, Borrower (in addition to Lender's attorneys' fees, if any, to be paid pursuant to Section 7.5) will pay Lender's reasonable attorneys' and paralegal fees (including, without

limitation, any attorney and paralegal fees and costs incurred in connection with any litigation or bankruptcy or administrative hearing and any appeals therefrom and any post-judgment enforcement action including, without limitation, supplementary proceedings) in connection with the enforcement of this Agreement; without limiting the generality of the foregoing, if at any time or times hereafter Lender employs counsel (whether or not any suit has been or shall be filed and whether or not other legal proceedings have been or shall be instituted) for advice or other representation with respect to the Project, this Agreement, or any of the other Loan Documents, or to protect, collect, lease, sell, take possession of, or liquidate any of the Project, or to attempt to enforce any security interest or lien in any portion of the Project, or to enforce any rights of Lender or Borrower's obligations hereunder, then in any of such events all of the reasonable attorneys' fees arising from such services, and any expenses, costs and charges relating thereto (including fees and costs of paralegals), shall constitute an additional liability owing by Borrower to Lender, payable on demand. Such attorneys' fees and expenses shall include fees and expenses of Lender's in-house counsel as specified in <u>Section 7.5</u>.

(p) <u>Appraisals</u>. Lender shall have the right to obtain a new or updated Appraisal of the Project from time to time, but not more frequently than two (2) times after Closing, provided Lender shall have the right to obtain a new or updated Appraisal at any time a Default or Event of Default exists. Borrower shall cooperate with Lender in this regard. If the Appraisal is obtained to comply with this Agreement or any applicable law or regulatory requirement, or bank policy promulgated to comply therewith, or if an Event of Default exists, Borrower shall pay the reasonable costs for any such Appraisal upon Lender's request.

#### (q) Financial Statements.

(i) Borrower and each Guarantor shall deliver or cause to be delivered to Lender annual financial statements with respect to Borrower and each Guarantor within ninety (90) days after the end of its fiscal year. Guarantor's financial statements shall be audited and Borrower's financial statements shall be certified by Borrower's managing member. Guarantor and, after the opening of the Commercial Space and/or Venue, Borrower shall also each deliver to Lender quarterly financial statements within forty five (45) days after the end of each fiscal quarter. Each such quarterly financial statement shall be accompanied by a certification by an authorized financial officer of Guarantor as to Guarantor's compliance with the Guarantor Financial Covenants and attaching information sufficient to demonstrate such compliance. All such financial statements shall be in a format approved in writing by Lender in Lender's reasonable discretion and in substance acceptable to Lender. Each financial statement shall be certified as true, complete and correct by its preparer and by Borrower or, in the case of each Guarantor's financial statements, by the Guarantor to whom it relates. Financial statements. In addition, such financial statements shall include disclosure of any pending or threatened litigation and judgments entered against Borrower or each Guarantor. Borrower and each Guarantor shall deliver to Lender Borrower's, each Guarantor's, and each entity in which a Guarantor has a material interest, federal and state tax returns by April 15 of each year (except that if the date on which such returns may be filed is extended beyond April 15, then such entity shall provide Lender with a copy of the extension request and shall furnish such tax returns to Lender within thirty (30) days after the date such returns are filed). Borrower shall inform Lender as to any filed or threatened

(in writing) litigation which would have a material adverse effect on Borrower's or each Guarantor's ability to perform their respective obligations under the Loan Documents promptly after learning thereof. Borrower and each Guarantor shall provide such additional financial information as Lender reasonably requires. Borrower shall during regular business hours permit Lender or any of its agents or representatives to have access to and examine all of its books and records regarding the development and operation of the Project and, in addition, agrees to provide Lender with copies of any purchase contracts pertaining to the Project. Borrower agrees that Lender may retain an investigator to research available public records and information relating to Borrower, the principals of Borrower and each Guarantor.

(ii) Borrower shall furnish to Lender within three (3) Business Days of receipt from time to time each proposed budget and "Operating Plan" received from Hotel Operator. Borrower shall not approve any such budget or Operating Plan without Lender's prior written consent, to be granted or withheld in Lender's reasonable discretion.

(iii) Borrower shall, within five (5) Business Days of receipt from time to time furnish to Lender all "Operating Reports" received under Section 10.2 of the Hotel Operating Agreement, all financial statements received under Section 10.3 of the Hotel Operating Agreement and all other material information or notices received from Hotel Operator.

(r) <u>Sign and Publicity</u>. Upon Lender's request, Borrower shall, at Lender's sole cost and expense, promptly erect a sign approved in advance by Lender in a conspicuous location on the Project during the Construction indicating that the financing for the Project is provided by Lender, so long as such sign complies with Laws and requirements of all Governmental Authorities. Lender reserves the right to publicize the making of the Loan. Without limiting the foregoing, following Closing, Lender shall have the right to announce publicly in print or otherwise that Lender has made and closed the Loan to Borrower. In connection therewith, Lender shall have the right to describe the Loan, including Borrower's name, the type of loan and the amount thereof (but excluding the interest rate, the amounts of Lender's fees and the specific nature of the non-recourse guaranty), and to identify the Project and the location thereof by way of description and/or photographs of the Project.

(s) Lost Note. Upon Lender's furnishing to Borrower an affidavit to such effect, Borrower shall, if the Note is mutilated, destroyed, lost or stolen, deliver to Lender, in substitution therefor, a new note containing the same terms and conditions as the Note

(t) Indemnification. Borrower shall indemnify Lender, including each party owning an interest in the Loan and their respective officers, directors, employees and consultants (each, an "<u>Indemnified Party</u>") and defend and hold each Indemnified Party harmless from and against all claims, injury, damage, loss, liability, cost and/or expense (including reasonable attorneys' fees, costs and expenses) of any and every kind to any persons or property by reason of (i) the Construction; (ii) the sale, operation or maintenance of the Project; (iii) any claim with respect to application, disposition or return of any Earnest Money Deposit or Upgrade Deposit, (iv) any breach of representation or warranty, default or Event of Default under this Agreement or any other Loan Document or related Document; or (v) any other matter arising in connection with the Loan, Borrower, Guarantors, any Residential Unit Purchaser or Sales Agreement, or the Project.

No Indemnified Party shall be entitled to be indemnified against its own gross negligence or willful misconduct. The foregoing indemnification shall survive repayment of the Loan.

# (u) No Additional Debt; Mezzanine Loan.

(i) Except for the Loan, Borrower shall not incur or guarantee any indebtedness (whether personal or nonrecourse, secured or unsecured) other than customary trade payables paid within ninety (90) days after they are incurred. Except for the liens securing the Loan and except for the Permitted Exceptions, Borrower shall keep the Project free and clear of liens, provided, however, mechanics' liens may be contested in compliance with <u>Section 15.1(f)</u>.

(ii) No mezzanine financing may be borrowed by Borrower or its constituent entities, except that Mezzanine Borrower shall be entitled to obtain a Mezzanine Loan upon delivery of forty-five (45) days' prior written notice to Lender, provided (A) the identity of the Mezzanine Lender has been approved in writing by Lender, (B) Mezzanine Lender has executed the Mezzanine Loan Intercreditor Agreement, as prepared by Lender's counsel in compliance with the requirements set forth in the definition of Mezzanine Loan Intercreditor Agreement, (C) the Mezzanine Loan Documents are acceptable to Lender in its reasonable discretion, (D) the Mezzanine Loan shall have been closed and the Qualifying Portion thereof shall have been fully funded for costs in the Budget, with evidence of such funding delivered to Lender, prior to Full Loan Opening, (E) Borrower shall have furnished to Lender current contact information for each of the individuals comprising the Mezzanine Lender, and (F) such other reasonable requirements of Lender have been satisfied. Such right of Borrower to obtain a Mezzanine Loan shall expire unless such Mezzanine Loan has closed and the Qualifying Portion thereof has fully funded prior to Full Loan Opening.

(v) <u>Compliance With Laws</u>. Borrower shall comply with all applicable requirements (including applicable Laws) of any Governmental Authority having jurisdiction over Borrower or the Project.

(w) <u>Organizational Documents</u>. Borrower shall not, without the prior written consent of Lender, permit or suffer (i) a material amendment or modification of its organizational documents, (ii) the admission of any new member, partner or shareholder in violation of <u>Section 17.2</u> of this Agreement, or (iii) any dissolution or termination of its existence.

(x) <u>Furnishing Reports</u>. Upon Lender's request, Borrower shall provide Lender with copies of all inspections, reports, test results and other information received by any Borrower, that in any way relate to the Project or any part thereof.

(y) <u>Management Contracts</u>. Borrower shall not enter into, modify, amend, terminate or cancel any management, sales or marketing contracts for the Project, without the prior written approval of Lender, not to be unreasonably withheld or delayed.

(z) <u>Furnishing Notices</u>. Borrower shall provide Lender with copies of all material notices pertaining to the Project received by Borrower from any Governmental Authority or insurance company within seven (7) days after such notice is received. Borrower shall promptly

notify Lender of any judgment entered against, or any material litigation filed against, Borrower or Guarantor.

(aa) <u>Construction Contracts</u>. Borrower shall not enter into, materially modify or materially amend, or terminate or cancel any material contracts for the Construction, without the prior written approval of Lender. Borrower shall not enter into any contract which would cause the Loan to cease to be In Balance. Borrower will furnish Lender promptly after execution thereof, executed copies of all contracts between Borrower, architects, engineers and contractors and all subcontracts between the General Contractor or contractors and all of their subcontractors and suppliers, which contracts and subcontracts may not have been furnished pursuant to <u>Section 9.1(a)</u> at Closing. The development agreement between the Borrower and Guarantor (or Guarantor's Affiliate) may be modified without Lender's consent so long as such modification does not materially adversely affect Lender and may be terminated upon satisfaction of the same conditions as are applicable to CJUF's replacement of Stratus Block 21 Investments, L.P. as the managing member of Borrower under <u>Section 17.2</u> below.

(bb) <u>Correction of Defects</u>. Within five (5) days after Borrower acquires knowledge of or receives notice of a material defect in the Improvements or any material departure from the Approved Plans and Specifications, or any other requirement of this Agreement, Borrower shall proceed with diligence to correct all such defects and departures.

(cc) <u>Hold Disbursements in Trust</u>. Borrower shall receive and hold in trust for the sole benefit of Lender (and not for the benefit of any other person, including, but not limited to, contractors or any subcontractors) all advances made hereunder directly to Borrower, for the purpose of paying costs of the Construction in accordance with the Budget. Borrower shall use the proceeds of the Loan solely for the payment of costs as specified in the Budget. Borrower shall pay all other costs, expenses and fees relating to the acquisition, equipping, use, sale and operation of the Project.

(dd) <u>Foundation Survey</u>. Not later than thirty (30) days after completion of the foundation with respect to the Improvements, Borrower shall furnish to Lender a survey of the Land with the foundation of the Improvements located thereon satisfying the requirements set forth in <u>Section 8.1(d)</u>.

(ee) <u>Alterations</u>. Without the prior written consent of Lender, Borrower shall not make any material alterations to the Project (other than completion of the Construction in accordance with the Approved Plans and Specifications).

(ff) <u>Cash Distributions</u>. Borrower shall not make any distributions to its partners or other equity holders, Guarantor or any Affiliate of Borrower or Guarantor until the Loan has been repaid, except for the disbursements set forth in <u>Section 12.7</u> hereof.

(gg) Injunctive Proceedings. If any proceedings are filed seeking to enjoin or otherwise prevent or declare invalid or unlawful the construction, sale, occupancy, maintenance or operation of the Project or the Residential Units, Borrower shall cause such proceedings to be diligently contested in good faith, and in the event of an adverse ruling or decision, shall prosecute all allowable appeals therefrom, and shall, without limiting the generality of the

foregoing, resist the entry or seek the stay of any temporary or permanent injunction that may be entered and use commercially reasonable efforts to bring about a favorable and speedy disposition of all such proceedings. Lender's reasonable costs and disbursements (including attorney's fees) in connection with any such proceedings, whether or not Lender is a party thereto, shall be deemed to be expenses of the Loan payable by Borrower in accordance with <u>Article 7</u> of this Agreement.

(hh) <u>Commencement of Work</u>. No work or Construction of any character shall be commenced upon, nor shall any materials be delivered to, the Project until (i) the Deed of Trust has been recorded in the records of Travis County, Texas and a file stamped copy thereof has been received by Lender, or, if work has commenced with Lender's consent, the Title Insurer has agreed to insure Lender to Lender's satisfaction and (ii) the subguard insurance required by Lender has been properly received and approved by Lender.

(ii) Operating Revenues. In the event the Project produces gross operating revenues, Borrower shall establish with Lender (i) an interest bearing operating account for the Project (other than the Hotel) into which Borrower shall deposit all revenue associated with the Project (other than the Hotel), and from which Borrower shall make withdrawals to pay all operating expenses of the Project (other than the Hotel) and, to the extent of any remaining Net Operating Income, interest due on the Loan in accordance with Section 10.4, and (ii) an interest bearing Hotel revenue account into which Borrower shall deposit or shall cause to be deposited all Net Operating Income distributed by Hotel Operator to Borrower, and from which Borrower shall pay interest due on the Loan in accordance with Section 10.4. Borrower's failure to deposit operating revenues or Net Operating Income into such accounts as set forth above shall constitute an Event of Default hereunder. In the event that Net Operating Income, together, in such accounts exceeds the monthly interest due, the then excess shall remain in the applicable account for the payment of future month's interest. Borrower hereby pledges each such account and all funds on deposit therein to Lender. Borrower hereby pledges, to the extent permitted by, and subject to the limits of, the Hotel Operating Agreement, all accounts to be established in accordance with the terms and provisions of the Hotel Operating Agreement, and all funds on deposit of the Hotel, Borrower shall deliver, or shall cause to be delivered, to Lender, a control agreement from each depository bank with which each such Hotel operating account has been established substantially similar in form and substance to the reasonable form of same prepared by Lender's legal counsel.

(jj) <u>Environmental Remediation</u>. Borrower shall perform all recommended remediation set forth in those certain Environmental Reports, in each case, by the dates such remediation is recommended to be completed in such report, except in the case of the Waste Management Plan, with respect to which all recommendations must be complied with by Full Loan Opening; provided, however, that the water filtration system discussed in the Waste Management System shall be completed at such later date as is sufficient to comply with the recommendations of the Waste Management Plan, and same shall continue to be operated and maintained in accordance with the Waste Management Plan during the construction and operation of the Project.

(kk) Hotel Documents. Borrower shall timely perform and comply with all of its obligations under the Hotel Documents and all other agreements with Hotel Operator, Starwood

Hotels & Resorts Worldwide, Inc., and their affiliates with respect to the Hotel, and shall keep all such agreements in full force and effect. Borrower shall not modify, amend or terminate any such agreements without Lender's prior written consent, to be given in its reasonable discretion.

(11) Pre-sale Requirement. The Pre-sale Requirement must remain satisfied throughout the Loan Term.

#### 15.2 Single Purpose Entity Covenants.

Borrower hereby represents, warrants and covenants that without Lender's prior written consent, which may be withheld in Lender's sole discretion, and except as otherwise expressly permitted hereunder, Borrower has not, will not and shall not:

(a) engage in any business or activity other than the ownership, management, construction and operation of the Project;

(b) acquire or own any material assets other than the Project, and such incidental personal property as may be necessary for the operation of the Project;

(c) merge into or consolidate with any Person or entity or dissolve, terminate or liquidate in whole or in part, transfer or otherwise dispose of all or substantially all of its assets or change in its legal structure or acquire by purchase or otherwise all or substantially all the business or assets of, or stock or other evidence of beneficial ownership of, any Person, without in each case the prior written consent of Lender, which consent may be withheld or delayed in Lender's sole and absolute discretion;

(d) fail to preserve its existence as an entity duly organized, validly existing and in good standing under the laws of the State of Delaware or without the prior written consent of Lender amend, modify, terminate or fail to comply with the provisions of Borrower's formation documents, as same may be further amended or supplemented, if such amendment, modification, termination or failure to comply would adversely affect the ability of Borrower to perform its obligations under the applicable Loan Documents or jeopardize Borrower's existence as a single purpose entity;

(e) own any subsidiary or make any investment in, any Person without the consent of Lender;

(f) commingle its assets with the assets of any of its Affiliates, or of any other Person;

(g) incur any debt, secured or unsecured, direct or contingent (including guaranteeing any obligation), other than other indebtedness expressly permitted under this Agreement, provided that any such debt is paid before such payments are past due;

(h) become insolvent and fail to pay its debts, liabilities and obligations of any kind, including all administrative expenses, from its own separate assets as the same shall become due;

(i) fail to maintain its records, books or accounts and bank accounts separate and apart from those of any Affiliate of Borrower, any Affiliate of a partner of Borrower and any other Person or entity;

(j) enter into any contract or agreement with any of its Affiliates except on terms and conditions that are intrinsically fair and substantially similar to those that would be available on an arms-length basis with third parties other than an Affiliate or as provided in this Agreement;

(k) seek the dissolution or winding up in whole, or in part, of Borrower;

(l) maintain its assets in such manner that it will be costly or difficult to segregate, ascertain or identify its individual assets from those of any partner, member, constituent, principal or Affiliate of Borrower, or any member, general partner, principal or Affiliate thereof, or any other person;

(m) hold itself out to be responsible for the debts of another Person except as specifically permitted in this Agreement;

(n) make any loans or advances to any third party, including any partner, member, constituent, principal or Affiliate of Borrower or any member, general partner, principal or Affiliate thereof;

(o) fail to file its own tax returns; provided, however, that Borrower may file a consolidated tax return with any of its Affiliates, but only because such consolidated tax return is required by applicable Laws, and provided such consolidated tax return includes footnotes identifying the separate assets or liabilities of Borrower and/or its Affiliates, as applicable;

(p) fail either to hold itself out to the public as a legal entity separate and distinct from any other Person or to conduct its business solely in its own name in order not (A) to mislead others as to the identity with which such other party is transacting business, or (B) to suggest that it is responsible for the debts of any third party (including any partner, principal, member of Affiliate or Borrower, or any partner, principal, member or Affiliate thereof) except as specifically permitted in this Agreement;

(q) fail to maintain adequate capital for the normal obligations reasonably foreseeable in a business of its size and character and in light of its contemplated business operations;

(r) file or consent to the filing of any petition, either voluntary or involuntary, to take advantage of any applicable insolvency, bankruptcy, liquidation or reorganization statute, or make an assignment for the benefit of creditors;

(s) fail to maintain a reasonably sufficient number of employees in light of its contemplated business operations if any employees are required for the contemplated business operations;

(t) fail to allocate fairly and reasonably any overhead expenses that are shared with any of their respective partners, members, principals or Affiliates of Borrower, any guarantor or indemnitor, or any partner, member, principal or Affiliate thereof, including paying for office

space and services performed by any employee of any of their respective partners, of any members, principals or Affiliates, any guarantor or indemnitor, or any partner, member, principal or Affiliate of any thereof;

(u) except to the extent required by generally accepted accounting principles or applicable law, fail to maintain separate financial statements, which shows its assets and liabilities, separate and apart from those of any other Person or entity and not have its assets listed in the financial statement of any other entity; or

(v) fail to correct any known misunderstanding regarding its separate identity.

#### 15.3 Authorized Representative.

Borrower hereby appoints the Authorized Representative as its authorized representative for purposes of dealing with Lender on behalf of Borrower in respect of any and all matters in connection with this Agreement, the other Loan Documents, and the Loan. The Authorized Representative shall have the power, in his discretion, to give and receive all notices, monies, approvals, and other documents and instruments, and to take any other action on behalf of Borrower. All actions by the Authorized Representative shall be final and binding on Borrower. Lender may rely on the authority given to the Authorized Representative until actual receipt by Lender of a duly authorized resolution substituting a different person as the Authorized Representative. If Borrower appoints more than one Authorized Representative, the action of any one Authorized Representative shall be binding and sufficient.

#### Article 16

# CASUALTIES AND CONDEMNATION

#### 16.1 Lender's Election to Apply Proceeds on Indebtedness.

(a) Subject to the provisions of <u>Section 16.1(b)</u> below, Lender may elect to negotiate, settle, collect, retain and apply upon the indebtedness of Borrower under this Agreement or any of the other Loan Documents all proceeds of insurance or condemnation (individually and collectively referred to as "<u>Proceeds</u>") after deduction of all expenses of collection and settlement, including reasonable attorneys' and adjusters' fees and charges. Lender shall have the right to participate with Borrower in negotiation of any settlement, adjustment or compromise of any claim arising in connection with a casualty to the Improvements or any condemnation of all or part of the Project; provided, however, if an Event of Default exists, Lender shall have the right to settle any claim without Borrower's participation or consent. Any Proceeds remaining after repayment of the indebtedness under the Loan Documents shall be paid by Lender to Borrower.

(b) Notwithstanding anything in Section 16.1(a) to the contrary, in the event of any casualty to the Improvements or any condemnation of part of the Project, Lender agrees to make available the Proceeds for restoration of the Improvements if (i) no Event of Default exists, (ii) all Proceeds are deposited with Lender, (iii) in Lender's reasonable judgment, the amount of Proceeds available for restoration of the Improvements (together with undisbursed proceeds of the Loan, if any, allocated for the cost of the Construction and any sums deposited with Lender



by Borrower for such purpose) is sufficient to pay the full and complete costs of such restoration, (iv) the cost of restoration does not exceed twenty percent (20%) of the stated amount of the Note, (v) if Lender determines that the cost of restoration exceeds \$5,000,000, Lender determines in its reasonable discretion that the values set forth in the Appraisal reviewed and approved by Lender at Closing remain valid; (vi) in Lender's reasonable determination, the Project can be restored to an architecturally and economically viable project in compliance with applicable Laws, (vii) Guarantor reaffirms the Completion and Carveout and the Limited Payment Guaranty in writing, (viii) Borrower shall have provided evidence reasonably acceptable to Lender that following restoration (and completion of the Project) the Pre-Sale Requirement will remain satisfied, (ix) the Hotel Documents, and the management agreement for the Venue, will each remain in full force and effect; and (x) in Lender's reasonable determination, such restoration is likely to be completed so that the Residential Units sufficient to satisfy the Pre-sale Requirement may be delivered to all Residential Unit Purchasers prior to the outside delivery dates contained in their respective Sales Agreements and in any event not later than the Maturity Date. On and after such time as the condominium is formed, Borrower agrees to vote its votes with respect to all unsold Residential Units (and to cause all Borrower appointed directors to vote their votes) in a manner consistent with the provisions of this <u>Article 16</u>.

#### 16.2 Borrower's Obligation to Rebuild and Use of Proceeds Therefor.

In case Lender does not elect to apply or does not have the right to apply the Proceeds to the indebtedness, as provided in <u>Section 16.1</u> above, Borrower shall:

(a) Proceed with diligence to make settlement with insurers or the appropriate Governmental Authorities and cause the Proceeds to be deposited with Lender;

(b) In the event the Proceeds and the available proceeds of the Loan are insufficient to assure Lender that the Loan will be In Balance, promptly deposit with Lender any amount necessary to place the Loan In Balance; and

(c) Promptly proceed with the assumption of construction of the Improvements, including the repair of all damage resulting from such fire, condemnation or other cause and restoration to its former condition.

Any request by Borrower for a disbursement by Lender of Proceeds and funds deposited by Borrower shall be treated by Lender as if such request were for an advance of the Loan hereunder, and the disbursement thereof shall be conditioned upon Borrower's compliance with and satisfaction of the same conditions precedent as would be applicable under this Agreement for an advance of the Loan.

#### Article 17

#### ASSIGNMENTS BY LENDER AND BORROWER

#### 17.1 Assignments and Participations.

Lender may from time to time, with written notice to Borrower, sell the Loan and the Loan Documents (or any interest therein) and may grant participations in the Loan. Borrower agrees to reasonably cooperate with Lender's efforts to do any of the foregoing and to execute all documents reasonably required by Lender in connection therewith that do not materially adversely affect Borrower's or Guarantor's rights under the Loan Documents or materially increase Borrower's or Guarantor's obligations under the Loan Documents.

#### 17.2 Prohibition of Assignments and Transfers by Borrower.

Borrower shall not assign or attempt to assign its rights under this Agreement and any purported assignment shall be void. Without the prior written consent of Lender, in Lender's sole discretion, Borrower shall not suffer or permit (a) any change in the ownership, or management or economic interests (whether direct or indirect) of the Project, or (b) any Transfer. Notwithstanding the foregoing, (A) so long as CJUF continues to hold the ownership interest in Borrower which CJUF held at Closing and no monetary Event of Default then exists (including without limitation default in making a Deficiency Deposit), or if a monetary Event of Default does exist and is cured by Borrower or CJUF, CJUF may with prior written notice to Lender (but without Lender's consent) replace Stratus Block 21 Investments, L.P. as the managing member of Borrower (with Stratus Block 21 Investments, L.P. retaining its current ownership interest in Borrower which CJUF held at Closing, CJUF may request that Lender approve CJUF replacing Stratus Block 21 Investments, L.P. as managing member of Default then exists, so long as CJUF continues to hold the ownership interest in Borrower (with Stratus Block 21 Investments, L.P. as managing member of Default then exists, so long as CJUF continues to hold the ownership interest in Borrower (with Stratus Block 21 Investments, L.P. as managing member of Borrower CJUF replacing Stratus Block 21 Investments, L.P. as managing member of Borrower (with Stratus Block 21 Investments, L.P. as managing member of Borrower (with Stratus Block 21 Investments, L.P. as managing member of an or ensitie current ownership interest but as non-managing member), and Lender shall grant or deny such consent in its reasonable discretion, and (C) the following Transfers of interests as Borrower's direct or indirect constituent entities shall be permitted without Lender's consent: (i) Transfers of direct or indirect interests in Canyon-Johnson Urban Fund II, L.P., (ii) Transfers of non-controlling interests in Guarantor, (iiii

# 17.3 Prohibition of Transfers in Violation of ERISA.

In addition to the prohibitions set forth in Section 17.2 above, Borrower shall not assign, sell, pledge, encumber, transfer, hypothecate or otherwise dispose of its interest or rights in this Agreement or in the Project, or attempt to do any of the foregoing or suffer any of the foregoing, nor shall any party owning a direct or indirect interest in Borrower assign, sell, pledge, Deed of Trust, encumber, transfer, hypothecate or otherwise dispose of any of its rights or interest (direct or indirect) in Borrower, attempt to do any of the foregoing or suffer any of the foregoing, if such action would cause the Loan, or the exercise of any of Lender's rights in connection therewith, to constitute a prohibited transaction under ERISA or the Internal Revenue Code or otherwise result

in Lender being deemed in violation of any applicable provision of ERISA. Borrower agrees to indemnify and hold Lender free and harmless from and against all losses, costs (including reasonable attorneys' fees and expenses), taxes, damages (including consequential damages) and expenses Lender may suffer by reason of the investigation, defense and settlement of claims and in obtaining any prohibited transaction exemption under ERISA necessary or desirable in Lender's sole judgment or by reason of a breach of the foregoing prohibitions. The foregoing indemnification shall be a recourse obligation of Borrower and shall survive repayment of the Note, notwithstanding any limitations on recourse contained herein or in any of the Loan Documents.

#### 17.4 Successors and Assigns.

Subject to the foregoing restrictions on transfer and assignment contained in this <u>Article 17</u>, this Agreement shall inure to the benefit of and shall be binding on the parties hereto and their respective successors and permitted assigns.

#### Article 18

# TIME OF THE ESSENCE

# 18.1 Time is of the Essence.

Time is of the essence under this Agreement.

#### Article 19

# **EVENTS OF DEFAULT**

### 19.1 Events of Default.

The occurrence of any one or more of the following shall constitute an "Event of Default" as said term is used herein:

(a) Failure of Borrower (i) (A) to make any principal payment when due, (B) to pay any interest within ten (10) days after the date when due or (C) to observe or perform any of the other covenants or conditions by Borrower to be performed under the terms of this Agreement or any other Loan Document concerning the payment of money, for a period of ten (10) days after written notice from Lender that the same is due and payable; or (ii) for a period of thirty (30) days after written notice from Lender, to observe or perform any non-monetary covenant or condition contained in this Agreement or any other Loan Documents, provided that if any such failure concerning a non-monetary covenant or condition is susceptible to cure and cannot reasonably be cured within said thirty (30) day period, then Borrower shall have an additional sixty (60) day period to cure such failure and no Event of Default shall be deemed to exist hereunder so long as Borrower commences such cure within the initial thirty (30) day period and diligently and in good faith pursues such cure to completion within such resulting ninety (90) day period from the date of Lender's notice; and provided further that if a different notice or grace period is specified under any other subsection of this <u>Article 19</u> with respect to a particular breach, or if another subsection of this <u>Article 19</u> applies to a particular breach and does not expressly provide for a notice or grace period, the specific provision shall control.



(b) The disapproval by Lender or Lender's Consultant at any time of any material aspect of the construction work and failure of Borrower to cause the same to be corrected to the satisfaction of Lender within the cure period provided in <u>Section 19(a)(ii)</u> above.

(c) (i) Failure of Borrower to commence Construction of the Project on or before the Required Construction Commencement Date, (ii) a discontinuance of the Construction for a period of twenty (20) consecutive days (other than by reason of Unavoidable Delays), or in any event a delay in the Construction of any component of the Project so that such component is not, in Lender's sole judgment, likely to be completed on or before the applicable Completion Date, (iii) failure of Borrower to comply with the Construction Schedule resulting in a delay of more than ninety (90) days behind the Construction Schedule with respect to any material component of Construction, or (iv) failure to complete Construction of the all or any component of the Improvements substantially in accordance with the Approved Plans and Specifications (or in accordance with any changes therein that may be approved in writing by Lender or as to which Lender's approval is not required) on or before the applicable Completion Date.

(d) The bankruptcy or insolvency of the General Contractor and failure of Borrower to procure a contract with a new contractor satisfactory to Lender within forty-five (45) days from the occurrence of such bankruptcy or insolvency.

# (e) Any Transfer or other disposition in violation of Section 17.2 or 17.3.

(f) Any warranty, representation, statement, report or certificate made now or hereafter by Borrower or any Guarantor is untrue or incorrect in any material respect at the time made or delivered, provided that if such breach is reasonably susceptible to cure, then no Event of Default shall exist so long as Borrower cures said breach (i) within the notice and cure period provided in (a)(i) above for a breach that can be cured by the payment of money or (ii) within the notice and cure period provided in (a)(ii) above for any other breach.

(g) Borrower or any Guarantor shall commence a voluntary case concerning Borrower or any Guarantor under the Bankruptcy Code; or an involuntary proceeding is commenced against Borrower or any Guarantor under the Bankruptcy Code and relief is ordered against Borrower or any Guarantor, or the petition is controverted but not dismissed or stayed within sixty (60) days after the commencement of the case, or a custodian (as defined in the Bankruptcy Code) is appointed for or takes charge of all or substantially all of the property of Borrower or any Guarantor; or Borrower or any Guarantor commences any other proceedings under any reorganization, arrangement, readjustment of debt, relief of debtors, dissolution, insolvency or liquidation or similar Law of any jurisdiction whether now or hereafter in effect relating to Borrower or any Guarantor; or there is commenced against Borrower or any Guarantor any such proceeding that remains undismissed or unstayed for a period of sixty (60) days; or Borrower or any Guarantor fails to controvert in a timely manner any such case under the Bankruptcy Code or any such proceeding, or any order of relief or other order approving any such case or proceeding is entered; or Borrower or any Guarantor by any act or failure to act indicates its consent to, approval of, or acquiescence in any such case or proceeding or unstayed for a period of sixty (60) days.

(h) Borrower or any Guarantor shall make an assignment for the benefit of creditors, or shall admit in writing its inability to pay its debts generally as they become due, or shall consent to the appointment of a receiver or trustee or liquidator of all of its property or the major part thereof or if all or a substantial part of the assets of Borrower or any Guarantor are attached, seized, subjected to a writ or distress warrant, or are levied upon, or come into the possession of any receiver, trustee, custodian or assignee for the benefit of creditors.

(i) Any Guarantor shall dissolve or its existence shall otherwise terminate.

(j) Any Guarantor shall breach any of the Guarantor Financial Covenants.

(k) Borrower is enjoined, restrained or in any way prevented by any court order from constructing or operating the Project or marketing or selling Residential Units for a period in excess of thirty (30) days.

(1) Failure by Borrower to (i) make any Deficiency Deposit to Lender within the time and in the manner required by <u>Article 11</u> hereof, or (ii) deposit Net Operating Income into the operating account addressed in <u>Section 15.1(ii)(i)</u> or Hotel operating revenue into the Hotel operating account addressed in <u>Section 15.1(ii)(ii)</u>.

(m) One or more final, unappealable judgments are entered against (i) Borrower in amounts aggregating in excess of \$250,000, or (ii) against any Guarantor in amounts aggregating in excess of \$250,000 for any Guarantor, and said judgments are not paid, stayed or bonded over within thirty (30) days after entry.

(n) Borrower or any Guarantor shall fail to pay any debt owed by it or him, as applicable, when due (either by maturity (without forbearance) or acceleration) to Lender or any other party (other than a failure or default for which Borrower's maximum liability does not exceed \$200,000 and any Guarantor's maximum liability does not exceed \$500,000).

(o) If a Material Adverse Change occurs with respect to Borrower, the Project or any Guarantor.

(p) Borrower shall fail to deliver the Insurance Policy within ninety (90) days after the date hereof.

(q) The Pre-Sale Requirement shall cease to be satisfied.

(r) Borrower shall default under the Hotel Documents and shall fail to cure such default within the shorter of (i) the applicable cure period set forth in <u>Section 19.1(a)</u>, and (ii) the applicable grace or cure period set forth in the applicable Hotel Document, or any Hotel Document shall otherwise cease to be in full force and effect (or any other event occurring thereunder which would give the Hotel Operator the right to terminate a Hotel Document).

(s) Borrower shall default under the management agreement for the Venue reviewed and approved by Lender in accordance with this Agreement, and shall fail to cure such default within the applicable grace or cure period set forth in such management agreement, or such management agreement shall otherwise cease to be in full force and effect.

(t) Borrower shall fail to deliver to Lender any executed Subcontract required under Exhibit O within thirty days (30) following in the delivery date for such Subcontract as set forth in Exhibit O.

(u) The occurrence of any other event or circumstance identified as an Event of Default in this Agreement or under any of the other Loan Documents and the expiration of any applicable grace or cure periods, if any, specified for such Event of Default herein or therein, as the case may be.

Notwithstanding anything to the contrary set forth herein, in the event that any Event of Default shall occur with respect to Guarantor under <u>Section 19.1(g)</u>, (h), (i), (j), (m), (n) or (o) (a "<u>Guarantor Financial Default</u>"), Borrower or Guarantor may cure such Event of Default by depositing with Lender, within ten (10) Business Days following Lender's written notice to Borrower of such Event of Default, an irrevocable and unconditional letter of credit in the amount of Twenty Million Dollars (\$20,000,000.00) from an issuer and otherwise in form and substance satisfactory to Lender in its sole discretion. Once such letter of credit is deposited with Lender pursuant hereto and for so long as same remains in effect, no Guarantor Financial Default shall be considered an Event of Default hereunder. Lender may draw upon such letter of credit and apply such drawn funds to the indebtedness under the Loan Documents in such order as Lender shall elect if an Event of Default occurs thereafter (except for a Guarantor Financial Default), or the letter of credit shall expire in less than thirty (30) days, or if the Loan is not paid in full on the Maturity Date.

# Article 20

# LENDER'S REMEDIES IN EVENT OF DEFAULT

#### 20.1 Remedies Conferred Upon Lender.

Upon the occurrence and during the continuance of any Event of Default, Lender may pursue any one or more of the following remedies concurrently or successively, it being the intent hereof that none of such remedies shall be to the exclusion of any other:

(a) Take possession of the Project and complete the Construction and do anything that is necessary or appropriate in its sole judgment to fulfill the obligations of Borrower under this Agreement and the other Loan Documents, including either the right to avail itself of and procure performance of existing contracts or let any contracts with the same contractors or others. Without restricting the generality of the foregoing and for the purposes aforesaid, Borrower hereby appoints and constitutes Lender its lawful attorney-in-fact with full power of substitution in the Project to complete the Construction in the name of Borrower; to use unadvanced funds remaining under the Note or that may be reserved, escrowed or set aside for any purposes hereunder at any time, or to advance funds in excess of the face amount of the Note, to complete the Construction; to make changes in the Plans and Specifications that shall be necessary or desirable to complete the Construction in substantially the manner contemplated by the Plans and Specifications; to retain or employ new general contractors, subcontractors, architects, engineers and inspectors as shall be required for said purposes; to pay, settle or compromise all existing bills and claims that may be liens or security interests, or to avoid such bills and claims becoming liens against the Project; to execute all applications and certificates in

the name of Borrower prosecute and defend all actions or proceedings in connection with the Improvements or Project; and to do any and every act that Borrower might do in its own behalf; it being understood and agreed that this power of attorney shall be a power coupled with an interest and cannot be revoked;

(b) Withhold further disbursement of the proceeds of the Loan and/or terminate Lender's obligations to make further disbursements hereunder;

(c) Declare the Note to be immediately due and payable;

(d) Use and apply any monies or letters of credit deposited by Borrower with Lender, regardless of the purposes for which the same was deposited, to cure any such default or to apply on account of any indebtedness under this Agreement that is due and owing to Lender;

(e) Fund proceeds of the Loan to Guarantor to complete the Project (if Guarantor is entitled to receive such proceeds under the Completion and Carveout Guaranty or Lender otherwise elects to fund such proceeds to Guarantor), and in such event such proceeds shall be considered disbursements of the Loan, whether or not requested by Borrower's authorization of such disbursements being deemed given hereby);

(f) Assess interest on all amounts outstanding under the Note at the Default Rate; and

(g) Exercise or pursue any other remedy or cause of action permitted under this Agreement or any other Loan Documents, or conferred upon Lender by operation of Law.

Notwithstanding the foregoing, upon the occurrence of any Event of Default under <u>Section 19.1(g)</u> with respect to Borrower, all amounts evidenced by the Note shall automatically become due and payable, without any presentment, demand, protest or notice of any kind to Borrower.

# Article 21

#### **GENERAL PROVISIONS**

#### 21.1 Captions.

The captions and headings of various Articles, Sections and subsections of this Agreement and Exhibits pertaining hereto are for convenience only and are not to be considered as defining or limiting in any way the scope or intent of the provisions hereof.

#### 21.2 Modification; Waiver.

No modification, waiver, amendment or discharge of this Agreement or any other Loan Document shall be valid unless the same is in writing and signed by the party against which the enforcement of such modification, waiver, amendment or discharge is sought.

#### 21.3 Governing Law.

Irrespective of the place of execution and/or delivery, this Agreement shall be governed by, and shall be construed in accordance with, the laws of the State of Illinois.

#### 21.4 Acquiescence Not to Constitute Waiver of Lender's Requirements.

Each and every covenant and condition for the benefit of Lender contained in this Agreement may be waived by Lender, provided, however, that to the extent that Lender may have acquiesced in any noncompliance with any construction or nonconstruction conditions precedent to the Closing Funding, the Full Opening of the Loan or to any subsequent disbursement of Loan proceeds, such acquiescence shall not be deemed to constitute a waiver by Lender of such requirements with respect to any future disbursements of Loan proceeds.

### 21.5 Disclaimer by Lender.

This Agreement is made for the sole benefit of Borrower and Lender, and no other person or persons shall have any benefits, rights or remedies under or by reason of this Agreement, or by reason of any actions taken by Lender pursuant to this Agreement. Lender shall not be liable to any contractors, subcontractors, supplier, architect, engineer, tenant or other party for labor or services performed or materials supplied in connection with the Construction. Lender shall not be liable for any debts or claims accruing in favor of any such parties against Borrower or others or against the Project. Lender, by making the Loan or taking any action pursuant to any of the Loan Documents, shall not be deemed a partner or a joint venturer with Borrower or fiduciary of Borrower. No payment of funds directly to a contractor or subcontractor or provider of services shall be deemed to create any thirdparty beneficiary status or recognition of same by Lender. Without limiting the generality of the foregoing:

(a) Lender shall have no liability, obligation or responsibility whatsoever with respect to the Construction. Any inspections of the Construction made by or through Lender are for purposes of administration of the Loan only and neither Borrower nor any third party is entitled to rely upon the same with respect to the quality, adequacy or suitability of materials or workmanship, conformity to the Plans and Specifications, state of completion or otherwise;

(b) Lender neither undertakes nor assumes any responsibility or duty to Borrower to select, review, inspect, supervise, pass judgment upon or inform Borrower of any matter in connection with the Project, including matters relating to the quality, adequacy or suitability of: (i) the Plans and Specifications, (ii) architects, contractors, subcontractors and material suppliers employed or utilized in connection with the Construction, or the workmanship of or the materials used by any of them, or (iii) the progress or course of Construction and its conformity or nonconformity with the Plans and Specifications; Borrower shall rely entirely upon its own judgment with respect to such matters, and any review, inspection, supervision, exercise of judgment or supply of information to Borrower by Lender in connection with such matters is for the protection of Lender only, and neither Borrower nor any third party is entitled to rely thereon; and

(c) Lender owes no duty of care to protect Borrower, Guarantor, or any Tenant or Residential Unit Purchaser against negligent, faulty, inadequate or defective building or construction.

# 21.6 Partial Invalidity; Severability.

If any of the provisions of this Agreement, or the application thereof to any person, party or circumstances, shall, to any extent, be invalid or unenforceable, the remainder of this Agreement, or the application of such provision or provisions to persons, parties or circumstances other than those as to whom or which it is held invalid or unenforceable, shall not be affected thereby, and every provision of this Agreement shall be valid and enforceable to the fullest extent permitted by law.

# 21.7 Definitions Include Amendments.

Definitions contained in this Agreement that identify documents, including, but not limited to, the Loan Documents, shall be deemed to include all amendments and supplements to such documents from the date hereof, and all future amendments, modifications, and supplements thereto entered into from time to time to satisfy the requirements of this Agreement or otherwise with the consent of Lender. Reference to this Agreement contained in any of the foregoing documents shall be deemed to include all amendments and supplements to this Agreement.

#### 21.8 Execution in Counterparts.

This Agreement may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which when so executed shall be deemed to be an original and all of which taken together shall constitute one and the same agreement.

#### 21.9 Entire Agreement.

This Agreement, taken together with all of the other Loan Documents and all certificates and other documents delivered by Borrower to Lender, embodies the entire agreement and supersede all prior agreements, written or oral, relating to the subject matter hereof.

#### 21.10 Waiver of Damages.

In no event shall Lender be liable to Borrower for consequential, punitive or exemplary damages, whatever the nature of a breach by Lender of its obligations under this Agreement or any of the Loan Documents, and Borrower, for itself and Guarantor, waives all claims for consequential, punitive or exemplary damages.

# 21.11 Claims Against Lender.

Lender shall not be in default under this Agreement, or under any other Loan Documents, unless a written notice specifically setting forth the claim of Borrower shall have been given to Lender and Lender shall have failed to cure such default within thirty (30) days after receipt of written notice from Borrower, provided that if any such default cannot reasonably be cured

within said thirty (30) day period, then Lender shall have an additional sixty (60) day period to cure such failure and no default shall be deemed to exist hereunder so long as Lender commences such cure within the initial thirty (30) day period and diligently and in good faith pursues such cure to completion within such resulting ninety (90) day period from the date of Borrower's notice. Borrower waives any claim, set-off or defense against Lender arising by reason of any alleged default by Lender as to which Borrower does not give such notice timely as aforesaid. Borrower acknowledges that such waiver is or may be essential to Lender's ability to enforce its remedies without delay and that such waiver therefore constitutes a substantial part of the bargain between Lender and Borrower with regard to the Loan.

# 21.12 Jurisdiction.

TO THE GREATEST EXTENT PERMITTED BY LAW, BORROWER HEREBY WAIVES ANY AND ALL RIGHTS TO REQUIRE MARSHALLING OF ASSETS BY LENDER. WITH RESPECT TO ANY SUIT, ACTION OR PROCEEDINGS RELATING TO THIS AGREEMENT (EACH, A "<u>PROCEEDING</u>"), BORROWER IRREVOCABLY (A) SUBMITS TO THE NON-EXCLUSIVE JURISDICTION OF THE STATE AND FEDERAL COURTS HAVING JURISDICTION IN THE COUNTY OF COOK AND STATE OF ILLINOIS, AND (B) WAIVES ANY OBJECTION THAT IT MAY HAVE AT ANY TIME TO THE LAYING OF VENUE OF ANY PROCEEDING BROUGHT IN ANY SUCH COURT, WAIVES ANY CLAIM THAT ANY PROCEEDING HAS BEEN BROUGHT IN AN INCONVENIENT FORUM AND FURTHER WAIVES THE RIGHT TO OBJECT, WITH RESPECT TO SUCH PROCEEDING, THAT SUCH COURT DOES NOT HAVE JURISDICTION OVER SUCH PARTY. NOTHING IN THIS AGREEMENT SHALL PRECLUDE LENDER FROM BRINGING A PROCEEDING IN ANY OTHER JURISDICTION NOR WILL THE BRINGING OF A PROCEEDING IN ANY ONE OR MORE JURISDICTIONS PRECLUDE THE BRINGING OF A PROCEEDING IN ANY OTHER JURISDICTION NOR WILL THE BRINGING OF A PROCEEDING IN ANY ONE OR MORE JURISDICTIONS PRECLUDE THE BRINGING OF A PROCEEDING IN ANY OTHER JURISDICTION TO ANY METHODS OF SERVICE OF PROCESS PROVIDED FOR UNDER APPLICABLE LAW, ALL SERVICE OF PROCESS IN ANY PROCEEDING IN ANY ILLINOIS STATE S COURT HAVING JURISDICTION OVER THE COUNTY OF COOK MAY BE MADE, TO THE EXTENT PERMITTED BY LAW, BY CERTIFIED OR REGISTERED MAIL, RETURN RECEIPT REQUESTED, DIRECTED TO BORROWER AT THE ADDRESS INDICATED BELOW, AND SERVICE SO MADE SHALL BE COMPLETE UPON RECEIPT; <u>EXCEPT</u> THAT IF BORROWER SHALL REFUSE TO ACCEPT DELIVERY (AS OPPOSED TO UNABLE TO RECEIVE DELIVERY), SERVICE SHALL BE DEEMED COMPLETE FIVE (5) DAYS AFTER THE SAME SHALL HAVE BEEN SO MAILED.

# 21.13 Set-Offs.

After the occurrence and during the continuance of an Event of Default, Borrower hereby irrevocably authorizes and directs Lender from time to time to charge Borrower's accounts and deposits with Lender (or its Affiliates), and to pay over to Lender an amount equal to any amounts from time to time due and payable to Lender hereunder, under the Note or under any other Loan Document. Borrower hereby grants to Lender a security interest in and to all such accounts and deposits maintained by Borrower with Lender (or its Affiliates).

# 21.14 Binding Effect.

The covenants, conditions, waivers, releases and agreements contained in this Agreement shall bind, and the benefits thereof shall inure to the parties hereto and their respective heirs, executors, administrators, successors and assigns.

# 21.15 Waiver of Accord and Satisfaction.

Borrower hereby expressly waives any and all rights to effect an accord and satisfaction of any secured obligation or any other debt of Borrower to Lender in accordance with section 3-311 of the UCC. Notwithstanding anything to the contrary contained in this agreement or any other loan document, except as expressly directed in a writing addressed to Borrower after the date hereof, any and all communications or notices by Borrower or any other loan party to Lender concerning disputed debts, obligations or liabilities, whether arising under this agreement or otherwise, including without limitation any instrument tendered as full satisfaction of a debt, shall be, in addition to the notices required under <u>Article 22</u> hereof, delivered to Corus Bank, Department 311, attention Rosa Paz, 3959 North Lincoln Avenue, Chicago, Illinois 60613.

# Article 22

## NOTICES

Any notice, demand, request or other communication that any party hereto may be required or may desire to give hereunder shall be in writing and shall be deemed to have been properly given (a) if hand delivered, when delivered; (b) if by Federal Express or other reliable overnight courier service, on the next Business Day after delivered to such courier service or (c) if by telecopier on the day of transmission so long as copy is sent on the same day by overnight courier as set forth below:

	If to Borrower:
	c/o Stratus Properties
	98 San Jacinto, Suite 220
	Austin, Texas 78701
Attention:	W.H. Armstrong III
Telephone:	(512) 478-6396
Facsimile:	(512) 478-5788
	And to:
	Armbrust & Brown, L.L.P.
	100 Congress Avenue, Suite 1300
	Austin, TX 78701
Attention:	Kenneth N. Jones, Esq.
Telephone:	(512) 435-2312
Facsimile:	(512) 435-2360

# And to: Canyon-Johnson Urban Fund II, L.P. 9665 Wilshire Boulevard, Suite 200 Beverly Hills, California 90212 K. Robert Turner Attention: Telephone: (310) 247-2700 Facsimile: (310) 247-8067 And to: Sidley Austin LLP 555 West Fifth Street, 40th Floor Los Angeles, California 90013 Attention: Brian C. Flavell, Esq. Telephone: (213) 896-6603 Facsimile: (213) 896-6600 If to Lender: COR<u>US</u> Bank, N.A. 3959 North Lincoln Avenue Chicago, Illinois 60613 Attention: Brian Brodeur Telephone: (773) 832-3452 Facsimile: (773) 832-3540 With a copy to: COR<u>US</u> Bank, N.A. 3959 North Lincoln Avenue Chicago, Illinois 60613 Joel Solomon, General Counsel Attention: Telephone: (773) 832-3526 Facsimile: (773) 832-3626

	And to:
	Katten Muchin Rosenman LLP
	525 West Monroe Street
	Chicago, Illinois 60661
Attention:	Mark C. Simon, Esq.
Telephone:	(312) 902-5301
Facsimile:	(312) 577-4517

or at such other address as the party to be served with notice may have furnished in writing to the party seeking or desiring to serve notice as a place for the service of notice.

Article 23

## WAIVER OF JURY TRIAL

BORROWER AND LENDER EACH WAIVE ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING TO ENFORCE OR DEFEND ANY RIGHTS UNDER THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS OR RELATING THERETO OR ARISING FROM THE LENDING RELATIONSHIP THAT IS THE SUBJECT OF THIS AGREEMENT AND AGREE THAT ANY SUCH ACTION OR PROCEEDING SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.

EXECUTED as of the date first set forth above.

# **BORROWER:**

# CJUF II STRATUS BLOCK 21 LLC, a Delaware limited liability company

By: Stratus Block 21 Investments, L.P., a Texas limited partnership, Manager

By: Stratus Block 21 Investments GP,

L.L.C., a Texas limited liability company, General Partner

By:/s/ John E. Baker Name: John E. Baker Title: Senior Vice President

By: Canyon-Johnson Urban Fund II, L.P., aDelaware limited partnership, Member

By: Canyon-Johnson Realty Advisors IILLC, a Delaware limited liabilitycompany, General Partner

By: <u>/s/ K. Robert Turner</u> Name:K. Robert Turner Title: Managing Partner

# COR<u>US</u> BANK, N.A.

By: <u>/s/ Brian Brodeur</u> Name: Brian Brodeur Title: Senior Vice President

# EXHIBIT A

Legal Description of Land

Lots 1 through 12, Block 21, of the Original City of Austin, Travis County, Texas, according to the map or plat of record in the General Land Office of the State of Texas, together with the area within the alley traversing said Block, which was vacated by Ordinance recorded under Document No. 1999086902 and described in Memorandum Designating the Vacation of a 20 foot wide alley on Block 21 and Block 22, in the City of Austin as recorded under Document No. 2004040650 of the Official Public Records of Travis County, Texas.

# EXHIBIT B

Permitted Exceptions

# EXHIBIT C

**Title Requirements** 

- 1. <u>Title Insurance Company Requirements</u>. The maximum single risk (i.e., the amount insured under any one policy) by a title insurer may not exceed 25% of that insurer's surplus and statutory reserves. Reinsurance must be obtained by closing for any policy exceeding such amount.
- 2. Loan Policy Forms. Standard 2006 American Land Title Association ("ALTA") form of loan title insurance policy, the 1970 (amended October 17, 1970) ALTA loan, or an equivalent Texas form policies must be used.
- 3. <u>Insurance Amount</u>. The amount insured must equal at least the original principal amount of the Loan.
- 4. <u>Named Insured</u>. The named insured under the Title Policy must be substantially the same as the following: "Cor<u>us</u> Bank, N.A. and its respective successors and assigns."
- <u>Creditors' Rights</u>. Any "creditors' rights" exception or other exclusion from coverage for voidable transactions under bankruptcy, fraudulent conveyance, or other debtor protection laws or equitable principles must be removed by either an endorsement or a written waiver or issuance of a 1970 Form Policy with 1982 modifications.
- 6. <u>Arbitration</u>. In the event that the form policy which is utilized includes a compulsory arbitration provision, the insurer must agree, if permitted by law, that such compulsory arbitration provisions do not apply to any claims by or on behalf of the insured. Please note that the 1987 and 1992 ALTA form loan policies include such provisions.
- 7. <u>Date of Policy</u>. The effective date of the Title Policy must be as of the date and time of the closing.
- 8. <u>Legal Description</u>. The legal description of the property contained in the Title Policy must conform to (a) the legal description shown on the survey of the property, and (b) the legal description contained in the Deed of Trust. In any event, the Title Policy must be endorsed to provide that the insured legal description is the same as that shown on the survey.
- 9. <u>Easements</u>. Each Title Policy shall insure, as separate parcels: (a) all appurtenant easements and other estates benefiting the property, and (b) all other rights, title, and interests of the borrower in real property under reciprocal easement agreements, access agreements, operating agreements, and agreements containing covenants, conditions, and restrictions relating to the Project.
- 10. <u>Exceptions to Coverage</u>. With respect to the exceptions, the following applies:
  - a) Each Title Policy shall afford the broadest coverage available in the state in which the subject property is located.

# C-i

b) The "standard" exceptions (such as for parties in possession or other matters not shown on public records) must be deleted.

c) The "standard" exception regarding tenants in possession under residential leases, should also be deleted. For commercial properties, a rent roll should be attached in lieu of the general exception.

d) The standard survey exception to the Title Policy must be deleted. Instead, a survey reading reflecting the current survey should be incorporated.

e) Any exception for taxes, assessments, or other lienable items must expressly insure that such taxes, assessments, or other items are not yet due and payable.

f) Any lien, encumbrance, condition, restriction, or easement of record must be listed in the Title Policy, and, if permitted by law, the Title Policy must affirmatively insure that the improvements do not encroach upon the insured easements or insure against all loss or damage due to such encroachment.

g) The Title Policy may not contain any exception for any filed or unfiled mechanic's or materialmen's liens.

h) In the event that a comprehensive endorsement has been issued and any Schedule B exceptions continue to be excluded from the coverage provided through that endorsement, then a determination must be made whether such exceptions would be acceptable to Lender. In the event that it is determined that such exception is acceptable, a written explanation regarding the acceptability must be submitted as part of the delivery of the loan documents.

(i) No exception(s) relating to the use of a purchaser's upgrade deposits or earnest money deposits in the construction of the Project shall be permitted.

If Schedule B indicates the presence of any easements that are not located on the survey, the Title Policy, if permitted by law, must provide affirmative insurance against any loss resulting from the exercise by the holder of such easement of its right to use or maintain that easement. ALTA Form 103.1 or an equivalent endorsement is required for this purpose.

11. <u>Endorsements</u>. With respect to endorsements, the following applies:

a) Each Title Policy must include an acceptable environmental protection lien endorsement on ALTA Form 8.1. Please note that Form 8.1 may take exception for an entire statute which contains one or more specific sections under which environmental protection liens could take priority over the Deed of Trust, provided, however, that such specific sections under which the lien could arise must also be referenced.

C-ii

b) Each Title Policy must contain an endorsement which provides that the insured legal description is the same as shown on the survey.

c) Each Title Policy must contain a comprehensive endorsement (ALTA Form 9) if a lien, encumbrance, condition, restriction, or easement is listed in Schedule B to the title insurance policy.

d) Lender may require the following endorsements where applicable and available:

-access -address -adjustable rate -assessments -assignment of leases and rents -assignment of loan documents -condominium endorsement -contiguity -doing business -due execution -first loss -last dollar -leasehold -mineral rights -Deed of Trust tax -reverter -revolving credit -single tax lot -subdivision -tie in -usury -zoning (ALTA 3.1 -with parking)

- 12. <u>Other Coverages</u>. If permitted by applicable law, each Title Policy shall insure the following by endorsement or affirmative insurance to the extent such coverage is not afforded by the ALTA Form 9 or its equivalent in a particular jurisdiction:
  - a) that no conditions, covenants, or restrictions of record affecting the property:
    - (i) have been violated,
    - (ii) create lien rights which prime the insured Deed of Trust,
    - (iii) contain a right of reverter or forfeiture, a right of reentry, or power of termination, or
    - (iv) if violated in the future would result in the lien created by the insured Deed of Trust or title to the property being lost, forfeited, or subordinated; and

b) that except for temporary interference resulting solely from maintenance, repair, replacement, or alteration of lines, facilities, or equipment located in easements and rights of way taken as certain exceptions to each Title Policy, such exceptions do not and shall not prevent the use and operation of the Project or the improvements as used and operated on the effective date of the Title Policy.

13. <u>Informational Matters</u>. The Policy must include, as an informational note, the following:

a) The recorded plat number together with recording information; and

- b) The property parcel number or the tax identification number, as applicable.
- 14. <u>Delivery of Copies</u>. Legible copies of all easements, encumbrances, or other restrictions shown as exceptions on the Title Policy must be delivered with the first draft of the title commitment.

Form of Survey Certification

# **CERTIFICATION FOR SURVEYS (LONG-FORM)**

I hereby certify to CORUS Bank, N.A., its successors and assigns, and (insert Borrower and Title Insurance Company) that the survey prepared by me entitled " " was actually made upon the ground and that it and the information, courses and distances shown thereon are correct; that the title lines and lines of actual possession are the same; that the size, location and type of buildings and improvements are as shown and all are within the boundary lines of the property; that there are no violations of zoning ordinances, restrictions or other rules and regulations with reference to the location of said building and improvements; that there are no easements, encroachments or use affecting this property appearing from a careful physical inspection of the same, other than those shown and depicted thereon; that all utility services required for the operations of the premises either enter the premises through adjoining public streets, or the survey shows the point of entry and location of any utilities which pass through or are located on adjoining private land; that the survey shows the location and direction of all storm drainage systems for the collection and disposal of all roof and surface drainage; that any discharge into streams, rivers or other conveyance system is shown on the survey; and that the parcels described heron do not lie within [or lie within] flood hazard areas in accordance with the document entitled "Department of Housing and Urban Development, Federal Insurance Administration-Special Flood Hazard Area Maps." This survey is made in accordance with the "Minimum Standard Detail Requirements for Land Title Surveys" jointly established and adopted by ALTA and ACSM in 2005 and includes items 1-4, 6-7(b)(i), 8-10, 11(b) and 13 of Table A. Pursuant to the Accuracy Standards as adopted by ALTA, NSPS, and ACSM and in effect on the date of this certification, the undersigned further certifies that: [Surveyor to complete with appropriate choice from Minimum Standard Detail Requirement]

D-i

#### EXHIBIT E

# INSURANCE REQUIREMENTS DURING AND POST CONSTRUCTION

I. <u>Insurance</u>. Borrower shall obtain, and maintain at all times during term of the Loan, such insurance as Corus Bank, N.A. ("Lender") may reasonably require, including, but not limited to the insurance coverage set forth below. Unless otherwise expressly defined herein, capitalized terms set forth in this Exhibit are terms of art, as used in and understood in the insurance industry or are defined terms in the Construction Loan Agreement to which this is attached.

# A. <u>During Construction</u>.

(a) <u>Builder's Risk</u>. From the closing of the loan until replaced by permanent property insurance, "All Risk" form of Builder's Risk Insurance, in such amount as Lender shall reasonably require, but in no event less than 100% of the replacement cost value of the Project (including Upgrades and any leasehold improvements) (the **"Builder's Risk Insurance"**). Such policy shall be written on a Builder's Risk Completed Value Form (100% non-reporting) or its equivalent and shall not contain a permission to occupy limitation or a coinsurance clause. Such policy shall not have exclusion for sidewalks, retaining walls or underground property. The policy shall include coverage for Flood and Earthquake with sub-limits no less than 25% of the replacement cost, but each not less than \$5,000,000 per occurrence and in the annual aggregate, unless building or Project is located in Earthquake Mercalli Zone VII or greater, or a Flood Zone "A", as defined by the National Insurance Flood Plan. If the Project is located in an Earthquake Mercalli Zone VII or greater, Borrower shall maintain Earthquake limits providing for 75% of the replacement cost, with a deductible not greater than 5% of replacement cost. If the Project is located in a Flood Zone "A", Borrower shall maintain Flood limits providing for 30% of the replacement cost (or such other limits approved by Lender), with a deductible not greater than 3% of replacement costs. Such insurance policy shall also include coverage for:

- (i) Loss suffered with respect to Borrower's materials, equipment, machinery, and supplies whether on-site, in transit, or stored off site, with a limit of no less than 100% replacement cost and subject to a minimum limit of \$250,000 for loans less than \$25 Million, \$500,000 for loans \$25 Million up to \$50 Million, and \$1,000,000 for loans \$50 Million or more for both transit and off site storage provided that Borrower shall obtain or cause to be obtained additional insurance whenever the value of materials in transit or storage exceed those limits;
- (ii) At least 25% of the Soft Costs contained in the Approved Budget, and including coverage for all types (including but not limited to interest expense; fees; and plans, specifications, blueprints and

E-i

models, in connection with any restoration following an insured loss);

(iii) If applicable, loss or delay of rental income for a period of at least 6 months on an actual loss sustained basis. Lender reserves the right to require an endorsement providing for an extended period of indemnity for the rental income insurance.

(b) <u>Comprehensive Broad Form Boiler and Machinery Insurance</u>, covering all mechanical and electrical apparatus and pressure vessels. Such insurance shall provide coverage against loss or damage from an accident to and/or caused by boilers and machinery, including but not limited to: heating apparatus, pressure vessels, pressure pipes, electrical or air conditioning equipment on a blanket comprehensive coverage form, in such amount as Lender shall reasonably approve but no less than \$10,000,000. All exclusions for testing shall be removed.

(c) <u>Professional Liability</u>. Borrower will require the architect, engineers (including Structural and MEP contractors) and all other design professionals retained by Borrower to purchase and maintain continuous professional liability coverage in the amount of \$1,000,000 per claim / and \$1,000,000 annual aggregate. This policy may be on a "claims made" basis, and shall include coverage for bodily injury and property damage and retroactive coverage back to the first date that professional services were provided to the Project. Evidence of this insurance shall be provided to Borrower in form of insurance certificate for a period of 3 years after substantial completion of the Project (unless the loan is paid off).

(d) <u>Commercial General Liability and Umbrella Liability</u> coverage, including but not limited to, coverage for Personal Injury, Bodily Injury, Death, Property Damage, with limits of not less than \$10,000,000 per occurrence and in the annual aggregate <u>per</u><u>location</u>. If commercially available in the location of the Project, the policy shall contain an Extended Products/Completed Operations Hazard Endorsement ("Extended Products Endorsement") for a period of at least three years or the state requirement (if any), whichever is greater, acceptable to Lender in its reasonable discretion. The policies described in this paragraph shall cover, without limitation: elevators, escalators, independent contractors, contractual liability (covering, to the maximum extent permitted by the commercial general liability policy, Borrower's obligation to indemnify Lender as required under this Exhibit) and Products and Completed Operations Liability coverage. Coverage should also include host liquor liability. Borrower shall add Lender, its directors, officers, employees and agents as additional insured.

(e) <u>Worker's Compensation</u>. If applicable, worker's compensation insurance covering Borrower and its employees at the site to the extent required, and in the amounts required by applicable Laws. An endorsement providing U.S.

E-ii

Longshore and Harbor Workers Compensation Act (USL&HW) coverage should be added on an "if any" basis if there is an exposure.

(f) <u>Employers Liability</u>. If applicable in the amount of \$1,000,000 per accident, \$1,000,000 per illness, per employee and \$1,000,000 per illness, in the aggregate.

(g) <u>Contractor's Liability</u>. Borrower shall cause the Contractor to maintain Commercial General Liability coverage, including, without limitation, products and completed operations insurance with no less than \$10,000,000 in limits per occurrence and in the aggregate **per project** through primary and umbrella liability policies. For all condominium projects if commercially available in location of the Project the Contractor shall be insured by the Extended Products Endorsement or by comparable coverage. Borrower shall also ensure that all subcontractors maintain similar coverage with limits appropriate to the hazard associated with their respective work on the site. Contractor shall be required to carry automobile liability insurance for all owned, hired and non owned vehicles with limits of at least \$10 Million. All parties engaged in work on the Project shall maintain statutory Workers Compensation, Employer's Liability with limits of at least \$1,000,000 and any legally mandated Disability insurance in force for all employees on the job. It is understood that umbrella or excess policies may be used to reach the required limits.

(h) Borrower may provide the Commercial General and Umbrella Liability and Workers Compensation and Employers Liability required to be carried pursuant to Sections I.A., Subsections (d), (e) and (f) of this Exhibit through the purchase of a Wrap-up or Owner Controlled Insurance Program or a Contractor Controlled Insurance Program. This program shall provide coverage for all parties engaged in construction operations at the Project with limits approved by Lender.

- (i) Surety bonds as required by the terms of the Application Letter/Commitment Letter.
- (j) Such other insurance reasonably required by Lender.

**B.** <u>Post Construction</u>. After the earlier of: (i) substantial completion of the Project, or (ii) cancellation or expiration of the Builder's Risk Policy, Borrower shall provide the following coverages:

(a) <u>"All Risk" insurance</u> including Flood and Earthquake, and such other insurable hazards as, under good insurance practices are insured against for other property and buildings similar to the premises in nature, use, location, height, and type of construction. The amount of such insurance shall be not less than one hundred percent (100%) of the replacement cost without depreciation of the Project. Such insurance policy shall contain an agreed amount endorsement.

E-iii

Flood and Earthquake sublimits shall be at least 25% of the replacement cost, but not less than \$5,000,000 each per occurrence and in the annual aggregate, unless the risk is location in a Mercalli Zone VII or greater or a Flood Zone "A", as defined by the National Insurance Flood Plan. If the Project is located in an Earthquake Mercalli Zone VII or greater Borrower shall maintain Earthquake limits providing for 75% of the replacement cost, with a deductible not greater than 5% of the replacement cost. If the Project is located in a Flood Zone "A", Borrower shall maintain Flood limits providing for 30% of the replacement cost (or such other limits approved by Lender), with a deductible not greater than 3% of replacement costs. Such insurance shall cover increased cost of law or ordinance insurance, costs of demolition and increased cost of construction with a sublimit of not less than \$1,000,000. If coverage is provided under a blanket policy, Lender shall be named as sole Loss Payee and Mortgagee for the Project.

(b) <u>Comprehensive Broad Form Boiler and Machinery</u>. Insurance, in the minimum amount of \$10,000,000 covering all mechanical and electrical equipment against physical damage and covering, without limitation, all tenant improvements and betterments that Borrower is required to insure pursuant to any lease on a replacement cost basis. Such insurance shall provide coverage against loss or damage from an accident to and/or caused by boilers and machinery, including but not limited to: heating apparatus, pressure vessels, pressure pipes, and electrical or air conditioning equipment on a blanket comprehensive coverage form, in such amount Lender shall reasonably approve. All exclusions for testing shall be removed. Coverage shall be extended to include loss of rental income for 6 months as a result of damage from an insured peril.

(c) <u>Business Interruption</u>. Upon Lender's reasonable request, business income and extra expense, against the perils insured by the "All Risk" Property, for a period of indemnity of twelve months.

(d) <u>Commercial General Liability</u>. Commercial General Liability Insurance, for Personal Injury, Bodily Injury, Death, Accident and Property Damage providing in combination no less than \$10,000,000 per occurrence (with sublimits approved by Lender) and in the annual aggregate, **per location**. The policies described in this paragraph shall cover, without limitation: elevators, escalators, independent contractors, contractual liability and Products and Completed Operations Liability coverage. It is understood that umbrella or excess policies can be used to meet the required limits.

(e) <u>Dram Shop</u>. Upon Lender's reasonable request or prior to any tenant selling alcoholic beverages on any part of the Project, Borrower either itself or through the Tenant shall provide evidence of so-called "Dram Shop" against claims or liabilities arising directly or indirectly to Persons or property on account of the sale or dispensing of alcoholic beverages. Coverage shall include loss of means of support. Limits shall equal those limits as may be required by

E-iv

applicable Laws or as Lender may reasonably specify. If state law allows, Lender shall be named as an additional insured on such policy.

(f) <u>Worker's Compensation</u>. If applicable, worker's compensation insurance covering Borrower and its employees at the site to the extent required, and in the amounts required by applicable Laws.

(g) Employers Liability. If applicable in the amount of \$1,000,000 per accident; \$1,000,000 per illness, per employee; and \$1,000,000 per illness, in the aggregate.

- (h) <u>Auto Liability for Owned (if any) and Hired and Non Owned with limits approved by Lender.</u>
- (i) <u>Other</u>. Such other insurances as may be reasonably requested by Lender.

### II. <u>Requirements of Insurance Policies</u>.

(a) All insurance policies shall be issued by an insurer or insurers with an A.M. Best rating of A:IX or better or a Standard and Poor's rating of "AA", or equivalent rating from another agency acceptable to Lender and be authorized in the state where the Project is located. All insurance acquired pursuant to this Exhibit shall be in form, amounts and with coverage and deductibles satisfactory to Lender, in Lender's sole discretion.

(b) The Builder's Risk insurance policies required to be carried pursuant to Section I.A., Subsections (a) and (b) of this Exhibit, and the All Risk required pursuant to Section I.B, Subsections (a) and (b), shall name Borrower as the insured and shall also name Lender as Loss Payee and Mortgagee, under a non-contributing standard mortgagee clause. Without Lender's prior written consent, Borrower shall not name any Person other than Lender, as loss payee under any property insurance policies that Borrower is required to insure pursuant to any Lease.

(c) The Commercial General Liability and Contractors Liability set forth in Section I.A., Subsections (d) and (g) and Section I.B., Subsection (d), shall name Lender, its directors, officers, and employees as Additional Insured.

(d) Lender shall be named as a dual-obligee on the Performance Bond required to be obtained by General Contractor pursuant to Section I.A., Subsection (i) of this Exhibit.

(e) The amount of any deductible under any insurance policy must be reasonably acceptable to Lender.

(f) Borrower may provide required insurance under blanket policies. Borrower shall not maintain any insurance on the Project that does not name Lender as Loss Payee.

(g) Borrower shall pay the premiums for the insurance policies as the same become due and payable. Borrower shall deliver to Lender certified copies of the insurance policies required to be maintained pursuant to this Exhibit within sixty (60) days after the date of this Agreement or ten (10) days after the issuance of the policies by the insurer, whichever is later, but in all events, no later than ninety (90) days after the date of this Agreement, and failure to do so will be an immediate Event of Default. Notwithstanding the foregoing, Lender shall not be deemed by reason of the custody of such insurance policies to have knowledge of the contents thereof. Borrower also shall deliver to Lender, within ten (10) days of Lender's request, a certificate of Borrower or Borrower's insurance agent setting forth the particulars as to all such insurance policies, that all premiums due thereon have been paid currently and that the same are in full force and effect. BORROWER SHALL DELIVER A CERTIFICATE OR OTHER EVIDENCE OF INSURANCE ACCEPTABLE TO LENDER EVIDENCING THE INSURANCE REOUIRED HEREUNDER ON THE CLOSING DATE. TOGETHER WITH RECEIPTS FOR THE PAYMENT OF PREMIUMS THEREON. ALL CERTIFICATES FOR PROPERTY INSURANCE MUST BE ON ACORD FORM 27 or the equivalent; ACORD 25 certificates are acceptable for liability insurance. Not later than fifteen (15) days prior to the expiration date of each of the insurance policies Borrower shall deliver to Lender a certificate of insurance evidencing renewal of coverage as required herein. Within ten (10) days after such renewal, Borrower shall deliver to Lender evidence of payment of premium satisfactory to Lender. Not later than ninety (90) days after the renewal of each of the insurance policies, Borrower shall deliver to Lender an original or certified copy (as required pursuant to this Section) of a renewal policy or policies.

(h) Each insurance policy shall contain a provision whereby the insurer agrees that so long as the Loan is outstanding, such policy shall not be canceled or fail to be renewed, lapsed or materially changed without in each case, at least thirty (30) days prior written notice to Lender, except ten (10) days for non-payment of premium

(i) In the event any insurance policy (except for general and other liability and Workers Compensation insurance) shall contain breach of warranty provisions, such policy shall provide that with respect to the interest of Lender, such insurance policy shall not be invalidated by and shall insure Lender regardless of; (A) any act, failure to act or negligence of or violation of warranties, declarations or conditions contained in such policy by any named insured; (B) the occupancy or use of the property for purposes more hazardous than permitted by the terms thereof; or (C) any foreclosure or other action or proceeding taken by Lender pursuant to any provision of this Agreement.

E-vi

(j) Any insurance maintained pursuant to this Agreement may be evidenced by blanket insurance policies covering the premises and other properties or assets of Borrower or its affiliates; provided that any such policy shall in all other respects comply with the requirements of this section. Lender, in its reasonable discretion, shall determine whether such blanket policies contain sufficient limits of insurance.

(k) Any insurance carried by Lender shall be for its sole benefit and shall not inure to the benefit of Borrower and Insurance required from Borrower shall be primary to any available, if any, to Lender.

(l) All required policies, other than professional liability, shall provide that insurers have waived rights of subrogation against Lender. The required insurance shall be primary without right of contribution from any insurance, which may be carried by Lender.

(m) The required limits are minimum limits established by Lender and nothing contained herein shall be construed to mean the required limits are adequate or appropriate to protect Borrower from greater loss.

If there are any questions regarding our insurance requirements, please contact our insurance consultants:

Alpha Risk Management, Inc., 60 Cutter Mill Road, Great Neck, NY 11021 Attention: Heidi M. Bucher, ARM hbucher@alphariskmanagement.com Tel: (516) 829-3500 x6 or Fax: (516) 829-6029

Final insurance policies should be forwarded for review directly to the attention of Heidi M. Bucher at Alpha Risk Management, Inc. Renewals should be sent to the attention of Josefina Reyes, Corus Bank, 3959 N. Lincoln, Chicago, Illinois 60613



### **Illinois Collateral Protection Act Notice**

Unless Borrower provides Lender with evidence of the insurance coverage required by this Agreement, Lender may purchase insurance at Borrower's expense to protect Lender's interest in the Project. This insurance may, but need not, protect Borrower's interest. The coverage that Lender purchases may not pay any claim that Borrower makes or any claim that is made against Borrower in connection with the Project. Borrower may later cancel any insurance purchased by Lender, but only after providing Lender with evidence that Borrower has obtained insurance as required by this Agreement. If Lender purchases insurance for the Project, Borrower will be responsible for the cost of that insurance, including interest and any other charges Lender may impose in connection with the placement of the insurance, until the effective date of the cancellation or expiration of the insurance. The cost of the insurance may be added to Borrower's total outstanding balance or obligation. The cost of insurance may be more than the cost of insurance Borrower may be able to obtain on its own.

-		٠	
E-'	V1	1	1

### <u>EXHIBIT F</u>

Architect's Certificate

\_\_\_\_\_ hereby certifies for the benefit of COR<u>US</u> Bank, N.A. that: The firm of The firm has been employed by \_\_\_\_\_\_ pursuant to a contract dated services commonly known as \_\_\_\_\_\_ which is located at \_\_\_\_\_\_. \_ pursuant to a contract dated \_\_\_\_\_\_ to provide architectural and engineering The contract provides for the following services: preparation of plans and specifications Pre-qualification of contractors Contract administration and supervision of construction Tenant space design The firm is duly licensed and in good standing under laws of the state of \_\_\_\_\_\_ License No. by\_ . The following are all of the permits or governmental agency approvals required for the construction and occupancy of the building: Issuing Agency Date Issued **Excavation Permit** Foundation Permit **Building Permit** EPA – Water EPA – Sewer

Cert. Of Occupancy Bldg. Cert. Of Occupancy – Tenant Other (Specify)

EPA – Air

F-i

All utilities necessary for the operation of the project are available with sufficient capacity at the boundaries of the project. If utility services must be brought to site, please explain:.

The plans listed on the attached Schedule I comprise all of the plans which will be necessary for the complete construction of the project, excepting tenant space designs, and when the project is built in accordance therewith the project will (excepting completion of tenant improvements) be ready for occupancy. The plans are complete and contain all detail necessary for construction. Calculations of the gross building and the net rentable building area are attached as Schedule II. The plans (and the project will, when constructed in accordance therewith) comply with all applicable building, zoning, land use, subdivision, environmental, fire, safety and other applicable governmental laws, statutes, codes, ordinances, rules and regulations.

The attached Schedule III, establishing a timetable for completion of the project and showing on a monthly basis the anticipated progress of the work, is realistic and can be adhered to.

The following design drawings or plans have been or will be prepared by other designers or contractors.

Type of Plans		Name of Preparing Firm	
The Specifications are:		shown on plans	
Bound sep	parately		
	By:		
	Title:		
	Date:		
		F-ii	

# EXHIBIT G

Initial Budget

# <u>EXHIBIT H</u>

Draw Request Form

# <u>EXHIBIT I</u>

Partial Plans and Specifications

(see attached)

## <u>EXHIBIT J</u>

Proposed Finish Standards

Proposed Construction Finish for Residential Units:

- 1. Building Foundations: CIP Concrete
- 2. Building Frame: CIP Concrete
- 3. Description of Exterior Wall System Including Design and Materials:
- Unitized Curtainwall System; Exterior Materials High Performance Vision Glass, Insulated Spandrel Glass, Composite Aluminum Panels
- 4. Description of Window System: See Above, Ext Wall System
- 5. Description of Common Area Amenities: Private Condo swimming pool, Terrace deck, access to all hotel amenities.
- 6. Ceiling Heights:
  - a. **Standard units:** Minimum ceiling: 8' in enclosed toilet rooms only, typically 9'-4" minimum, 10-6" max with curtain slots up to 11'-1".
  - b. Subpenthouses: 9'-0" minimum, 12'-6" max with curtain slots up to 15'-1".
  - c. Penthouse levels 35 and 36: 10-0" minimum, 14'-0" max with curtain slots up to 15'-1".
    - d. Penthouse level 37: 10'-0" minimum, 20'-0" max.

7. Description of Kitchen and Bath Millwork, Countertops, and Plumbing Fixtures: Bulthaup Kitchen Cabinets; Stone Countertops (selection by buyer); Plumbing Fixtures by Dombracht and Kohler

8. Flooring (Units and Common Areas): Mafi Hardwood Engineered Wood Floors, Stone Tile (selections vary), Carpet

- 9. Description of Appliances: Miele Brand Appliances
- 10. Other Features: n/a



### 1. Building Foundations: CIP Concrete

2. Building Frame: CIP Concrete

#### 3. Description of Exterior Wall System Including Design and Materials:

Unitized Curtainwall System; Exterior Materials - High Performance Vision Glass, Insulated Spandrel Glass, Composite Aluminum Panels

### 4. Description of Window System: See Above, Ext Wall System

5. Description of Common Area Amenities: Hotel pool, Terrace deck, ballroom, spa, fitness, lobby bar, restaurants, valet, shopping, dining, hike and bike trail.

### 6. Ceiling Heights:

- a. First Floor: 12'6" to 18'0" (bathrooms and secret bar at 9' 12')
- b. Second Floor: 12' to 16'
- c. Tower Floors: 8' to 9'3"

7. Description of Kitchen and Bath Millwork, Countertops, and Plumbing Fixtures: Built-in stained and lacquered closets and desk, Silestone (or similar) counters, Kohler plumbing fixtures

8. Flooring (Units and Common Areas): Carpet in Units, Concrete, Hardwood, rugs, porcelain tile, and carpet.

#### 9. Description of Appliances: n/a

10. Other Features: n/a

## J-ii

### EXHIBIT K

## PLEASE INCLUDE IN DRAW PACKAGE SENT TO CORUS BANK, N.A.

FORM OF BAILEE LETTER FOR PUBLIC WAREHOUSEMEN

[WAREHOUSEMEN LETTERHEAD]

[DATE]

Corus Bank, N.A.

3959 North Lincoln Avenue

Chicago, Illinois 60613

Attention: Corus Construction Manager

Re: W Hotel and Residences Condominiums

Austin, Texas

(the "Project")

Corus Construction Manager:

This letter is to confirm that the undersigned, **[Insert name of Public Warehousemen]** ("<u>Warehousemen</u>"), is holding and will hold, from time to time, for storage in its public warehouse located at the address set forth above, the materials described on <u>Schedule 1</u>, attached hereto and made a part hereof, for the Project (the "<u>Stored Materials</u>"). Such <u>Schedule 1</u> may be revised and updated as additional materials are delivered to Warehousemen and stored in accordance with this letter.

Warehousemen is holding such Stored Materials on behalf of **[Insert name of Contractor]**, as the "<u>Contractor</u>" of the Project and [\_\_\_\_\_], as owner of the Project ("<u>Owner</u>") and the Owner owns and has title to the Stored Materials or will own and obtain title to the Stored Materials upon receipt of a bill of sale for such Stored Materials. Pursuant to a certain Construction Loan Agreement (the "<u>Loan Agreement</u>") between Owner and Corus Bank, N.A. ("<u>Lender</u>"), Owner has granted to Lender a security interest in, among other things, the Stored Materials.

We acknowledge and agree that Lender's security interest in the Stored Materials is senior to all liens, claims and interests, other than our warehouseman's lien for any accrued and unpaid warehousing fees charged by us for the actual storage of the Stored Materials. To protect Lender's security interest in the Stored Materials, from and after the date of this letter, all warehouse receipts and other documents of title which evidence any Stored Materials now or hereafter delivered by Owner to us shall be non-negotiable and issued to or for the account of Lender. We shall provide Lender with a copy of such receipts or other documents upon Lender's request for those items. We will notify all of our successors and assigns of the existence of the agreements contained herein.

K-i

Notwithstanding the issuance of such receipts or other documents to or for the account of Lender, we acknowledge and agree that Lender has authorized us to release any of the Stored Materials to any authorized agent of Owner upon Owner's request (including, without limitation, Contractor); provided, however, upon our receipt of written direction from Lender, we shall refuse to release any of the Stored Materials to Owner or Owner's agent (including, without limitation, Contractor) and we shall only release such Stored Materials to Lender or the party designated by Lender in such written direction.

> Very truly yours, [PUBLIC WAREHOUSEMEN] By: Name: Title:

> > K-ii

## DESCRIPTION OF MATERIALS

[Warehousemen: Please insert/attach a detailed inventory of the Stored Materials, including, without limitation, a statement of the quantity stored, relevant invoices, warehouse receipts or other documents of title and any applications for payment.]

### EXHIBIT L

### PLEASE INCLUDE IN DRAW PACKAGE SENT TO CORUS BANK, N.A.

FORM OF BAILEE LETTER

#### [SUBCONTRACTOR LETTERHEAD]

[DATE]

Corus Bank, N.A.

3959 North Lincoln Avenue

Chicago, Illinois 60613

Attention: Corus Construction Manager

Re: W Hotel and Residences Condominiums

Austin, Texas

(the "Project")

### Corus Construction Manager:

This letter is to confirm that the undersigned, **[Insert name of Sub-Contractor]** ("<u>Bailee</u>"), is holding for processing and/or storage the materials described on <u>Schedule 1</u>, attached hereto and made a part hereof, for the Project (the "<u>Stored Materials</u>"). Such <u>Schedule 1</u> may be revised and updated as additional materials are delivered to Bailee and stored in accordance with this letter.

Bailee is holding such Stored Materials on behalf of **[Insert name of Contractor]**, as the "<u>Contractor</u>" of the Project and [\_\_\_\_\_], as owner of the Project ("<u>Owner</u>") and the Owner owns and has title to the Stored Materials or will own and obtain title to the Stored Materials upon receipt of a bill of sale for such Stored Materials. Pursuant to a certain Construction Loan Agreement (the "<u>Loan Agreement</u>") between Owner and Corus Bank, N.A. ("<u>Lender</u>"), Owner has granted to Lender a security interest in, among other things, the Stored Materials.

In order to protect Owner's ownership interest and Lender's security interest in the Stored Materials, Bailee agrees, acknowledges, represents and warrants as follows:

- (i) We are holding and will hold the Stored Materials on bailment for processing or warehousing;
- The Stored Materials are Owner's property and are subject to Lender's security interest, and we are holding and will hold the Stored Materials for Lender's benefit;

### L-i

- Lender's security interest in the Stored Materials shall be senior to all liens, claims and interests, including any fees charged by us for the actual processing or storage of the Stored Materials and we will notify all of our successors and assigns of the existence of the agreements contained herein;
- (iv) If, at any time after the date of this letter, Lender shall notify us in writing that Owner has defaulted on its obligations to Lender under the Loan Agreement, we will comply with Lender's written instructions as to the disposition of the Stored Materials; and
- (v) Until we are notified in writing by Lender that the financing arrangements under the Loan Agreement have been terminated and Lender has been paid in full, we shall not deduct from or offset against any amounts due and owing to us by Owner, by applying any of the Stored Materials in payment for such amounts.

Very truly yours, [BAILEE] By: Name: Title:

L-ii

## DESCRIPTION OF MATERIALS

[Bailee: Please insert/attach a detailed inventory of the Stored Materials, including, without limitation, a statement of the quantity stored, relevant invoices and any applications for payment.]

# EXHIBIT M

List of Sales Agreements and Price List Schedule

# EXHIBIT N

Subcontract Delivery Dates

# EXHIBIT O

Approved Form of Sales Agreement

# <u>EXHIBIT P</u>

Materials Purchases Not Subject to Retainage

# EXHIBIT Q

Allowable Tenant Improvements

# EXHIBIT R

Delivery Schedule for Plans and Specifications

# EXHIBIT S Form of Second Estoppel and Agreement from City of Austin

#### PROMISSORY NOTE

#### Project Commonly Known as

"W Hotel & Residences"

#### May 2, 2008 Chicago, Illinois

\$165,000,000

FOR VALUE RECEIVED, CJUF II STRATUS BLOCK 21 LLC, a Delaware limited liability company ("Maker"), with a mailing address of c/o Stratus Properties Inc., 98 San Jacinto, Suite 220, Austin, Texas 78701, promises to pay to the order of CORUS BANK, N.A., a national banking association ("Lender"), at the address set forth in Section 4.1(e) of the Loan Agreement (as defined in Section 7 below), or such other place as Lender may designate in writing, in the manner provided hereinafter the principal sum of up to One Hundred Sixty-Five Million Dollars (\$165,000,000), or so much thereof as may now or hereafter be disbursed by Lender to or for the benefit of Maker (the "Loan"), on or before the Maturity Date (as defined in the Loan Agreement) or the date to which the indebtedness evidenced hereby is accelerated pursuant to the terms of this Note and the terms of the other Loan Documents, with interest, in the manner and upon the terms and conditions set forth below. All capitalized terms not expressly defined herein shall have the same meanings as set forth in the Loan Agreement.

1. Interest Rate. Interest shall accrue from the date of first disbursement of the Loan on the principal balance thereof remaining from time to time outstanding at the rate ("Interest Rate") established below. Unless the Loan is then bearing interest at the Default Rate, the Interest Rate shall be equal to the greater of: (1) the sum of: (i) three and one half percent (3.5%) per annum plus (ii) the three month London Interbank Offered Rate quoted in the Money Rates section of <u>The Wall Street Journal</u> (the "LIBOR Rate"); or (2) six percent and one-half percent (6.5%) per annum. The LIBOR Rate shall be automatically adjusted without notice to Maker on the first (1st) day of each January, April, July, and October following the calendar month in which the Loan Opening Date occurs (the "Quarterly Adjustment Date") based on the LIBOR Rate as of the Quarterly Adjustment Date (or, if the Quarterly Adjustment Date, effective retroactively to the Quarterly Adjustment Date). Interest shall be computed based on a 360-day year and charged for the actual number of days elapsed.

2. <u>Principal and Interest Payments</u>. Commencing on the first day of the first month after the initial disbursement of the Loan, and continuing on the first day of each month thereafter until the Loan has been repaid in full, interest on the Loan shall be payable monthly in arrears on the first day of each month ("**Payment Date**") in the amount of all interest accrued and unpaid. The final payment of the entire unpaid principal balance of the Loan plus all accrued and unpaid interest, Exit Fees, charges, fees, and expenses, if not sooner paid, shall be due and payable on the Maturity Date.

3. <u>Prepayment</u>. During the twelve (12) months immediately following the execution of this Note ("Lockout Period"), Maker shall not be permitted to prepay the Loan in part or in full from any source. At Lender's option, any attempted unpermitted prepayment

during the Lockout Period shall be placed in a non-interest bearing account and held until such time as prepayment is allowed. After the Lockout Period, Maker may prepay the Loan in part or in full (in accordance with the provisions hereof) upon not less than seven (7) days prior written notice to Lender, except that no such notice shall be required in the case of the sale of individual Residential Units. In the event that Maker repays any portion of the Loan after the 12-month anniversary of this Note but before the 30-month anniversary of this Note from a source other than proceeds realized from the sale of individual Residential Units or of a component of the Project (i.e., the Venue, the Hotel or the Commercial Space), in each case, in accordance with and as permitted by the Loan Documents, then Maker shall pay Lender a prepayment charge (the "**Prepayment Charge**") equal to (x) in the event of a partial prepayment, two percent (2.0%) of the amount prepaid, or (y) in the event of prepayment in full, the sum of (i) two percent (2.0%) of the amount prepaid and (ii) two percent (2.0%) of any Unfunded Commitment. The Prepayment Charge shall be in addition to any Exit Fee still outstanding. There shall be no Prepayment Charge on casualty or condemnation proceeds which Lender requires be applied to the Loan.

4. <u>Applicability of Charges</u>. Upon acceleration of the Loan at any time after the date of this Note and prior to the 30-month anniversary of the date of this Note, in addition to the Exit Fee that may be due as provided in the Loan Agreement and to the extent permitted by law, Maker hereby agrees that there shall be added to the indebtedness evidenced hereby and secured by that certain Construction Deed of Trust, Assignment of Rents, Security Agreement and Fixture Filing of even date herewith (the "**Deed of Trust**") a charge equal to the Prepayment Charge (calculated as if the outstanding Loan were being prepaid in full as of the date of such acceleration, regardless of whether such acceleration would be deemed a "prepayment" under applicable Law). Maker hereby acknowledges that the aforesaid charge payable to the Lender herewinder is a reasonable pre-estimate of the Lender's losses and is not a penalty.

**5.** <u>Application of Payments</u>. Subject to the application of payments required by <u>Section 14.10</u> of the Loan Agreement and other applicable provisions of the Loan Documents, Lender shall have the right unilaterally (and without the consent of Maker) to allocate any and all payments that may be received by or tendered to Lender made by Maker or any other person at any time or from time to time and that relate in any way to the Loan or any other of the obligations then due and payable under the Loan Documents in any order of priority as Lender in its sole and exclusive discretion shall elect, including without limitation to: (i) to the payment of any costs and expenses of Lender that Maker is responsible for under the Loan Documents; (ii) to accrued but unpaid interest, Exit Fees, penalties, late payment fees; and (iii) to principal. Maker (1) irrevocably waives the right to direct the application of payments and collections received by Lender from or on behalf of Maker and/or any other person, and (2) agrees that Lender shall have the continuing exclusive right to apply and reapply any and all such payments and collections against the Loan or any other obligations then due and payable under the Loan Documents in such manner as Lender may deem appropriate (subject to any express provisions of the Loan Documents to the contrary), notwithstanding any entry by Lender upon any of its books and records.

6. Late Charges. If any installment of interest, principal or other amount due under the Loan Documents is not paid on or before the fifteenth (15th) day after the date such payment is due, Maker shall pay Lender a late charge in an amount equal to ten percent (10%) of the amount due on any payment (including, but not limited to, any payment of principal, interest,

unfunded loan fee, if applicable, or any other payment due other than the payment of the total principal amount due upon maturity of the Loan [whether at stated maturity or by acceleration]) to defray part of the increased cost of collecting the late payments and the opportunity cost incurred by Lender because of the unavailability of the funds.

7. Security for Payment. The payment of this Note is further evidenced by the Construction Loan Agreement of even date herewith between Maker and Lender (the "Loan Agreement"), and is secured, *inter alia*, by the Deed of Trust. All of the agreements, conditions, covenants, provisions and stipulations contained in the Loan Documents are hereby made a part of this Note to the same extent and with the same force and effect as if they were fully set forth herein, and Maker covenants and agrees to keep and perform them or cause them to be kept and performed, strictly in accordance with their terms.

8. <u>Defaults and Acceleration</u>. IT IS HEREBY EXPRESSLY AGREED BY MAKER THAT TIME IS OF THE ESSENCE HEREOF. At any time during the existence of any Event of Default, at the option of Lender, the entire unpaid principal balance of the Loan with interest accrued thereon and all other sums due by Maker hereunder or under the provisions of the other Loan Documents shall, upon demand (except as otherwise set forth in the Loan Agreement) become due and payable immediately.

9. Default Rate. After maturity, whether by acceleration or at stated maturity, or while any Event of Default exists, subject to the limitations set forth in Section 14 below, Maker promises to pay interest on the amount of principal due and outstanding hereunder and any fees, costs and advances at the rate of ten percent (10%) per annum plus the Interest Rate that would otherwise be then in effect, and shall be adjusted on each Quarterly Adjustment Date thereafter (the "Default Rate"), but in no event will the Default Rate exceed the applicable maximum rate of interest allowed by law, and shall be payable upon demand, and all unpaid interest that has accrued under this Note, whether prior or subsequent to the occurrence of the Event of Default, shall be paid at the time of, and as a condition precedent to, the curing of the Event of Default. The Default Rate, when operative, shall be adjusted based on changes in the Interest Rate, but in no event will the Default Rate exceed the applicable maximum rate of interest allowed by law.

10. <u>Nature of Remedies</u>. The remedies of Lender as provided in this Note, the Deed of Trust and any of the other Loan Documents shall be cumulative and concurrent, and may be pursued singly, successively, or together against either or any of Maker, the mortgaged property, any guarantor hereof or any other security, at the sole and absolute discretion of Lender.

11. <u>Waivers, Consents, Etc.</u>. Maker and any endorsers, sureties or guarantors hereof and any and all others who are now or may become liable for all or part of the obligations of Maker under this Note (all of the foregoing being collectively referred to herein as "**Obligors**"), agree to be jointly and severally bound hereby and jointly and severally waive presentment for payment, demand, notice of nonpayment, notice of dishonor, protest of any dishonor, notice of protest, and protest of this Note and all other notices in connection with the delivery, acceptance, performance, default, or enforcement of the payment of this Note and, except as expressly provided to the contrary herein, in the other Loan Documents or by Laws, agree that the liability of each of them shall be unconditional, joint and several, without regard to the liability of any other party and shall not in any manner be affected by any indulgence, extension of time, renewal, waiver or modification granted or consented to by Lender. Each Obligor hereby consents to any and all extensions of time, renewals, waivers or modifications that may be

granted by Lender with respect to the payment or other provisions of this Note, and to the release of the collateral, or any part thereof, with or without substitution, and agree that additional makers, endorsers, guarantors, or sureties may become parties hereto without notice to them or affecting their liability hereunder. Maker hereby authorizes Lender, at any time after, and during the existence of, an Event of Default hereunder, to apply any money, or other property which Lender may have or hold on deposit or otherwise for Maker towards the payment of this Note. By execution hereof, Maker hereby specifically pledges and grants to Lender a security interest in any money or other property which Lender may have or hold on deposit for the undersigned.

12. <u>Non-Waiver</u>. Lender shall not by any act of omission or commission be deemed to waive any of its rights or remedies hereunder unless such waiver is in writing and signed by Lender and then only to the extent specifically set forth therein. A waiver on one event shall not be construed as continuing or as a bar to or waiver of such right or remedy in connection with a subsequent event.

13. <u>Business Loan</u>. Maker warrants and represents to Lender that Maker shall use the proceeds represented by this Note solely for proper business purposes, and consistently with all applicable Laws and statutes. Maker further warrants and represents to Lender and covenants with Lender that Maker is not in the business of extending credit for the purpose of purchasing or carrying margin securities (within the meaning of Regulation U issued by the Board of Governors of the Federal Reserve System), and no proceeds represented by this Note will be used to purchase or carry any margin securities or to extend credit to others for the purpose of purchasing or carrying any margin securities.

14. Interest Laws. It being the intention of Lender and Maker to comply with applicable Laws, it is agreed that notwithstanding any provision to the contrary in this Note, the Deed of Trust or any of the other Loan Documents, no such provision shall require the payment or permit the collection of any interest ("Excess Interest") in excess of the maximum amount of interest permitted by law to be charged for the use or detention, or the forbearance in the collection, of all or any portion of the indebtedness evidenced by this Note. If any Excess Interest is provided for, or is adjudicated to be provided for, in this Note, the Deed of Trust or any of the other Loan Documents, then in such event: (a) the provisions of this paragraph shall govern and control; (b) neither Maker nor any of the other Obligors shall be obligated to pay any Excess Interest; (c) any Excess Interest that Lender may have received hereunder shall, at the option of Lender, be (i) applied as a credit against the then outstanding principal balance of the Loan, accrued and unpaid interest thereon not to exceed the maximum amount permitted by law, or both, (ii) refunded to the payor thereof, or (iii) any combination of the foregoing; (d) the applicable interest rate or rates hereunder shall be automatically subject to reduction to the maximum lawful contract rate allowed under the applicable usury laws of the aforesaid State, and this Note, the Deed of Trust and the other Loan Documents shall be deemed to have been, and shall be, reformed and modified to reflect such reduction in such applicable interest rate or rates; and (e) neither Maker nor any of the other Obligors shall be deemed to have been, and shall be, reformed and modified to reflect such reduction in such applicable interest rate or rates; and (e) neither Maker nor any of the other Obligors shall be deemed to have been and shall be, reformed and modified to reflect such reduction in such applicable interest rate or rates; and (e) neither Maker nor any of the other Obligors shall have

15. <u>Subsequent Holders</u>. Upon any endorsement, assignment or other transfer of this Note by Lender or by operation of law, the term "Lender" as used herein, shall mean the endorsee, assignee or other transferee or successor to Lender then becoming the holder of this Note.

16. <u>Subsequent Obligors</u>. This Note and all provisions hereof shall be binding on all persons claiming under or through Maker. The terms "Maker" and "Obligors", as used herein, shall include the respective successors, assigns, legal representatives, executors, administrators, devisees, legatees and heirs of Maker and any other Obligors.

17. Interpretation. Whenever possible, each provision of this Note and the other Loan Documents shall be interpreted in such manner as to be effective and valid under applicable Law, but if any provision of this Note or any of the other Loan Documents shall be prohibited or invalid under such Law, such provision shall be ineffective only to the extent of such prohibition or invalidity, without invalidating the remainder of said documents. As used in this Note, the singular shall include the plural, and masculine, feminine and neuter pronouns shall be fully interchangeable, where the context so requires. The headings of sections and paragraphs in this Note are for convenience only and shall not be construed to limit or define the content, scope or intent of the provisions hereof.

18. Counterparts. This Note may be executed in one or more counterparts, each of which shall be deemed an original.

19. <u>Governing Law</u>. THE VALIDITY OF THIS NOTE, ITS CONSTRUCTION, INTERPRETATION AND ENFORCEMENT, AND THE RIGHTS OF MAKER AND LENDER SHALL BE DETERMINED UNDER, GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF ILLINOIS.

**20. Jury Waiver.** TO THE MAXIMUM EXTENT PERMITTED BY LAW, MAKER AND LENDER EACH HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY ACTION, CAUSE OF ACTION, CLAIM, DEMAND OR PROCEEDING ARISING UNDER OR WITH RESPECT TO THIS NOTE, OR IN ANY WAY CONNECTED WITH, RELATED TO, OR INCIDENTAL TO THE DEALINGS OF MAKER AND LENDER WITH RESPECT TO THIS NOTE, OR THE TRANSACTIONS RELATED HERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT, TORT OR OTHER WISE. TO THE MAXIMUM EXTENT PERMITTED BY LAW, MAKER AND LENDER EACH HEREBY AGREES THAT ANY SUCH ACTION, CAUSE OF ACTION, CLAIM, DEMAND OR PROCEEDING SHALL BE DECIDED BY A COURT TRIAL WITHOUT A JURY AND THAT MAKER OR LENDER MAY FILE A COPY OF THIS EXECUTED NOTE WITH ANY COURT OR OTHER TRIBUNAL AS WRITTEN EVIDENCE OF THE CONSENT OF MAKER AND LENDER TO THE WAIVER OF ITS RIGHT TO TRIAL BY JURY.

**21.** <u>Submission to Jurisdiction</u>. MAKER HEREBY IRREVOCABLY SUBMITS TO THE NON-EXCLUSIVE JURISDICTION OF ANY UNITED STATES FEDERAL OR ILLINOIS STATE COURT WITH JURISDICTION OVER COOK COUNTY, ILLINOIS IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS NOTE OR ANY OF THE OTHER LOAN DOCUMENTS AND MAKER HEREBY IRREVOCABLY AGREES THAT ALL CLAIMS IN RESPECT OF SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN ANY SUCH COURT AND IRREVOCABLY WAIVES ANY OBJECTION IT MAY NOW OR HEREAFTER HAVE AS TO THE VENUE OF ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN SUCH A COURT OR THAT SUCH COURT IS AN INCONVENIENT FORUM. NOTHING HEREIN SHALL LIMIT THE RIGHT OF LENDER TO BRING PROCEEDINGS AGAINST MAKER IN THE COURTS OF

ANY OTHER JURISDICTION. ANY JUDICIAL PROCEEDING BY THE MAKER AGAINST LENDER, DIRECTLY OR INDIRECTLY, ANY MATTER IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH THIS NOTE OR ANY OTHER LOAN DOCUMENT SHALL BE BROUGHT ONLY IN A COURT IN CHICAGO, ILLINOIS.

22. <u>Notices</u>. Any notice that Lender or Maker may desire or be required to give to the other hereunder shall be deemed given when presented in accordance with <u>Article 22</u> of the Loan Agreement.

23. <u>Conflicts</u>. In the event of any conflict between the provisions of this Note and the provisions of the Loan Agreement, the provisions of the Loan Agreement shall control.

## SIGNATURE PAGE TO FOLLOW

IN WITNESS WHEREOF, this Note has been executed and delivered under seal on the date first set forth above.

## MAKER:

CJUF II STRATUS BLOCK 21 LLC, a Delaware limited liability company

- By: Stratus Block 21 Investments, L.P., a Texas limited partnership, Manager
  - By: Stratus Block 21 Investments GP, L.L.C., a Texas limited liability company, General Partner
    - By: <u>/s/ John E. Baker</u> John E. Baker, Senior Vice President
- By: Canyon-Johnson Urban Fund II, L.P., a Delaware limited partnership, Member
  - By: Canyon-Johnson Realty Advisors II LLC, a Delaware limited liability company, General Partner
    - By: <u>/s/ K. Robert Turner</u> K. Robert Turner, Managing Partner

August 11, 2008

Securities and Exchange Commission 100 F Street, N.E. Washington, D.C. 20549

Commissioners:

We are aware that our report dated August 11, 2008 on our review of interim financial information of Stratus Properties Inc. for the three and six month periods ended June 30, 2008 and 2007 and included in the Company's quarterly report on Form 10-Q for the quarter ended June 30, 2008 is incorporated by reference in its Registration Statements on Form S-8 (File Nos. 33-78798, 333-31059, 333-52995 and 333-104288).

Very truly yours,

/s/ PricewaterhouseCoopers LLP

Dallas, Texas

Certification

I, William H. Armstrong III, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Stratus Properties Inc.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 11, 2008

<u>/s/ William H. Armstrong III</u> William H. Armstrong III Chairman of the Board, President and Chief Executive Officer

- 1. I have reviewed this quarterly report on Form 10-Q of Stratus Properties Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 11, 2008

/s/ John E. Baker John E. Baker Senior Vice President & Chief Financial Officer Certification Pursuant to 18 U.S.C. Section 1350 (Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002)

In connection with the Quarterly Report on Form 10-Q of Stratus Properties Inc. (the "Company") for the quarter ending June 30, 2008, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), William H. Armstrong III, as Chairman of the Board, President and Chief Executive Officer of the Company, hereby certifies, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that, to the best of his knowledge:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: August 11, 2008

/s/ William H. Armstrong III William H. Armstrong III Chairman of the Board, President and Chief Executive Officer

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

This certification shall not be deemed filed by the Company for purposes of § 18 of the Securities Exchange Act of 1934, as amended.

Certification Pursuant to 18 U.S.C. Section 1350 (Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002)

In connection with the Quarterly Report on Form 10-Q of Stratus Properties Inc. (the "Company") for the quarter ending June 30, 2008, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), John E. Baker, as Senior Vice President & Chief Financial Officer of the Company, hereby certifies, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that, to the best of his knowledge:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: August 11, 2008

/s/ John E. Baker John E. Baker Senior Vice President & Chief Financial Officer

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

This certification shall not be deemed filed by the Company for purposes of § 18 of the Securities Exchange Act of 1934, as amended.